MOVING TOWARDS DISASTER: EXAMINING THE CHANGING PATTERNS OF SOCIAL VULNERABILITY IN A MULTI-HAZARD URBAN ENVIRONMENT

BY

MD SHAKIL BIN KASHEM

DISSERTATION

Submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Regional Planning in the Graduate College of the University of Illinois at Urbana-Champaign, 2015

Urbana, Illinois

Doctoral Committee:

Assistant Professor Bev Wilson, Chair Professor Robert Olshansky Associate Professor Arnab Chakraborty Associate Professor Shannon Van Zandt, Texas A&M University

ABSTRACT

Studies of social vulnerability have repeatedly emphasized the importance of identifying the drivers of vulnerability, but very few studies have focused on empirically characterizing those drivers within the domain of vulnerability science that would help in effective policymaking. This dissertation is an initial step in this direction, examining social vulnerability in the context of multiple cities and evaluating the changing patterns of vulnerability in a multihazard urban environment. It adopts a political-economic framing of vulnerability production (Dooling and Simon 2012) that conceptualizes vulnerability as a dynamic condition, produced through the historic interaction of economic, cultural, and social processes. It hypothesizes that the nature and distribution of social vulnerability in urban areas changes over time, and that the provision of subsidized low-income housing influences the hazard exposure of socially vulnerable populations. This is accomplished first by studying three cities in the Gulf coast region (Houston, New Orleans, and Tampa) and then by focusing on Houston, Texas as a case study city for a more detailed empirical analysis. The initial component of this research integrates neighborhood change theories and theories of social vulnerability to explain the changing patterns of social vulnerability in Houston, New Orleans, and Tampa over a 30 year time period (1980-2010). Next, the Houston case study further explores how vulnerable groups navigate the multi-hazard urban environment and how subsidized housing policies have influenced this interaction over time.

The pattern of social vulnerability observed within the case study cities indicates that despite having drastically different population growth trajectories and being situated in different political and economic settings, the spatial concentration of social vulnerability has gradually decreased in the study cities in recent decades. Specific trends in vulnerability are identified for each of the cities and the potential for constraining climate adaptation efforts is discussed. After analyzing the location of subsidized housing in Houston, this study found that among the two most widespread housing subsidy programs (Housing Choice Vouchers and the Low Income Housing Tax Credit), supply based subsidies exemplified by the LIHTC significantly increases neighborhood social vulnerability when it is located in areas exposed to technological hazards. Limitations in the present administration of the subsidy programs are identified and policy alternatives are discussed that may help to reduce their contribution to vulnerability.

ACKNOWLEDGEMENTS

It was a memorable journey for me to complete my doctoral study at UIUC and to work on this dissertation. I am immensely grateful to people who guided me or simply accompanied me from time to time in this apparently lonesome journey. First and foremost, I would like to express my deep gratitude to my advisor, Professor Bev Wilson, whose consistent guidance and inspiration was the key for me completing this dissertation. Right from the start of taking on this project and then materializing it, he supervised me all the way without being overbearing and kept me focused on my work. I am highly indebted to him for his meticulous comments and edits which helped me to clarify my thoughts and ideas in this dissertation. At a personal level too I was fortunate to have him as my mentor, someone who cares about helping his students to flourish academically and professionally.

I am also very grateful to my dissertation committee members for their insightful comments and suggestions. It was amazing how they sent me their individual comments pointing out the places where I needed to update my writing or to address prior to submission for formal publication. Professor Rob Olshansky inspired me from the very beginning of undertaking this study on social vulnerability and gave me valuable suggestions on how to contextualize it at the local community level while also considering the broader region. Professor Arnab Chakraborty also provided valuable comments on my study approach and on the issues that can be improved or explored in future research. I am particularly indebted to Professor Shannon Van Zandt for all her support in conducting this study. Being situated in Texas and also having varied research experience on housing and disasters, she was not only a great help for me to conduct my case study in Houston but also a great inspiration for me to integrate housing policy, vulnerability, and disaster risk in my work.

I am deeply grateful to Professor Salim Rashid (Economics, UIUC) who encouraged me to come to Illinois for my doctoral study in the first place and was just like a father figure to me all the times. I am also deeply grateful to Late Professor Andy Isserman who helped me to transfer to the planning PhD program right after my first semester in Economics. I am thankful to Professor Faranak Miraftab, who helped me in reorienting my research ideas during my first year here and then continuously provided all kinds of support and encouragement throughout my doctoral study. Thanks should also go to Professor Jesse Ribot (Geography, UIUC) whose course on climate change and social vulnerability helped me to refine my research focus. I would like to thank people at HHA, particularly Brian Gage (Senior Policy Advisor) and Ross LaFour (Policy Analyst), who helped me to collect data on subsidized low-income housing in Houston.

I am thankful to my PhD colleagues at UIUC who gave valuable inputs to my research ideas during the seminar presentations and also provided all kind of supports to feel at home at Urbana-Champaign. I am particularly grateful to Sang Lee and Sofia Sianis, with whom I participated in an informal group for managing our dissertations, which helped me a lot in the last few months of writing. I am also thankful to my other PhD colleagues here: Ahmad Gamal, Shruti Syal, Steve Sherman, Dwayne Baker, Deniz Ay, Esteban Lopez, Maximilian Eisenburger, Zach Kennedy, Sung-won Lee, Andrew McMillan, and LaKisha David, for their encouragement at different occasions. Special thanks should go to Andi Irawan and Troy Mix. Andi not only collaborated with me in different research projects, but also encouraged me to finish up my dissertation. Troy, besides being a good friend of mine, helped me in developing the materials for teaching UP 418 (GIS for Planners). I am also thankful to Riaz Uddin and Raktim Mitra (Ryerson University), both from my alma mater in Bangladesh and provided me all kinds of support and encouragement to come to Illinois.

I am grateful to the Bangladeshi community at Urbana-Champaign who were like an extended family to me far away from my own family back home. Particularly I am grateful to: Hasib Uddin, Ashiqur Rahman, Yakut Ali Rana, Piyas Bal, Kallol Das, Ahmed Khurshid, Samira Masoom, Munawar Hafiz, Mehedi Bakht, Fariba Khan, Atanu Khan, Abdullah Muzahid, Hossain M Azam, Mohammad Sharif Ullah, Imranul Hoque, Sonia Jahid, and others who helped me here in different occasions. I am also grateful to my friends and colleagues at ATLAS: Maryalice Wu, Dawn Owens-Nicholson, Kathleen Santa Ana, and Laura Beth, who besides being very good friends of mine taught me how to be a good researcher and educator.

Above all, I am grateful to my family, particularly my wife Kazi Nusrat Jahan who is my source of motivation all the time. I am also thankful to my mom Ferdous Ara, dad Md. Abul Kashem, brother Shoaib Bin Kashem, sisters Mahmuda Sultana and Mahfuza Sultana, and brothers-in-law Saiful Alam and Ariful Islam, and my parents-in-law who continuously provided me moral support. Lastly, I would like to express my gratitude to the Department of Urban and Regional Planning for providing financial support through my teaching assignments and for providing travel grants that helped me to attend conferences and to conduct my case study.

TABLE OF CONTENTS

CHAPTER 1: INTRODUCTION	1
CHAPTER 2: LITERATURE REVIEW AND RESEARCH FRAMEWORK	8
CHAPTER 3: RESEARCH METHODOLOGY	23
CHAPTER 4: CHANGING PATTERNS OF SOCIAL VULNERABILITY IN	
COASTAL CITIES	
CHAPTER 5: SUBSIDIZED HOUSING AND SOCIAL VULNERABILITY IN A	
MULTI-HAZARD URBAN AREA	75
CHAPTER 6: HOUSING POLICY FOR MULTI-HAZARD ENVIRONMENT	
CHAPTER 7: CONCLUSION	134
REFERENCES	136
APPENDIX A: RESULTS OF PCA ANALYSIS	156
APPENDIX B: CORRELATION BIPLOTS	159
APPENDIX C: MODEL RESULTS AND COMPARISONS	162
APPENDIX D: HUD GUIDELINE FOR CALCULATING ACCEPTABLE	
SEPARATION DISTANCE (ASD)	174

CHAPTER 1 INTRODUCTION

1.1 Introduction

Prior studies of social vulnerability and disaster risk management have repeatedly emphasized identifying the drivers of vulnerability, but very few studies have focused on empirically characterizing those drivers. Understanding these drivers within a multi-hazard urban context is important given the differential hazard exposure and disaster outcomes experienced by vulnerable low-income and minority population groups (Crowder & Downey, 2010; Finch et al., 2010; Van Zandt et al., 2012). In recent years, urban sustainability and climate adaptation have gained increased attention due in large part to growing concerns related to the impacts of climate change (Blanco et al., 2009). The devastation wrought by Hurricanes Katrina and Sandy are two recent examples that clearly highlighted the vulnerabilities of major coastal cities and have renewed calls for a planning response. Billions of dollars are now being spent to restore the economies of impacted localities and to prepare for future climatic impacts but comparable resources have not been dedicated to the social dimensions of vulnerability, namely recovery of the urban poor or how to ensure secure and safe housing for vulnerable population groups. This neglect of social vulnerability is an important limitation of adaptation efforts, which promises to reduce climate change impacts but "tend to exclude the possibility of non-adaptation from consideration" (Orlove, 2009). Despite these efforts, socially vulnerable populations may be increasingly concentrated in hazard exposed areas (Mohai & Saha, 2006; Wisner et al., 2004), significantly increasing the overall vulnerability of cities. What is required now is to consider the dynamics of social vulnerability and to identify the drivers that make the vulnerable population groups less able to avoid hazardous areas. This dissertation explores the changing patterns of social vulnerability and adopts a political-economic framing of vulnerability production (Dooling & Simon, 2012) to critically examine the drivers of vulnerability. It offers policy recommendations that move beyond the conventional notion of vulnerability as static and exogenous to urban development and politics. Finally acknowledging that there are multiple drivers of social vulnerability in a multi-hazard urban environment, this study focuses on subsidized housing and evaluates how market dependent housing subsidy programs for lowincome people can act as one of the drivers contributing to conditions that are ripe for disaster.

1.2 Study Background

Although vulnerability has been defined differently across disciplines, it generally refers to susceptibility to harm (Adger, 2006; Füssel & Klein, 2006). However, even within this general definition of vulnerability, there are significant differences in its conceptualization in the policy arena and scholarly discourse. In the climate change literature for example, vulnerability is often understood to be a function of exposure to a hazard, sensitivity to that hazard, and the ability to respond accordingly (Parry et al., 2007; Smit & Wandel, 2006). In this biophysical conception of vulnerability, risk is seen as exogenous to society (McLaughlin & Dietz, 2008) as something that can be projected onto a socially-static landscape. It has been widely recognized, however, that adaptive capacity, exposure, and sensitivity are not static, and instead vary widely over time and across subpopulations, as do the damages experienced (Handmer, Dovers, & Downing, 1999; Turner et al., 2003). From this point of view, a social constructivist approach to vulnerability locates risk within society and places the burden of explanation of vulnerability within the social system (Ribot, 2009). An integrative vulnerability framework on the other hand, links both riskhazard and social constructivist models of vulnerability, and understands vulnerability as having "an external dimension, which is represented by the 'exposure' of a system, as well as internal dimension, which comprises the 'sensitivity' and 'adaptive capacity' to these stressors" (Füssel & Klein, 2006). This study adopts an integrative framework of vulnerability to explain social vulnerability dynamics, where the external dimensions of hazard risks interact with internal dimensions of social vulnerability within the broader political-economic context of a region.

While scholarship on hazard vulnerability (or biophysical risk) offers significant insights for understanding the geography of unequal risk, it provides an incomplete basis for examining contemporary generative processes (Collins, 2009, 2010). Political-economic theory and analysis focuses on this generative process by explicitly considering changes in economic policies, technological systems, institutional arrangements, and demographic processes, all of which shape social vulnerability (Collins & Jimenez, 2012). Politically and economically powerful institutions and individuals, through their political ideologies and strategies, influence access to human and natural resources for different social and economic groups, and thereby act as the drivers for production of vulnerability (Dooling & Simon, 2012; Pelling, 2003). Following the work on vulnerable spatialities of Findlay (2005), Dooling & Simon (2012) elaborated this production-oriented framework for analyzing how interactions between political economies of

resource use and normative planning and management interventions influence which places and populations are made vulnerable. In recent years, a number of studies have examined the effects of neoliberal¹ policies on public space, neighborhood change, gentrification, and other forms of uneven development (Hackworth & Smith, 2001; Lees, 2003; Newman & Ashton, 2004; Perkins, 2012). It has also been argued that the increased concentration of poverty in the United States over the last 40 years was more a result of housing policy than the result of income inequality (Reardon & Bischoff, 2011). Given this contextualization, this study examines the extent to which subsidized housing programs functioning within a neoliberal policy environment contribute to hazard exposure for socially vulnerable communities.

Poverty and access to stable, affordable housing are key factors in determining a household's ability to withstand socio-economic stresses in urbanizing environments (Moser, 1998; Sanderson, 2000). But in recent decades, policies designed to facilitate the provision of affordable housing in U.S. cities have undergone important changes. Since the early 1970s, a significant number of public housing projects have been dismantled and replaced with marketoriented solutions such as rental assistance vouchers and HOPE VI housing developments. Disinvestment in inner-cities, the shifting of new investment to suburbs, designating areas as blighted, and post-disaster reconstruction have justified the clearance of "slummed" neighborhoods resulting in direct subsidies to private real estate investors and the demolition of public housing (Angotti, 2008; Crump, 2002; Kamel, 2012). However, these market-oriented programs suffered from funding deficits since their inception and subsequent reductions in federal housing spending have exacerbated unmet housing needs (Malpass, 2003; Popkin et al., 2004). A significant proportion of low-income households, especially inner-city minorities, remained trapped in substandard, overcrowded, and over-priced housing (HUD 1993; 2007). Prior research has considered whether programs for low-income housing succeeded in deconcentrating poverty (Goetz, 2005; Massey & Denton, 1993; Wilson, 1987) or provided any improvement in life outcomes for participating household and communities (Freeman, 2003; Goering et al., 2002; Van Zandt & Mhatre, 2009), but very few studies have explored the extent to which these housing provisions have increased or decreased the hazard exposure of vulnerable

¹ Harvey (2005) identifies neoliberal policies as emerging from political-economic practices that "proposes that human well-being can best be advanced by liberating individual entrepreneurial freedoms and skills within an institutional framework characterized by strong private property rights, free market, and free trade." As Perkins (2012) asserts, neoliberal modes of environmental governance usually prioritize market profitability and personal responsibility.

populations (e.g. Cutter et al. 2001; Houston et al. 2013).

Prior research on social vulnerability in urban areas has tended to focus on exposure to natural hazards (Maantay & Maroko, 2009b; Zahran et al., 2008) or differential outcomes in the context of recovery planning and disaster impacts (Van Zandt et al., 2012; Zhang & Peacock, 2009), but the changing pattern of social vulnerability and the generative process of vulnerability has yet to be examined. Adopting the production-oriented framework of vulnerability (Dooling & Simon, 2012) this dissertation investigates the intensity and persistence of conditions of social vulnerability (Collins, 2010; Davis, 1998; Peet & Watts, 1996). Since the location and availability of subsidized low-income housing is considered a critical determinant in the spatial distribution of socially vulnerable populations (Moser, 1998; Sanderson, 2000), it is important to understand how these programs, in their present form, may influence social vulnerability within a multi-hazard landscape.

Exponential population growth along the coasts of the United States in recent decades has increased the number of people and amount of property vulnerable to high winds, waves, and storm surge flooding of catastrophic coastal storms (Burby, 1998; Deyle et al., 2008; Godschalk et al., 1999). Therefore, coastal cities are the most appropriate test cases for the vulnerability framework discussed above. This dissertation selects three coastal cities (Houston, New Orleans, and Tampa) for exploring the changing pattern of social vulnerability over a 30 year time period (1980-2010). These cities were chosen due to the similarity of their geographic locations (i.e., all are located in Gulf Coast region) and because they exhibit significantly different patterns of population growth. The study then focuses exclusively on Houston to evaluate how subsidized housing programs may have influenced social vulnerability and hazard exposure. Houston is adopted as the subject of a detailed case study due to its high level of natural and technological hazard exposure (Nicholls et al., 2008; Schiller, 2010) and its situation within the archetypal neoliberal state of Texas (Miller et al., 2011).

1.3 Objectives, Research Questions, and Hypothesis

The primary objective of this study is to evaluate how urban growth and decline shapes and changes patterns of social vulnerability and how subsidized low-income housing programs influence social vulnerability in a multi-hazard environment. This dissertation will be guided by following central questions:

- 1. How do the spatial patterns of social vulnerability change over time in coastal metropolitan areas and how do these patterns vary across different cities (in terms of the distribution of vulnerable population groups and indicators of social vulnerability)?
- 2. Are socially vulnerable populations more likely to live in hazard exposed (both natural and technological) areas and if so, has this relationship changed over time?
- 3. To what extent have subsidized housing programs influenced the hazard exposure of socially vulnerable population groups?
- 4. How can existing subsidized housing programs be modified to better address the problems of hazard exposure and disaster risk reduction?

Considering the above research questions, this study tests three related hypotheses:

- H1. With the growth of a city, vulnerable population groups become less concentrated over time. The suburbanization of poverty, limited availability of affordable housing, and gradual gentrification of inner-city poor neighborhoods can contribute to such trends. The dimensions of social vulnerability also vary through changing demographic composition.
- H2. Over time, vulnerable population groups increasingly move into hazard exposed areas of a city, which further skews the already uneven geography of hazard exposure. A lack of housing security, decreasing provisions for a social safety net, and more limited housing options due to urban revitalization and gentrification are some of the key drivers behind this uneven geography.
- H3. Subsidized housing programs have failed to reduce the overall hazard exposure of socially vulnerable populations and to some extent, have contributed to an increased level of hazard exposure. Exposure to technological hazards may be higher than exposure to natural hazards due to the concentration of these land uses in space over time and their attendant impacts on land values and rent in these areas of the city.

1.4 Dissertation Outline

To address the research questions mentioned above, this study begins by exploring the patterns of social vulnerability in three coastal cities, then narrows its focus to a detailed case study of subsidized low-income housing in Houston, Texas. Chapter 2 reviews literature on

social vulnerability, and particularly how different schools of thought have contributed to formalizing the concept of urban social vulnerability. It also gives a brief review of low-income housing subsidy programs in the U.S. and how these programs may influence the location decisions of vulnerable populations in a multi-hazard environment. This chapter also presents the overall framework of the research, summarizing the key theoretical basis and how the study is conducted in two stages. Chapter 3 presents the study methodology and briefly describes the three case study cities before elaborating the data processing methods, procedure for calculating social vulnerability, approach for analyzing the temporal trends of vulnerability, methods for calculating natural and technological hazard exposure, and later how the modeling framework was selected for evaluating the impacts of subsidized housing and hazard exposure on changing patterns of social vulnerability at census tract level.

Chapter 4 discusses the results of analyzing the changing patterns of social vulnerability in three coastal cities over a thirty year time period (1980-2010). It also presents a brief review of relevant neighborhood change theories before synthesizing social vulnerability and neighborhood change theory to explain the identified patterns of social vulnerability. This chapter concludes with a call for developing a land use planning framework that is more responsive to the changing patterns of vulnerability in a city. Chapter 5 presents the results from the detailed case study of Houston (Harris County), Texas evaluating the spatial distribution of socially vulnerable populations within the multi-hazard landscape and how the subsidized lowincome housing that existed there in 2000 and 2010 was distributed across different natural and technological hazard zones. This chapter also interprets the results of spatial econometric models that evaluate the interaction between subsidized housing and hazard exposure and how this may have contributed to an increase in social vulnerability within hazardous areas of the county.

Chapter 6 focuses on identifying limitations in present housing policies and policy alternatives that would ensure that low-income housing subsidies are channeled into lower hazard areas of the city. It evaluates the environmental requirements of housing projects that receive assistance from the U.S. Department of Housing and Urban Development (HUD), then looks at the requirements of two of the most popular subsidy programs—Housing Choice Vouchers (HCV) and Low Income Housing Tax Credit (LIHTC). Based on interviews with officials at the Houston Housing Authority and site visits to selected tax credit properties, this chapter further explains and contextualizes the findings from the spatial econometric models

presented in Chapter 5. Later this chapter proposes policy alternatives for modifying existing provisions of HCV and LIHTC, particularly how these programs can be more responsive to the hazard characteristics of an area and how these programs can ensure that socially vulnerable populations are not concentrating in those areas. This dissertation concludes with Chapter 7, which summarizes the findings of this study and reiterates the importance of considering the changing patterns of vulnerability in the current initiatives for adaptation planning as well as how our subsidized housing programs can be an integral part of climate adaptation initiatives.

CHAPTER 2

LITERATURE REVIEW AND RESEARCH FRAMEWORK

2.1 Introduction

Vulnerability science is multidimensional with researchers working in the fields of disaster management and hazards, environmental justice, food and water security, and climate change contributing theories that have helped to define and advance the field of vulnerability studies (Eakin & Luers, 2006; Ionescu et al., 2009; Kasperson et al., 2005; Turner et al., 2003). Scholars evaluating the dynamic tensions of vulnerability² examine more explicitly the connections between pre-existing and emerging economic, environmental and social conditions that impact vulnerable communities (Andrey & Jones, 2008; Dooling & Simon, 2012; Hogan & Marandola, 2005; Pelling, 2003). Prior studies adopting a political-economic framing of vulnerability production have specifically explored these broader dynamic tensions, but detailed research that evaluates the outcomes of low-income housing programs can also make valuable theoretical contributions and yield useful policy recommendations. This dissertation represents an initial step in this direction. The remainder of this chapter discusses theoretical models of vulnerability itself, and then considers the political economic framing of vulnerability and the environmental justice framing of vulnerability. A brief overview of subsidized low-income housing programs is provided later, followed by a more detailed discussion of residential location choice and the hazard exposure of subsidized housing to explain their linkages under the political-economic and environmental justice framing of vulnerability.

2.2 Theoretical Models of Vulnerability

Interaction between biophysical and social vulnerability has always been a contentious issue among researchers. The biophysical conception of vulnerability considers risk as exogenous to society (McLaughlin & Dietz, 2008) and climate-related hazards can therefore, be mapped onto a socially-static landscape where adaptive capacity and sensitivity are assumed to be fixed in a particular geographic area. In contrast, a social constructivist approach relies on the theories of political economy and political ecology to uncover and evaluate the structural origins

² Dynamic tensions of vulnerability refer to the processes through which conditions of, and experiences with, vulnerability are produced through specific cross-scale interactions that are historic in nature (Dooling & Simon, 2012).

of vulnerability. Bohle et al. (1994) provided an early model of social vulnerability rooted in the political economy tradition and in their view, vulnerability is best understood and studied using concepts rooted in human ecology, political economy, and entitlement theory. Three of the most prominent theoretical models of vulnerability that draw upon this political economy tradition are briefly discussed here.

2.2.1 Pressure and Release Model

The Pressure and Release (PAR) model (Blaikie et al., 1994) addressed more specifically the construction of some of the contextual factors influencing vulnerability. In the PAR model, vulnerability is seen as part of a risk equation: Risk = Hazard + Vulnerability. Risk, in this application, is distinct from its traditional definition as the probability of event occurrence (Cutter, 1996), or as Sarewitz et al. (2003) put it, event risk. Risk in the above formula is what Sarewitz et al. called outcome risk—that is the probability of a *specific* outcome occurring. Vulnerability within the PAR model is conceptualized as stemming primarily from social structures and characteristics. Vulnerability originates from a variety of root causes, leading to what are termed as dynamic pressures. These dynamic pressures localize the influence of the broader root causes (e.g., poverty) into unsafe living conditions. Unsafe living conditions in turn interact with the probable or actual occurrence of hazard events to produce what may be called a higher outcome risk for less advantaged members of society. This model developed the idea that the consequences of a hazard event depend not only on the event in question, or even solely on the direct human-environmental systems, but also on the broader structure and characteristics of human society.

2.2.2 Hazards of Place Model

The Hazards of Place (HOP) model of vulnerability (Cutter, 1996) attempted to bridge the gap between vulnerability approaches focused solely on exposure, and those centered exclusively on social conditions related to resistance or resilience. Cutter proposed that the idea of *place* be used to unify these approaches and the diagram shown in Figure 2.1 offers an overview. Interactions between people and their environment occur within a particular place and places have a unique hazard potential, which arises from the interaction between hazard risk and socially determined mitigation activities (or lack thereof). Geographic contexts (characteristics) within the study area, such as elevation and proximity, work to modify hazard potential across space. This modified, spatially differentiated hazard potential is called biophysical vulnerability. The hazard potential is also modified by the social fabric of the area, which in turn modifies and differentiates the hazard potential across space. The social fabric consists of those characteristics that describe the distribution and composition of the population measured, for example, by socio-demographic, economic and welfare variables. The social contributions to the spatial differences are called *social vulnerability*.



Figure 2.1 Hazards of Place (HOP) model of vulnerability (after Cutter (1996))

Biophysical and social vulnerability interact to create an overall place vulnerability, but place vulnerability can, in turn, modify both the event risk of an area as well as the mitigation approaches used in the area. The linkages displayed in the model (Figure 2.1) communicate an understanding of the dynamic nature of vulnerability—namely that changes in the physical and social setting of an area may result in changes in vulnerability, risk, mitigation, etc.

2.2.3 Vulnerability Framework for Sustainability Science Model

The Vulnerability Framework for Sustainability Science (VFSS) model (Turner et al., 2003) was also developed to bridge gaps in existing approaches to studying vulnerability. But while the HOP model reconciled competing understandings of the factors contributing to vulnerability (physical and social), the VFSS approach sought to bridge the risk-hazards and the political economy and political ecology traditions as different ways of conceptualizing vulnerability. While it includes vulnerability stemming from biophysical subsystems seen in the

risk-hazards approaches, it also reconciles this with the multi-scaled and structural explanations of the political economy approaches³. Like the HOP model, it employs the geographic concept of place as the lens through which the interactions of social and physical systems are analyzed. In this framework, vulnerability occurs within a specific place, but is influenced by human and environmental influences at regional and global levels.

Although the models discussed above explore vulnerability from different perspectives, all of them can be traced to the political-economic framing of vulnerability. This dissertation makes use of these established theoretical frameworks, but draws most heavily on the HOP model (Cutter, 1996) and VFSS framework (Turner et al., 2003). In order to evaluate the pattern of social vulnerability within the study cities, it adopts Cutter's approach, but for exploring the generative dynamics of vulnerability it takes cues from Turner et al.'s VFSS framework given the latter's emphasis on political-economic explanations of vulnerability dynamics.

2.3 Political-Economic Framing of Vulnerability

A political-economic perspective on hazard exposure and vulnerability provides a foundation for describing uneven patterns of risk (Bolin & Stanford, 1998; Hewitt, 1983; Susman et al., 1983; Wisner et al., 1976, 2004). Like the PAR model of vulnerability (discussed above), this perspective considers risk as the product of people's exposure to an environmental hazard *and* their social vulnerability⁴. Political-economic theory and analysis focuses on the generative process of vulnerability by integrating changes in economic policies, technological systems, institutional arrangements, and demographic processes, all of which shape contemporary experiences of vulnerability. It questions the neoliberal objective of aggregate economic growth that promises to improve human well-being, but ultimately ignores the production of social vulnerability (Cannon & Müller-Mahn, 2010). Empirical analysis and theorizing about the neoliberal project provides the basis for understanding and challenging environmental injustice created by the neoliberal agenda.

While a political-economic framing allows us to understand the broader generative process of vulnerability, it too is insufficient to explore the vulnerability dynamics in an urban context when taken in isolation. Risk exposure to urban environmental hazards is a complex

³ Specifically the linkages outlined from global root causes to local unsafe living conditions in the PAR model.

⁴ The capacity to anticipate, respond to, and recover from exposure to a chronic stressor or perturbation (Wisner et al., 2004).

phenomenon, with overlapping risks associated with the household, workplace, or neighborhood, and pollution risks from industrial contamination (Hardoy et al., 2001). Poverty and access to stable, affordable housing are key factors in determining a household's ability to withstand socio-economic stresses in urbanizing environment (Dooling & Simon, 2012; Moser, 1998; Sanderson, 2000). In addition, people without access to safe housing are frequently the group most harmed by environmental hazards (Pelling, 2003), and on the other hand, neoliberal policies also impede a community's ability to preserve and deliver affordable housing (Kamel, 2012). In addition to housing affordability, studies have also demonstrated the importance of local stresses as contributors to vulnerability in the context of health, racial, gender and age composition of affected households and communities (Phillips et al., 2009). Within the urban context, it is argued that social vulnerability emerges as a response to, and a byproduct of, phenomena operating at larger scales, including national policies, global financial markets, and regional environmental disasters (Collins, 2009; Dooling, 2012). In order to better conceptualize the role of these larger-scale phenomena vulnerabilities must be understood as conditions that are created and maintained through historical relationships and arise from the interaction of economic, cultural, and social processes (Andrey & Jones, 2008; Blaikie et al., 1994; Hogan & Marandola, 2005; Pelling, 2003).

A production-oriented framework of vulnerability (Dooling & Simon, 2012) analyzes how interactions between political economies of resource use and normative planning and management interventions—at both global and local scales—influence which places and populations are made vulnerable as well as the intensity and persistence of conditions of vulnerability (Collins, 2010; Davis, 1998; Mustafa, 2005; Orsi, 2004; Wisner et al., 2004). It focuses on articulating how the conditions and experience of vulnerability are produced, regulated, manipulated and resisted. By detailing the relationship between vulnerability and planning agendas that guide urban sustainability, gentrification, suburban development, climate change adaptation, and other planning initiatives, this production-oriented framing places vulnerability within the broader field of urban political ecology (Dooling, 2012; Heynen et al., 2006). This framing focuses on the contradictions of a planning agenda that ignores the dynamics of social vulnerability and thereby, exacerbates existing risks and harms groups of people that are the least able to avoid the risks (Dooling, 2012).

12

2.4 Environmental Justice Framing of Vulnerability

An environmental justice framing of vulnerability contributes important insights into the underlying dynamics of hazard exposure and vulnerability (Boone & Fragkias, 2013). The environmental justice literature examines inequalities in technological hazard exposure by race and class and has coalesced around concern and action regarding the societal distribution of environmental hazards and their health effects (Buzzelli, 2007). It specifically explores the nature and extent of disproportionate exposures to health hazards, ranging from toxic waste sites and air pollution to the landfill siting process, and how this exposure varies across population groups (Chakraborty & Armstrong, 2001; Crowder & Downey, 2010; Hamilton, 1995; Maantay, 2001). Buzzelli (2007) presents a schematic (Figure 2.2) of the environmental justice conundrum, indicating that as Socio-Economic Position (SEP) rises, the corresponding exposure to environmental health hazards among individuals and neighborhoods diminishes. This general framing of environmental justice is augmented by two competing viewpoints (Crowder & Downey, 2010): the *racial income inequality thesis* and *residential discrimination thesis*, which are briefly discussed below.



Figure 2.2 Schematic of environmental (in)justice (Buzzelli, 2007).

The *racial income inequality thesis* (Downey, 2005; Oakes et al., 1996) argues that, racial differences in exposure and proximity to environmental hazards largely reflect group differences in socioeconomic resources (Crowder & Downey, 2010). It holds that environmentally hazardous neighborhoods have relatively low property values and rents, which make those areas more accessible to lower-income families. Low-income families on the other hand are overrepresented by non-white families and as a result, higher proportions of non-white populations also live in hazardous areas.

The *residential discrimination thesis* (Bullard, 1999; Mohai & Bryant, 1998) is more critical than the racial income-inequality thesis in explaining differential exposure to hazards. It argues that housing market discrimination reduces the ability of minority households to move out of, or to avoid moving into, hazardous neighborhoods, thereby creating or maintaining environmental racial inequality (Crowder & Downey, 2010). Studies based on this thesis show how discriminatory actions by real estate agents (Pearce, 1979; Yinger, 1995), local governments (Shlay & Rossi, 1981), and mortgage lenders (Ross & Yinger, 2002) limit residential location choice for minority home seekers (Galster & Keeney, 1988; Massey & Denton, 1993).

Besides these two viewpoints some studies have assessed the argument that environmental injustice emerge because environmental hazards are disproportionately sited in minority neighborhoods who lack political influence to resist it (Downey, 2005; Pastor et al., 2001). However, it is argued that in the context of high levels of residential mobility, initial siting decisions may have relatively less influence on prevailing patterns of hazard exposure (Crowder & Downey, 2010). Also, as both the income-inequality and discrimination perspectives predict, in comparison to whites, members of minority groups will be less likely to leave and more likely to enter, polluted neighborhoods, thereby increasing their overall proximity and exposure to environmental hazards (Downey, 2005; Hamilton, 1995; Mohai & Bryant, 1998; Oakes et al., 1996). Although environmental justice studies focus more on assessing fairness and vulnerability studies focus on the biophysical and social system, combining them in a common framework can strengthen and advance the goals of each (Boone & Fragkias, 2013). This study facilitates this integration by exploring vulnerability dynamics in a multi-hazard environment.

2.5 Subsidized Low-Income Housing in the U.S.

Federally assisted public housing began in 1930s, primarily to stimulate employment after the crisis of the Great Depression in the U.S. (Cutter et al., 2001; Wyly & DeFilippis, 2010). Over time, low-income housing assistance has evolved with the federal government playing a reduced role that favors state and locally led partnerships supplemented with a blend of block grants, tax credits to private developers, and vouchers (Wyly & DeFilippis, 2010). After widespread criticism of public housing as a key contributor to the creation and exacerbation of social and economic segregation in cities, federal housing policy since the 1970s has focused on two objectives-to help depressed neighborhoods through the construction of new affordable housing and to deconcentrate poor subsidized-housing residents (Galster, 2013; Scally & Koenig, 2012), but these policies have had relatively little impact in improving the conditions of low income neighborhoods (Galster, 2013). Many studies have argued that the placement of subsidized housing creates negative spillover effects in neighborhoods such as the devaluation of land, increased crime, and middle class flight and eventually lead to further concentrations of poverty (Carter et al., 1998; Galster et al., 1999; Holloway et al., 1998). However, other studies have argued that much of the negative reaction to low-income subsidized housing developments is unwarranted (Freeman, 2003; Green et al., 2002; Oakley, 2008). Further complicating matters is the fact that the outcomes of subsidized housing programs tend to vary by specific program and across local housing markets. A brief review of relevant subsidized housing programs is presented here⁵ and followed by a discussion of residential location choice and the hazard exposure of subsidized housing.

2.5.1 Scattered-Site Public Housing

As evidence mounted that public housing was contributing to the concentration of innercity poverty, at the beginning of 1970s, the U.S. Department of Housing and Urban Development (HUD) encouraged local Public Housing Authorities (PHAs) to develop more housing on a "scattered-site" basis (Galster, 2013). This scattered-site housing was typically operationalized through the construction and/or acquisition of low-density buildings with fewer than 15 units per site in locations that were not disproportionately minority-occupied (Hogan, 1996). However, this strategy was not widely adopted across the nation due to the near-elimination of funding

⁵ For a more detailed treatment, see Schwartz (2010).

from federal sources and there was huge variation in the density and locations of "scattered sites" across PHAs (Galster, 2013). In general, it is argued that scattered site public housing programs have offered superior neighborhood environments for low-income tenants compared to conventional, large-scale, concentrated public housing developments, but the outcome also vary by the contexts of local housing markets (Galster, 2013).

2.5.2 Housing Choice Vouchers Program

Vouchers, authorized by Section 8 of the Housing and Community Development Act of 1973, have been regarded as inherently superior to public housing in that they enable poor families to live in moderate- or middle-income neighborhoods (Wyly & DeFilippis, 2010). Since its inception in 1974, this housing assistance program had followed the general formula that the qualifying household must contribute a share of its income (currently 30%) toward rent of an apartment that meets certain quality standards and whose landlord agrees to sign a minimum one-year lease with the tenant and the PHA administering the voucher (Galster, 2013). In 1998 Section 8 was renamed the Housing Choice Voucher (HCV) program. By encouraging residents of high-poverty neighborhoods to relocate to outlying areas, the program is intended to make more jobs accessible to program participants (Briggs, 1997; Goering et al., 2002), but an enduring problem is that its effectiveness depends on the availability of affordable units in neighborhoods with low concentrations of poor persons. Pendall (2000) attributes the shrinking supply of affordable units to a combination of increased competition for land and unfavorable tax codes that hinder the development of multifamily housing. Efforts to close older public housing projects under the HOPE VI program (discussed later) also negatively affect the supply of affordable units. Further, despite the option to move out of high-poverty neighborhoods, many program participants choose not to move at all and instead opt to remain close to family, friends, and other social institutions (Briggs, 1997; Popkin et al., 2004).

Feins & Patterson (2005) conducted a longitudinal analysis using a national sample of those entering the HCV program from 1995 to 2002 and found that the trajectory of moves was not into significantly better neighborhoods (measured on many characteristics) over time. Research suggests that even after controlling for socioeconomic status, minorities are less likely than whites to move to predominantly white neighborhoods and more likely to move to minority or racially mixed neighborhoods (South & Crowder, 1998; Stearns & Logan, 1986). These results indicate that merely increasing the effective affordability of vacant apartments of decent quality via a voucher is not enough to realize much improvement (on average) in the geographic outcomes for program participants relative to comparable renters who are not subsidized (Galster, 2013).

2.5.3 Low-Income Housing Tax Credit Program

Established by Congress in 1986, the Low-Income Housing Tax Credit Program (LIHTC) provides tax credits to developers for the construction and rehabilitation of affordable rental housing. Administered by the Internal Revenue Service (IRS) in partnership with state housing finance agencies (HFAs), the LIHTC has subsidized the production of over 2.2 million rental units between 1987 and 2010 (Khadduri et al., 2012). Subject to broad guidelines, each state develops criteria for awarding these tax credits and holds annual competitions among prospective developers for projects designed with a minimum share of "affordable" units (Galster, 2013). Unlike the public housing or housing vouchers that often serve extremely poor households, the LIHTC targets households with incomes below 50 or 60 per cent of area median income (AMI) (Deng, 2011). While it has been criticized for failing to reach households with the most serious housing needs, this higher-income eligibility also makes the program popular among both nonprofit and for-profit developers. However, studies evaluating the LIHTC program have revealed a more mixed picture. Freeman (2004) found that relative to other neighborhoods, LIHTC neighborhoods experienced larger declines in poverty and similar increases in home values, but Rosenthal (2008) argued that the concentration of LIHTC units lays the foundation for deterioration of a neighborhood's economic status in the next decade. On the other hand, (Green et al., 2002) found that LIHTC projects either increased nearby property value or had no impact, but they were less likely to generate a negative impact.

Oakley (2008) found that the LIHTC program is more successful than other programs at locating developments in less disadvantaged neighborhoods in terms of income level and minority concentration, but was not as successful at avoiding clustering. Abt Associates (2006) concluded that in large metropolitan areas, LIHTC units are likely to be located in high-growth tracts and in areas of increasing poverty, but that the majority of LIHTC units are located in moderate-poverty neighborhoods. These studies indicate a lack of consensus on whether the LIHTC program is successful in providing affordable housing in low poverty areas or with respect to how it impacts the characteristics of neighborhoods where it is built.

2.5.4 HOPE VI Program

Housing Opportunities for People Everywhere (HOPE) programs were initiated in 1994 with the aim of revitalizing "severely distressed" public housing sites (characterized by physical decay, high vacancies, drugs, gangs and violence) through locally developed PHA-private developer/financier partnerships (Galster, 2013). This program financed the demolition and rehabilitation of public housing units, the construction of new units on site, the temporary relocation of displaced tenants, and the provision of HCVs to displaced tenants who were unwilling or unable to return to the redeveloped sites. All totaled, HOPE VI demolished about 150,000 dilapidated public housing units in 224 different projects nationwide (Landis & McClure, 2010). The national HOPE VI tracking study found that after the first eight years of the program only 19% of original residents were living on the redeveloped sites, 29 % were in other public housing, 33% were using HCVs, and 18% had left housing assistance (Popkin et al., 2004). In summary, HOPE VI has had only minimal success in substantially increasing housing opportunities for former public housing residents in non-poor environments (Galster, 2013).

2.6 Location Choice and Hazard Exposure of Subsidized Housing

In light of the preceding discussion, the location of subsidized housing is clearly important for understanding the production of social vulnerability in urban areas. Housing policy has increasingly aimed to deconcentrate poverty and increase the employment opportunities available to low income households, but whether these programs have really been successful remains uncertain. Further, the degree to which subsidized housing has managed to avoid hazardous locations is another issue that has yet to be rigorously explored. The outcomes of residential location choice can vary based on the specific characteristics of local housing markets and program provisions (Pendall, 2000; Turner, 1998) and since the supply-side (i.e., scattered-site or LIHTC program) and demand-side (i.e., vouchers) policies have different mechanisms for providing housing assistance, their location outcomes can also be expected to vary (Kucheva, 2013). The location patterns of subsidized housing are discussed in the subsequent paragraph, and then prior studies on the hazard exposure of subsidized housing are briefly reviewed.

Studies have shown that the use of HCVs can be higher in disadvantaged neighborhoods because landlords in these areas often eagerly recruit HCV holders (Galster, 2013). Private landlords are more likely to be faced with high vacancies in these neighborhoods and respond by aggressively marketing their units to voucher holders (Galster et al., 1999). For the LIHTC, a

number of studies have documented the effect of a specific location incentive called the qualified census tract (QCT), which provides additional credit for construction in designated low-income census tracts (Baum-Snow & Marion, 2009; Freeman, 2004; Hollar & Usowki, 2007; Oakley, 2008). However, Lang (2012) showed that, even after controlling for QCT designations, differences in the market rent level affect the location of subsidized housing and specifically, developers are more likely to build subsidized housing in locations with low rent. Lang (2012) argues that subsidization is less likely to be profitable in locations with relatively high market rent because the opportunity cost of building subsidized housing is also higher in these locations. Burge (2011) shows that only a small portion of the cost of the LIHTC subsidy is used to reduce rent for tenants and rent savings diminish over the lifetime of the apartment units. The outcome Burge demonstrates may be the result of incentives to build subsidized housing in locations where the prevailing rent is already low.

Although previous research has thoroughly examined the location of subsidized housing in terms of the socio-economic status of a neighborhood, very few studies have explored the level of hazard exposure experienced by participants in these programs. Cutter et al. (2001) examined the relationship between the location of environmental risks and federally assisted public housing in a sample of eight medium-sized United States metropolitan areas. They found that families living in HUD housing had a greater risk potential from hazardous facilities based on proximity and the reported releases from them, and that minority populations (defined as percentage non-White) had significantly greater locational exposure than non-minority populations. Houston et al. (2013) assessed the spatial distribution of subsidized housing units provided through two federally supported, low-income housing programs in Orange County, California, in relation to neighborhood walkability, transit access, and traffic exposure. They argued that, since LIHTC development proposals receive points in a competitive process for access to local amenities, these developments may be more sensitive to site feasibility considerations and may tend to be located in transportation corridors with lower property values and higher traffic. On the other hand, since the HCV program is not location based and allows participants to locate within the private rental market, their neighborhoods should differ spatially from areas prioritized by developers leveraging capital through the LIHTC program. From their analysis, Houston et al. (2013) also found LIHTC projects more likely to be located in hightraffic areas than voucher users.

2.7 Study Framework

The analysis of social vulnerability generally focuses on social, economic, political, and institutional factors that lead to differential susceptibility or sensitivity (of different social groups) in the face of risk exposures (Tate et al., 2010). It is argued that the mounting financial, human, and environmental impacts in the United States are a function of the increasing movement of people and property into highly exposed areas (Cutter et al., 2007), however, very few studies have explored how social vulnerability changes over time across cities with different population growth trends. The first stage of this dissertation explores how the dimensions and spatial distribution of social vulnerability changed over a 30 year time period (1980-2010) in three coastal cities (Houston, New Orleans, and Tampa). This analysis documents and interprets observed similarities and differences in the dynamics of social vulnerability across the study cities. This study recognizes that multiple drivers contribute to this changing pattern of social vulnerability, but focuses on subsidized low-income housing programs as a potentially important, yet understudied factor ripe for further exploration. In its second stage, this dissertation delves deeper into the relationship between subsidized housing and hazard exposure in Houston, Texas. Under the vulnerability production framing (Dooling & Simon, 2012), it explores the extent to which the market-based programs for providing low-income housing are contributing to vulnerability within the city.

The spatial clustering of minorities and low-income households in neighborhoods vulnerable to hazardous and toxic materials is not uncommon in the United States (Massey & Denton, 1993; Mohai & Saha, 2006). Previous studies have proposed different but not mutually exclusive explanations for this including a lack of financial capacity (Galster & Keeney, 1988), discriminatory siting of hazardous facilities (Pastor et al., 2001), housing market discrimination (Dawkins, 2004; Freeman, 2004; Galster & Godfrey, 2005), and a lack of adequate knowledge about the risks (Zhang, 2010). On the other hand, studies of subsidized housing have also shown that these programs usually fail to deconcentrate poverty, in some cases create negative spillover effects on neighborhoods, and may even further concentrate poverty (Galster, 2013; Galster et al., 1999; Schill & Wachter, 1995). There is also evidence that market-oriented housing programs can actually incentivize low-income households to live in disadvantageous, low-rent neighborhoods (Freeman, 2004; Galster et al., 1999; Hollar & Usowki, 2007; Oakley, 2008). Since both natural and technological hazards exert a negative effect on residential property

values (Shultz & Fridgen, 2001; Speyrer & Ragas, 1991), it is reasonable to expect that subsidized housing programs may contribute to more low-income households living in those hazardous areas. Because minorities and low-income home owners usually do not have adequate knowledge about the risks associated with hazardous and toxic materials (Zhang, 2010), the probability that they may move into multi-hazard areas is even higher. However, it is also true that in some cases hazard exposure, in terms of distance to rivers and coastline, can also be perceived as an amenity (Smith et al., 2009) for the recreational opportunities and scenic views it may offer. As a result, such areas can serve as an important counterexample that defies the general trend (i.e., high hazard exposure coupled with low social vulnerability). These areas also can be expected to have fewer subsidized housing units due to higher land values. Ultimately, the above proposition should still hold that when located in hazardous areas, a higher proportion of subsidized units may be associated with a higher proportion of low-income minority households in the neighborhood.

The conceptual framework of this dissertation is presented in Figure 2.3 below. As shown, under the broader political-economic framing of vulnerability this dissertation examines the ways in which vulnerable populations may move into hazard exposed areas. The first stage of the study explores the changing pattern of social vulnerability (in three coastal cities) over time and in the second stage it delves into one case study city, hypothesizing that subsidized low-income housing programs may act as a key driver of social vulnerability in hazardous areas.



Figure 2.3: Conceptual framework of the dissertation

CHAPTER 3 RESEARCH METHODOLOGY

As discussed in the previous chapter, this dissertation is conducted in two stages—the first stage explores social vulnerability in three case study cities and the second stage examines the influence of subsidized housing programs on social vulnerability and hazard exposure in Houston, Texas. The methodology is discussed in detail here after a brief introduction to the case study cities.

3.1 Case Study Cities

For the first stage of the dissertation, similarity of geographic location and variation in population trends were the primary criteria for selecting case study cities. All three case study cities are located in the U.S. Gulf Coast region, but within different state jurisdictions—Houston in Texas, New Orleans in Louisiana, and Tampa in Florida. All of these cities are considered highly exposed to climate change impacts (Nicholls et al., 2008), but have varying levels of population and assets exposure (Table 3.1).

Case study cities	Population growth (2000-10)	Current climate exposed pop. (thousands)*	Current climate exposed assets (billion \$)*
Houston, TX	7.46%	59	12.21
(Harris County)			
New Orleans, LA	-29.06%	1,124	233.69
(Orleans Parish)			
Tampa, FL	10.63%	415	86.26
(Hillsborough County)			

Table 3.1: Case Study Cities

* Estimates from Nicholls et al. (2008), Appendix 3: City Data and Rankings

Since the boundary of the cities has changed over time, their respective counties⁶ are used to spatially delineate the study cities for this research. Figure 3.1 shows different trajectories of population change over time in each of the three study areas. While Houston has experienced consistent growth since 1960, Tampa was characterized by relatively moderate growth, and New Orleans experienced consistent decline.

⁶ Counties encompassing the core cities are considered for this study to maintain a consistent geographic area over time. Metropolitan areas not considered since MSA boundaries have also changed over time and outlying rural areas may significantly influence the results of this study.



Figure 3.1: Population trends of case study cities

For the second stage of the dissertation, Houston (i.e., Harris County) serves as a detailed case study for examining the longitudinal relationship between subsidized low income housing and hazard exposure. The distinctly neoliberal approach to policy making that characterizes Texas also makes Houston the most suitable choice, given the political-economic framing of this study. Since the city boundary of Houston has changed, Harris County is adopted as the case study unit of analysis in order to maintain a consistent geographical extent over time. This area is also a prime candidate for the dissertation due to its multi-hazard environment. Harris County is the third most populous county in the United States⁷ with about 4 million residents⁸ and is located in the low-lying Texas coastal area that is exposed to both hurricane and flood hazards (Cutter, 2003). Besides natural hazards, there are also hundreds of petrochemical manufacturing and distribution facilities across the county, which elevates the potential for toxic chemical hazards (EPA, 2008). Harris County is also demographically and socioeconomically diverse. According to 2012 American Community Survey data, the Hispanic and Latino population accounted for 41.5 percent of the total

⁷ Both in the 2000 and 2010 Decennial Census.

⁸ Total population of Harris County was 3,400,578 in 2000, which increased to 4,092,459 in 2010, as per the 2000 and 2010 Decennial Census respectively.

population, with 17.3 percent of population below poverty level in this county. This diversity further recommends Harris County and the City of Houston as an interesting case for evaluating the location decisions of vulnerable low-income population groups in this urban agglomeration and how existing subsidized housing programs are influencing this vulnerability pattern.

3.2 Evaluating the Changing Patterns of Social Vulnerability

A variety of methods have been employed to measure social vulnerability at many different scales, but the one constant has been a consensus on the multidimensionality of this concept. For example, vulnerability is often understood to include poverty (Fothergill & Peek, 2004; Long, 2007), race and ethnicity (Fothergill et al., 1999; Peacock et al., 1997), gender (Enarson & Morrow, 1998; Enarson et al., 2007), and age (Anderson, 2005; Smith et al., 2009). In order to evaluate social vulnerability dynamics in the study cities (Research Question 1), the Social Vulnerability Index (SoVI) is calculated decennially at the census tract level over a 30 year time period (1980-2010). This was accomplished using the inductive approach for measuring social vulnerability developed by (Cutter et al., 2003) with some modifications considering data availability and suggested refinements to this foundational method by subsequent studies (Finch et al., 2010; Tate, 2012). Cutter et al. (2003) considered 42 variables for their study at the county level, but since this dissertation measures social vulnerability at the census tract level and focuses more on the social construction of vulnerability, it considers 26 variables, following the approach of (Finch et al., 2010). The list of variables used for SoVI calculation is presented in Table 4.1 of Chapter 4. For 2010, these variables were collected from U.S. Census American Community Survey (ACS) 2008-2012, and for 1980 to 2000, data were drawn from the decadal census reports of U.S. Census Bureau⁹. Since the ACS data are based on sample surveys over a five-year period, 2008-2012 is considered representative data for 2010. The boundaries of census tracts also changed significantly in 2010, which makes it very difficult to construct a longitudinal database of the type required here. For the purposes of longitudinal comparison, all the data (1980 to 2010) were converted to 2010 census geography following the Longitudinal Tract Data Base (LTDB) approach developed by Logan et al. (2014). Census tracts with high group quarters populations (e.g. jails, university campuses, etc.) or too few residents in any time period were excluded from the analysis. These collected data were then normalized to

⁹ Census reports of 1980 to 2000 are collected from Social Explorer (http://www.socialexplorer.com/).

percent, per capita, or density functions given the differential population and area of census tracts. Each of the variables is standardized as a z-score¹⁰ before being incorporated into the Principal Components Analysis (PCA) so that no individual variable influences the final vulnerability score more than another. Before applying PCA to calculate the SoVI, biplots with alpha-bags (La Grange et al., 2009) were created to explore distributions of data over the years and to visualize the changing patterns of the various dimensions of social vulnerability.

3.3. Multidimensional Biplots for Exploring Social Vulnerability

The distribution of selected vulnerability indicators and their relationship with each other can be visualized through two-dimensional scatter-plots, but such plots can only present two variables and cannot show how all the variables are distributed in relation to each other. Biplots were introduced by Gabriel (1971) as a graphical display tool consisting of a vector for each row and a vector for each column of a matrix of rank two. Here the prefix "bi-" refers to the simultaneous display of both the rows and columns of a data matrix and not to the dimension of the display space. It is also described as a multivariate analogue of the ordinary scatter plot (Gower & Hand, 1996). While a scatter-plot portrays the covariation of two variables, a biplot can contain as many axes as there are variables so as to provide information on all variables in a single plot. Biplots represent different variables by axes similar to ordinary scatter-plots and calibrate these axes in the original scales of measurement (Gardner et al., 2005). Here the individual observations are represented as points, and variables are represented as labeled, calibrated axes (La Grange et al., 2009). To accommodate more than two variables, the axes of biplots are not perpendicular as in ordinary scatter-plots, but they are still used in a similar way to provide information on all variables. As a multivariate extension of an ordinary scatter plot, non-statisticians should comfortably understand the basics of the biplot, enabling them to interpret the biplot display with relative ease (Walters & Roux, 2008).

Since this dissertation explores how different dimensions of vulnerability are distributed over time, correlation biplots are created for each of the time periods (and for each of the study cities) separately. The biplot follows an ordinary eigenvalue¹¹ problem and aims to optimally

 $^{^{10}}$ The z-score was calculated as follows: $Z_{Tract} = (Score_{Tract} - Mean_{Metro}) / Standard Deviation_{Metro})$

¹¹ In Principal Component Analysis (PCA) the variance for each principal component is given by the eigenvalue of the corresponding eigenvector. The eigenvalue of each component indicates the percentage of variation in the total data explained by this component. For more discussion on eigenvectors and eigenvalues see Manly (1986).

represent the multidimensional variation of the observations in two dimensions (Gardner et al., 2005). Like ordinary scatter plots, biplot axes do not intersect perpendicularly. In this case angles among the axes indicate correlations among the variables, where a *smaller angle* refers to a *larger correlation* (Le Roux & Gardner, 2005). Such a biplot can be overlaid with an alpha-bag (Gardner, 2001) to better present and aid interpretation of a large number of variables. An alpha-bag is a contour enclosing the exact innermost α percentage of sample points in a biplot (Gardner et al., 2005). The distribution of the selected variables within the alpha-bags indicates how different dimensions of vulnerability are related and comparing alpha-bags of different time periods indicate how this relationship changes over time. Chapter 4 further elaborates on biplots and shows how the dimensions of social vulnerability have changed over time for each of the study cities.

3.4 Calculation of SoVI and Exploring Its Distribution

Biplots with alpha-bags (for all four census years) visually indicate the changing pattern of vulnerability dimensions, but for quantifying the relative levels of vulnerability, the Social Vulnerability Index (SoVI) is calculated at the census tract level. Since the SoVI algorithm of Cutter et al. (2003) has been found to be robust enough to withstand minor changes in variable composition and scale (Schmidtlein et al., 2008) and precision (Tate, 2012), this dissertation adapts this approach to measure social vulnerability at the census tract level. The SoVI is derived through Principal Component Analysis (PCA) which identifies a smaller set of independent factors that account for a majority of the overall variance within the original data. These component parts are then interpreted and assigned a general socioeconomic or demographic interpretation based on which factors loaded highest on each component. One of the key steps in this process is to choose the optimum number of factors to retain from the principal component analysis. Cutter et al. (2003) applied the Kaiser criterion¹² for selecting the number of factors to retain (O'Connor, 2000; Patil et al., 2008). Considering this limitation of the original approach of Cutter et al. (2003), this dissertation applies parallel analysis¹³ for determining the optimum

¹²In which all components with an eigenvalue greater than one are retained.

¹³ The rationale behind Parallel Analysis is that sampling variability may produce an eigenvalue greater than one even if all eigenvalues of a correlation matrix are exactly one and no large components exist (Zwick & Velicer, 1986). Parallel analysis applies principal components analysis to a random matrix of identical dimensionality and compares it to the research data set. Through a Monte Carlo simulation approach it produces a distribution of

number of factors as suggested by (Tate, 2012).

The aim of applying PCA to the standardized variables is to identify a certain number of factors that explain most of the variation in the dataset. These factors are interpreted and named based on the characteristics of the variables that are most closely associated with it and typically informed by one or more dominant variables. All factors are rescaled so that positive values indicated higher levels of vulnerability, while negative values are consistent with a decrease in vulnerability. The factor scores are then aggregated in an additive model to derive the overall composite social vulnerability score. The additive model assigns equal weight to all components, which makes no a priori assumptions about the relative importance of each component in producing social vulnerability (Cutter et al., 2003; Finch et al., 2010). This process of SoVI calculation is applied to each of the study areas (Houston, New Orleans, and Tampa) and for all time periods (1980, 1990, 2000, and 2010) separately. To determine the patterns of similarity and dissimilarity in the clustering of social vulnerability, the degree of spatial autocorrelation among the census tracts is evaluated. This portion of the analysis measures and visually depicts the spatial pattern of social vulnerability in each of the different study city contexts and how it has changed over time. Local Indicators of Spatial Association (LISA) cluster analysis (Anselin, 1995) helps to identify the areas with significantly higher concentrations of vulnerable populations.

Comparing the z-scores of the SoVI at the tract level shows which areas of the city (with respect to other parts of the city) have experienced a significant increase or decrease in vulnerability over time. The degree of spatial autocorrelation of the SoVI evaluates the pattern of change in social vulnerability throughout the study cities and how it varies in different city contexts. Chapter 4 discusses further on the statistical measures applied to compare SoVI and presents the results for all three study cities.

3.5 Subsidized Low-Income Housing as Driver of Social Vulnerability

This second stage of the study is conducted in two steps. First, the level of exposure to both natural (flood and hurricane) and technological hazards are evaluated for the entirety of

eigenvalues for each principal component in the random dataset. Simulated eigenvalue distributions are compared to the observed eigenvalues and whatever number of eigenvalues in the observed dataset exceeds the corresponding value in the simulated dataset is considered as the optimal number of factors to retain. This procedure usually results in a fewer number of factors (compared to Kaiser criterion) and considered as a superior alternative for determining the optimal number of factors (Tate, 2012).

Harris County, Texas, and population trends (for different vulnerable groups) within these areas are explored (**Research Question 2**). In the second step, spatial econometric models are applied to evaluate the extent to which subsidized low-income housing is influencing the hazard exposure of vulnerable population groups in Harris County (**Research Question 3**). Given the availability of data on subsidized housing, this part of the study was conducted for the period of 2000 to 2010.

3.5.1 Hazard Exposure and Social Vulnerability

In order to capture the multi-hazard landscape of Harris County, both natural and technological hazards are included in this study. Given the high risk of floods and hurricanes in Harris County area, these two hazards are considered under the domain of natural hazards. For technological hazards, exposure through the facilities identified in the Toxic Release Inventory (TRI) of EPA is considered due to the prevalence of such industries in the low-lying areas of Houston. The impact areas for flooding and hurricane storm surge are identified based on publicly available outputs of computer modeling. The U.S. National Hurricane Center used the SLOSH (Sea, Land, and Overland Surge from Hurricanes) model to estimate worst-case-scenario storm-surge boundaries (Tate et al., 2010) and produced surge zones for each of the Saffir-Simpson categories. GIS floodplain-boundary representations of these model outputs are available from FEMA Digital Flood Insurance Rate Map (DFIRM) database. Category-1 hurricane risk zones and 100-year floodplains were identified from these sources and combined to identify and map the natural hazard exposed areas of Harris County. The proportion of residential lots falling within these areas was used as the indicator for determining the relative natural hazard exposure level of each census tract. This approach is similar to Cadastral-based Expert Dasymetric System (CEDS) methodology of Maantay et al. (2007) that considers residential area or the number of residential units as proxies for population distribution. This technique gives a more realistic population distribution compared to traditional mapping methods and is particularly useful for evaluating flood exposure in an urban environment (Maantay & Maroko, 2009a). Parcel information for Harris County was used in this study as the ancillary data required for the CEDS approach. These data were collected from the Harris County Appraisal District (HCAD) public downloads FTP site and the locations of residential lots were extracted. ArcGIS tools were used to combine the flood and hurricane risk maps, clip the residential lots located within risk areas, then intersect these layers with census tract layers to

calculate the percentage of residential lots within the natural hazard areas of each census tract. Areas with a higher percentage of its residential lots falling in the flood or hurricane risk areas are considered to be more exposed to natural hazards.

For evaluating technological hazards, environmental pollution data collected by the United States Environmental Protection Agency's Toxic Release Inventory (TRI) were used for this study. The TRI is one of the most widely used data sources for environmental justice studies. Any facility that releases toxic chemicals is required to submit reports to the Environmental Protection Agency if certain minimum release thresholds and other criteria are met as mandated by the Emergency Planning and Community Right-to-Know Act (EPCRA) of 1986. The locations of Toxic Release Inventory (TRI) facilities in Harris County were obtained by querying the TRI.NET application of the US Environmental Protection Agency (EPA). This application allows users to select, sort, and filter TRI data by geography, year, and hazard type. The location and total toxic release data of all facilities in Harris County for all years from 2000 to 2010 were collected to evaluate the relative hazardousness of the county's census tracts. All the locations were checked after mapping them by both geocoding addresses and plotting latitude and longitude coordinates. Since the release levels may vary from year to year for a certain facility, data for all years were considered in calculating the average level of hazardousness.

An effective method for defining proximity and potential exposure to environmental hazards remains a widely debated issue in the environmental justice literature (Chakraborty & Maantay, 2011; Conley, 2011). Since this study is interested in proximity to hazards and its relative influence on social vulnerability, exposure based on distance decay or continuous distance from the hazard was deemed the best approach. But here as well there is no recognized standard for what the threshold distance of exposure from a facility should be or how the distance decay function should be defined. Considering this methodological uncertainty, two approaches were taken here that have contrasting distance decay specifications and can accommodate multiple threshold distances. The first approach uses a power function (Conley, 2011) and the second approach considers cumulative exposure (Cutter et al., 2001) within a specified threshold distance. For both of these approaches, the total release from a given TRI facility is used as an indicator of its relative level of influence without considering the toxicity level of the chemicals released by them. It can be assumed that the total release will be commensurate with facility size and hence, a better indicator of relative influence on its

surrounding region. The two approaches taken for evaluating technological exposure are explained below.

The first approach uses the power equation, taken from the physical model of gravity, and considers that the impact of a facility is proportional to the size of the release and inversely proportional to the distance (from a census tract centroid) raised to a parameterized component (Conley, 2011). The equation can be shown as:

$$y_i = \sum_{j}^{\#Facilities} \frac{w_j^{\theta}}{d_{ij}^{\theta}}$$

Here y_i is the weighted cumulative exposure of census tract *i*, w_j is the total release from facility *j* and d_{ij} is the distance from facility *j* to tract *i*. In this equation θ is a positive constant that determines the rate of decrease to which a facility may impact its surrounding region. To operationalize a situation where a facility exerts a very high impact on its immediate neighboring area and then its influence decreases rapidly with distance, θ is assigned a value of 2 in this study (Figure 3.4). An additional reason for this specification is that no threshold distance or cutoff point for facility impact (which is considered in the second approach) is assumed and hence, a given facility will still have a minimal impact on far away tracts.

In the second approach, risk exposure at the census tract level was measured based on the potential exposure model of Cutter et al. (2001). Like the first approach, it is also based on the distance to existing hazard source and can incorporate multiple hazard sources, but has the option to impose a threshold distance or cutoff point on facility impact. The index measured through this model is labeled as *proximal exposure* as it does not purport to represent actual exposure. This proximal exposure model includes multiple hazard sources in order to compute a *cumulative proximal exposure* (CPE) for a census tract Cutter et al. (2001). Cumulative proximal exposure is defined as the sum of proximal exposure associated with each tract:

$$WCPE_{i} = \sum_{j}^{\#Facilities} w_{j} \left(1.0 - \frac{d_{ij}^{\theta}}{T} \right)$$

Here, $WCPE_i$ is the weighted cumulative proximal exposure to population in census tract *i* from distance to facility at locations 1 through *j* (total number of facilities), w_j is the total release from facility *j*, d_{ij} is the distance from tract *i* to facility *j*, *T* is the threshold distance at which exposure becomes negligible from a facility, and θ is the rate of reduction of exposure at increasing
distance from *j*. Unlike the power function of the first model, this model assumes a distance, *T*, at which the proximal exposure becomes zero. This point is called the "negligible threshold" beyond which there is no locational risk. In this case the impact from a facility to its neighboring region decreases much more slowly (compared to power function), but does not assume any impact beyond the threshold distance (Figure 3.2). For the present analysis, θ is assigned a value of 2 (following Cutter et al. (2001)) and the negligible threshold distances (i.e., sensitivity analysis). All the results presented in Chapter 5 were tested for threshold distances of 1, 3, and 5 miles¹⁴, but only the results from conservative estimate of 1 mile are presented after finding that it yielded the most consistent results.



Figure 3.2: Comparison of distance decay functions. (For both functions $\theta = 2$, w=1000, and for CPE T=5 is considered to show a hypothetical TRI facility may impact its nearby region.)

The two approaches discussed above produce two different hazard landscapes for the study area and hence, should provide a deeper understanding of the dynamics between subsidized housing and social vulnerability in the study area. For both of the approaches,

¹⁴ Since there is no widely accepted threshold distance, this study tested the results in these three distance bands of 1,3, and 5 miles following the approach of Cutter et al. (2001) who also evaluated their results on 1 to 5 miles distance range. While Cutter et al. (2001) tested it for census tract level analysis, a subsequent study of Conley (2011) applied it for County level data with much larger distance threshold (5, 10, 15,....500 miles).

technological exposures at the census tract level were calculated for all years from 2000 to 2010 and then the average level of exposure was calculated, which represents the relative hazardousness of an area. Population trends, for different dimensions of social vulnerability (i.e. minority, low-income, etc.) were explored within census tracts having different levels of exposure to natural or technological hazards.

3.5.2 Low-Income Housing in a Multi-Hazard Environment

Data on the location and type of subsidized housing from the Department of Housing and Urban Development (HUD) were used for this study. HUD maintains the LIHTC Database and the *Picture of Subsidized Households* data set, which disaggregates building-based public and private "fixed" subsidies (Public and Indian Housing, Section 236, Section 8 New Construction and Substantial Rehabilitation, FHA, LIHTC) from person-based "voucher" subsidies (Section 8 vouchers and certificates). These data were further verified against information collected from the Houston Housing Authority (HHA) and Texas Department of Housing and Community Affairs (TDHCA) to check the accuracy of the location and number of subsidized housing units. After geocoding the locations of the housing units, the total number of housing units (by type) in each of the census tracts was calculated. Considering variation in the coverage and approach of different housing assistance programs, this research focused on two programs that are among the most popular and widely used (Schwartz, 2010)-Section 8 or the Housing Choice Voucher (HCV) program and the Low Income Housing Tax Credit (LIHTC) program. These two programs also represent contrasting approaches for providing housing assistance, the supply-side (LIHTC program) and the demand-side (vouchers). However, HUD's Picture of Subsidized Households data set is consistently available only beginning in 1996 and to make it comparable with census data, 2000 and 2010 were chosen as the two time periods for HCV data. For the LIHTC data (collected from HUD's LIHTC Database), the locations of the properties were geocoded, then the year when a project was placed in service was extracted and used to assign them to either 2000 or 2010 (i.e. all active projects built before 2000 are assigned to year 2000, and all projects developed after 2000 are assigned to year 2010). For all of the LIHTC properties in a census tract the number of low income units available were aggregated to obtain total number of subsidized households in that tract attributable to this program.

Before applying regression analysis to the social vulnerability and housing subsidy dataset, some of the key variables were explored through descriptive statistics. Principal Component Analysis (PCA), applied for developing the SoVI, revealed the key indicators that explain most of the variation in the original data, which were race and poverty. Whether an increase in housing subsidies in a hazardous area contributes to social vulnerability there (e.g., higher poverty) compared to other areas which did not experience an increase in housing subsidies between 2000 and 2010 is a question that emerged from this initial exploration of the data. As hypothesized here, we should see a significant increase in vulnerable populations in an area (compared to other areas) when it receives increased number of housing subsidy and is also located in a hazardous area. While prior studies have extensively explored the neighborhood impacts of housing subsidy and found mixed results (Freeman, 2003; Galster et al., 1999; Holloway et al., 1998; Oakley, 2008), this kind of interaction between hazard exposure and housing subsidies, alongside their combined effects on neighborhood social vulnerability is yet to be explored.

3.5.3 Model Selection and Analysis

Both standard OLS regression and spatial regression models were estimated to evaluate the above mentioned hypothesis, which necessitates controlling for base year neighborhood socio-economic characteristics along with changes in these characteristics in the 10 year period between 2000 and 2010. Particularly, considering the spatial pattern of vulnerability concentration and diffusion (as presented in Chapter 4), a spatial Durbin model (LeSage & Pace, 2009) was estimated to evaluate the extent to which subsidized housing may have contributed to the location of vulnerable populations in hazard exposed areas. Here the dependent variables are the number of residents exhibiting key dimensions (determined from the SoVI analysis) of social vulnerability (e.g., African-American population in poverty, Hispanic population in poverty, etc.) while four interaction variables (e.g., two hazard types by two subsidy types) are the key independent variables for this model. All other variables related to social vulnerability and hazard exposure were treated as control variables. The rationale for estimating a spatial Durbin model (SDM) is due to the patterns of spatial dependence among social vulnerability, hazard exposure, and housing subsidy observed by previous studies. First, prior studies have shown significant clustering of minorities and low-income households in neighborhoods vulnerable to hazardous and toxic materials (Bullard, 1999; Massey & Denton, 1993; Mohai & Saha, 2006). Second, subsidized low-income housing has been found to be geographically concentrated in

certain areas (Oakley, 2008), and sometimes highly clustered in areas characterized by high poverty rates, minority concentrations, and poor educational opportunities (Van Zandt & Mhatre, 2009). The provision of incentives through qualified census tracts (QCTs) for the LIHTC program was also found to be contributing to higher numbers of low-income housing in highpoverty areas (Abt Associates, 2006; Hollar & Usowki, 2007). Thirdly, studies have found negative spill-over effects of subsidized housing on land value, crime, middle class flight, and concentration of poverty (Galster et al., 1999; Holloway et al., 1998; Massey & Kanaiaupuni, 1993). In light of this evidence, it can be expected that subsidized housing and poverty concentration are not only related within a neighborhood, but also influencing surrounding neighborhoods, and to capture the spill-over effects (if there are any) the SDM should be a better modeling approach. An unconstrained SDM is a general-to-specific approach which can consider such indirect spatial interactions as when the exogenous explanatory variables influence not only the dependent variable in an area, but also in neighboring areas as well (LeSage & Pace, 2009).

The most commonly encountered specification in spatial econometrics is the spatial lag model (Anselin, 1988; LeSage & Pace, 2009; Ord, 1975):

$$y = \rho W y + X \beta + \varepsilon \tag{1}$$

where y is a vector of observations on a dependent variable taken at each locations, X is a matrix of exogenous variables, W is row-standardized n by n spatial weight matrix, β is a vector of parameters, ε is a vector of independent and identically distributed disturbances and ρ is a scalar spatial lag parameter. The spatial error model can be written as (Anselin, 1988; LeSage & Pace, 2009; Ord, 1975):

$$y = X\beta + u \tag{2}$$
$$u = \lambda W u + \varepsilon \tag{3}$$

Here λ is the spatial autoregression parameter of error term *u*. While spatial lag model (1) treats spatial correlation as a process or effect of interest and incorporates spatial dependence (*Wy*), the spatial error model treats spatial correlation as a nuisance and examines whether the error term has a spatial dependency (*Wu*). The spatial Durbin model, on the other hand, includes a spatial lag of the dependent variable as well as spatial lags of the explanatory variables:

$$y = \rho W y + X\beta + W X\theta + \varepsilon \tag{4}$$

where θ is a vector of parameters. Here ρ is the spatial autoregressive parameter that measures the degree of spatial dependence between the numbers of vulnerable population of nearby census

tracts. The WXθ term in Eq. (4) captures the extent to which the characteristics of neighboring areas (for hazard exposure and housing subsidy) influence the size of vulnerable populations in an area and specifically captures the spill-over effect of subsidized housing shown by previous studies (Galster, 2013; Galster et al., 1999; Schill & Wachter, 1995). The Wy term is expected to capture the clustering of minority and low-income populations in certain areas of the city and the SDM is solved using maximum likelihood estimation (Bivand & Gebhardt, 2000; LeSage & Pace, 2009).

Before applying the spatial Durbin model, Elhorst (2010) suggests that other modeling alternatives be explored to find out which model is the most likely candidate to explain the data and test hypothesized relationships. LeSage & Pace (2009) advocate for use of the SDM model as the model to test for spatial interaction effects for two main reasons. First, in the case that the true spatial process is one in which spatial dependence exists between the *dependent variable in* one area and the exogenous covariates in neighboring areas, and these variables happen to be correlated with independent variables not omitted from the model, the SDM will produce unbiased coefficient estimates (but a spatial lag model will not). This is considered to be a salient issue for this study, since vulnerability indicators (poverty, minority populations etc.) tend to be clustered in space, thus implying correlation between the values in neighboring tracts. Second, when the true spatial process is one in which the outcome variable is spatially correlated with the exogenous covariates within the same tract or in which there exists a spatially correlated error term (i.e. spatial error model), the SDM model will continue to produce unbiased coefficient estimates¹⁵. So it can be inferred that, except under circumstances in which the spatial lag or error model is an unequivocally better fit to the underlying spatial process, the SDM model is the preferred option. It should also be noted that in recent simulation experiments conducted by Beer & Riedl (2012), the spatial Durbin model also outperformed a structural equations model (SEM) specification.

Elhorst (2010) proposed a procedure to determine which model, among OLS, spatial lag, SEM, and SDM is the most likely candidate to explain the data. In this procedure, first an OLS model is tested to check if a spatial lag or spatial error model is more appropriate for describing the data. The classic LM-tests (Anselin, 1988) and the robust LM-tests (Anselin et al., 1996) can

¹⁵ Spatial dependence in the error term modeled in the spatial error model is usually referred to as nuisance dependence (Anselin, 2003). The spatial error model usually loses its popularity given its lack of interpretation of spatial dependence and, more importantly, its susceptibility to omitted variables (Brown et al., 2009).

be used for this purpose. If the OLS model is rejected in favor of a spatial lag specification, the spatial error model, or in favor of both models (as indicated by LM tests), then the spatial Durbin model (SDM) should be estimated. If these models are estimated by maximum likelihood, Elhorst (2010) suggested using a likelihood ratio (LR) test to evaluate the hypothesis H_0 : $\theta = 0$ and H_0 : $\theta + \rho\beta = 0$. The first hypothesis examines whether SDM can be simplified to spatial lag model (Eq. 1) and if it cannot be rejected, then the spatial lag model best describes the data (provided that the robust LM also pointed to the spatial error model (Eq. 2, if $\theta = -\rho\beta$ then $\lambda = \rho$), and if it cannot be rejected, then the spatial error model best describes the data (provided that the robust LM also pointed to the spatial error model best describes the data (provided that the robust LM also pointed to the spatial error model best describes the data (provided that the robust LM also pointed to the spatial error model best describes the data (provided that the robust LM also pointed to the spatial error model best describes the data (provided that the robust LM also pointed to the spatial error model best describes the data (provided that the robust LM also pointed to the spatial error model best describes the data (provided that the robust LM also pointed to the spatial error model). If none of these conditions are satisfied, then SDM should be adopted, because it generalizes both the spatial lag and spatial error models (Elhorst, 2010).

In case the estimated OLS model is not rejected in favor of both the spatial lag and the spatial error model, Elhorst (2010) suggested reestimating the OLS model with spatially lagged independent variables (*WX*). This approach allows testing of the hypothesis H₀: $\theta = 0$, and if this hypothesis also cannot be rejected, then the OLS model should be adopted. On the other hand, if this hypothesis is rejected, then there should be another iteration applying the SDM and testing the additional hypothesis H₀: $\rho = 0$. If this hypothesis is also rejected, then SDM should be considered; otherwise, the spatial lag model should be considered as the best description of the data¹⁶. This procedure proposed by Elhorst (2010) was adopted for this study to identify statistical models describing the changing patterns of social vulnerability in Harris County, Texas. Chapter 5 discusses these models and their findings in greater detail.

3.6 Analysis of Policy and Planning Practice for Subsidized Housing

Findings from the spatial analysis described above indicates the extent to which subsidized low-income housings are clustered in Houston and how they are contributing to the concentration of vulnerable populations in hazardous areas. In this last stage of the dissertation, policies and planning approaches for locating subsidized housing are evaluated and suggestions are offered for making them more consistent with climate change adaptation and disaster risk reduction efforts (**Research Question 4**). HUD regulations for locating subsidized housing in a

¹⁶ For elaborate discussion on model comparison procedure and more statistical background, see Elhorst (2010)

multi-hazard environment were reviewed to compare those provisions with the HCV and LIHTC programs. Housing officials with the Houston Housing Authority were interviewed to gauge awareness of the prevailing issue of environmental justice in the study area and what factors are considered when permitting (or issuing tax credits) for construction of any subsidized housing. Since prior studies have found significant variation in location outcomes (of subsidized housing) depending on local housing markets and planning approaches (Galster, 2013; Pendall, 2000), it is necessary to evaluate how these aspects of Houston may have influenced the observed distribution of subsidized housing. A variety of documents on housing subsidized housing is made and which factors have guided the decision making process. Short visits to some of the LIHTC projects in Houston were made to see first-hand how they are situated with respect to hazard sources and to get a better sense of their surrounding neighborhoods. The findings from the interviews, document analysis, and field visits helped to highlight the limitations of present approaches for affordable housing and to identify opportunities for subsidized housing to become more consistent with climate change adaptation and disaster risk reduction efforts.

CHAPTER 4 CHANGING PATTERNS OF SOCIAL VULNERABILITY IN COASTAL CITIES

4.1 Theoretical Framework

Planning for climate adaptation tends to overemphasize future climate variability, which can lead to non-adaptive or even mal-adaptive outcomes (Barnett & O'Neill, 2010; Macintosh, 2013; Orlove, 2009). This approach runs the risk of focusing solely on identifying the climatic hazards of a place rather than exploring why people are located in hazard exposed areas in the first place. Recent research indicates that in addition to hazard exposure and physical vulnerability, social vulnerability, or the variation in characteristics such as income, race/ethnicity, gender, and household composition (among others), matters when predicting the impacts of coastal surge and flooding (Highfield et al., 2014). Adaptation approaches that ignore the social dynamics of a city can create an environment for non-adaptation when, despite efforts to climate-proof places, people are pushed into harm's way. This becomes more problematic in a multi-hazard urban environment where zoning restrictions or flood-proofing policies may further the climate adaptation of certain places, but the socially vulnerable population may only find it affordable to live in places which are exposed. The intersection of adaptation and social vulnerability demands a new approach to land use planning that is more responsive to how socially vulnerable people find their places within a multi-hazard urban area.

Although theories of neighborhood change and social vulnerability represent two distinct literatures, when considered jointly they should yield important insights into the vulnerability dynamics of a city. While neighborhood change theories explain the underlying forces that drive shifts in the demographic composition of neighborhoods (Glaeser et al., 2008; Peterman, 2000; Temkin & Rohe, 1996), social vulnerability theories explore how factors such as poverty, race, age, migration, etc. reduce the capacity of marginalized population groups to withstand hazard events or delay their recovery process (Blaikie et al., 1994; Hewitt, 1997; Peacock et al., 1997). This chapter integrates these two theoretical frameworks in order to explore changing patterns of social vulnerability in three coastal cities (Houston, New Orleans, and Tampa). Recognizing the historical path dependency of social vulnerability, this chapter analyzes three decades of data (1980-2010) to understand how the different dimensions of vulnerability interacted with each

other over time and to identify emerging dimensions of vulnerability that warrant attention in future adaptation efforts. Since city boundaries often change significantly over time, the respective counties of the three coastal cities are taken as the study units to maintain geographic consistency. Houston is located in Harris County, Texas; New Orleans in Orleans Parish, Louisiana, and Tampa in Hillsborough County, Florida. As mentioned in Chapter 3, despite being located in the same Gulf Coast region, these cities are experiencing drastically different population trends, and also have different planning and policy approaches. One commonality among them is that all three cities are considered to be highly exposed to projected climate change impacts (Nicholls et al., 2008). Studying the changing pattern of social vulnerability within and across these cities should yield important insights for adaptation planning efforts in the region and beyond. The next section further discusses neighborhood change theories and how integrating those with social vulnerability theories will further the aims of this study. Changing patterns of vulnerability are explored through two different approaches, first by multidimensional biplots and then by measuring the Social Vulnerability Index (Cutter et al. 2003) and its spatial dimensions over time. Similarities and dissimilarities in longitudinal patterns of social vulnerability among the cities are identified and explained through the lens of neighborhood change theories.

4.2 Social Vulnerability and Neighborhood Change

Social vulnerability theorists contend that vulnerability is a social condition, a measure of societal resistance and resilience to hazards (Blaikie et al., 1994; Hewitt, 1997). In other words, this perspective emphasizes the socioeconomic characteristics that influence a community's ability to prepare for, respond to, cope with, and recover from a hazard event (Cutter et al., 2003; Laska & Morrow, 2006; Peacock et al., 1997) and is most often described using individual characteristics (e.g., age, race, health, income, type of dwelling unit, employment). This approach situates disasters and their impacts within broader social contexts and processes (Wisner et al., 2004) and emphasizes social factors that influence or shape the susceptibility of various groups to harm (Cutter et al., 2003). Studies of social vulnerability have extensively documented the disproportionate impacts of hazards events on socially vulnerable population groups (Cutter, 1996; Cutter et al., 2003; Fothergill & Peek, 2004; Highfield et al., 2014; Peacock et al., 2007; Zahran et al., 2008). Prior research has also shown that minority and low-

income households usually have a lower level of disaster preparedness (Mileti & Darlington, 1997; Peacock, 2003; Russell et al., 1995), are less likely to hold earthquake or flood insurance instruments (Blanchard-Boehm, 1998; Fothergill, 2004), and are less likely to receive and believe official disaster warnings (Fothergill & Peek, 2004; Perry & Mushkatel, 1986; Perry & Nelson, 1991). Although these studies contribute important insights to better understanding the differential outcomes of disasters, they typically fail to explore the generative process of social vulnerability in a place—specifically, why vulnerable population groups move into hazardous areas and how present patterns of social vulnerability have evolved over time. Within the framework of social vulnerability (Eriksen & Kelly, 2007), but these studies are also limited in their scope by evaluating only cross-sectional variation. There have been attempts to explore longitudinal change in social vulnerability (Cutter & Finch, 2008), but very few studies (if any) have done so at the urban neighborhood level or attempted to explain the drivers for the changing geography and composition (of different dimensions) of social vulnerability.

Neighborhood change theorists have long investigated the underlying drivers of shifts in the social characteristics of urban neighborhoods and this literature also offers significant insights into the changing pattern of social vulnerability within the cities. While social vulnerability explores *what* makes people vulnerable to hazards, neighborhood change theories explain, among other things, why vulnerable people move to or concentrate in certain areas of a city. Integrating these two theoretical perspectives allows for a broader and more realistic framing of the changing patterns of social vulnerability, which is currently lacking in the vulnerability science literature. Neighborhood change theories explain the macro- and microlevel socioeconomic, political, and institutional forces that drive changes in neighborhood characteristics (Temkin & Rohe, 1996) and emphasize understanding the dynamics of neighborhood change to fully grasp the implications for urban planning and policy (Glaeser et al., 2008; Li & Morrow-Jones, 2010; Peterman, 2000). Within these theories there are three major schools of thought (Temkin & Rohe, 1996)-ecological, subcultural, and political economy. Within ecological change are the invasion-succession (Burgess, 2008), filtering (Hoyt, 1939), and neighborhood life-cycle models (Downs, 1981; Hoover & Vernon, 1959), which basically treat neighborhood change as a natural evolution process. Subcultural models (Firey, 1945; Gans, 1962; Suttles, 1972) reject the economic determinism of the ecological models and

stress the attachment of residents to their neighborhoods as a key determinant of why and how residents live in certain parts of the city. Political economy models (Castells, 1983; Harvey, 1973; Logan & Molotch, 1987), on the other hand, highlight the institutional forces that influence neighborhood change. Rather than viewing urban development as a process of market equilibrium, as promoted by urban ecological theorists, political economists argue that social, economic, and political forces are the key drivers of neighborhood change. Considering the overall focus of this study and that it adopts a political economic framing of vulnerability production, political economy theories of neighborhood change are used to explain the changing patterns of social vulnerability within the study cities.

4.3 Political Economy Theories of Neighborhood Change

Political economy theorists explain neighborhood change through two broad streams of thought-the "urban growth machine" thesis (Logan & Molotch, 1987; Molotch, 1976), and the "urban restructuring" or "globalization" thesis (Borja & Castells, 1997; Sassen, 2000; Soja et al., 1983). While urban growth machine theorists argue that neighborhood change occurs through active exploitation of the real estate market and political process by local elites, theories of urban restructuring focus on how capital and labor restructuring at the global scale influences urban growth and movement within cities. One of the basic tenets of growth machine theorists is that the local growth coalition, driven by their fixation on economic growth, can bend the policy priorities of localities toward developmental rather than redistributional goals (Logan et al., 1997). This can be particularly problematic for vulnerable populations in poor neighborhoods who may face displacement and hardship, as happened through the 1960s "urban renewal" projects and now as a result of the gentrification process. Economic and labor restructuring, as argued by the urban restructuring theorists, also makes it harder for vulnerable populations to find better paying jobs or affordable housing. The transformation of the economy from manufacturing to services has relocated better paying jobs from the inner city to the suburban fringe and thereby, increased unemployment in poor neighborhoods (Dickens, 1999). Economic restructuring and increased liberalization also led the federal government to reduce funding for affordable housing (as discussed in Chapter 2) and made it difficult for vulnerable populations to find houses in safer areas. Decreased public spending on social services has strained low-income residents of many inner-city neighborhoods (Fainstein & Fainstein, 1985) and economic

restructuring has contributed to poverty rates in predominantly black neighborhoods rising faster than in white neighborhoods (Galster et al., 1997).

Globalization of the economy coupled with local economic restructuring is rapidly changing the demographic composition of cities, particularly in the growing coastal cities of the U.S. Demand for low-wage workers fueled an influx of immigrant workers from Latin America and Asia, and thereby created a "heterogeneous mosaic of new and old ethnicities" (Soja, 2000) breaking down the dominant black-white race paradigm within the cities, as Soja et al. (1983) showed for Los Angeles. These forces are also changing the patterns of social vulnerability in urban neighborhoods and this reality needs to be accounted for in climate adaptation initiatives. The next section explores the evolving and increasingly heterogeneous nature of socially vulnerable groups within the three coastal urban counties using biplots and in later sections through the analysis of Social Vulnerability Index (SoVI).

4.4 Exploring Social Vulnerability through Biplots

An historical exploration of social vulnerability requires an evaluation of how the different dimensions of vulnerability are related to each other and how they have changed over time in different cities. Dimensionality reduction through Principal Component Analysis (PCA) is one approach, but visualizing these dimensions can help to explain and communicate how social vulnerability has evolved over time, given demographic shifts and urban growth. Biplots are effective for analyzing multivariate data in that they can simultaneously provide information on both the samples and the variables in a two- or three-dimensional representation (La Grange et al., 2009; Le Roux & Gardner, 2005). For this study, the neighborhoods or census tracts are the samples and variables selected for measuring social vulnerability (as shown in Table 4.1) are the variables represented as a calibrated axis on the biplot. The distribution of the sample points on a biplot indicates how different census tracts are located along the axis of all the variables and at the same time, how the variables (or axes) are related to one another. Correlation biplots are created for this analysis that scale all the variables to unit variance¹⁷ and then adjusts the points and axes in such a way that the cosines of the angles between the axes approximate the correlations between the corresponding variables (La Grange et al., 2009). This means that if two

¹⁷ For calculating unit variance, all elements of a data matrix X are divided by their standard deviations (i.e., $\frac{[X]_{ij}}{sd(X_{(j)})}$). For correlation biplots, data are also centered around their means (i.e., $[X]_{ij} - mean(X_{(j)})$).

axes are closer together they are positively correlated and if they are in opposite directions (around 180°) that would indicate a negative correlation between them. As a result, the relative position of the axes visually communicates the degree of similarity among the variables they represent in a concise and accessible way.

#	Variables	Normalized Variable
1	Black or African-American population	% of Black population
2	Hispanic population	% of Hispanic population
3	Asian population	% of Asian population
4	Native American population	% of Native American population
5	Population under 5 years old	% of population under 5 years old
6	Population 65 years or older	% of population 65 years or older
7	Group quarters population	% of population living in group
		quarters
8	Foreign born population	% of foreign-born population
9	Household size	Average number of people per
		household
10	Female population	% of female population
11	Female-headed households	% of families with female-headed
		households with no spouse present
12	Female labor force participation	% of female population in civilian labor
		force
13	Public transportation dependence	% of workers (Civilian pop. 16+ and
		employed) using public transport
14	Education attainment	% of population over 25 years old with
		less than 12 years of education
15	Unemployment rate	% of the civilian labor force
		unemployed
16	Manufacturing employment	% of persons (16+ years old) employed
		in manufacturing, transportation and
		public administration

Table 4.1: Variables used for creating biplots and to calculate the SoVI.

Variables	Normalized Variable
Employment in service occupations	% of persons (16+ years old) employed
	in service occupations
Poverty rate	% of population in poverty
Social security recipients	% of population who are social security
	recipients
Average household income	Average household income last year (\$)
Renter-occupied housing units	% of renter-occupied housing units in
	total occupied housing units
Number of mobile homes	% of housing units that are mobile
	homes
Average gross rent	Average gross rent (\$) for renter-
	occupied housing units
Average home value	Average dollar value of owner
	occupied housing units
Population in civilian labor force	% of population in civilian labor force
Housing density	Number of houses per sq. mile
	VariablesEmployment in service occupationsPoverty rateSocial security recipientsAverage household incomeRenter-occupied housing unitsNumber of mobile homesAverage gross rentAverage home valuePopulation in civilian labor forceHousing density

Table 4.1 (cont.)

All biplots created for this analysis are presented in Appendix Figures A1 to A3. Figures 4.1 to 4.3 presented below also show the same biplots, but highlight three important race/ethnicity and poverty variables for easier interpretation of the findings from this biplot analysis. Alpha bags (Gardner, 2001; La Grange et al., 2009), containing 90% of the samples (i.e. $\alpha = 0.9$) are superimposed on the biplots to identify the prominent dimensions of vulnerability at different time periods. Alpha bags can be understood as a multivariate extension of the univariate boxplot which can contain any specified proportion of data nearest to the median (unlike the central 50% contained by boxplots) and as a result, is effective in highlighting differences in variation or the role of outlying values (Walters & Le Roux, 2008). When data points are more dispersed in the biplot, there would be an alpha bag of larger size to encompass defined percentage of samples (i.e. 90% for this study) and if some data points (i.e., census tracts) deviate significantly from the median value due to a higher value of any particular

variable (e.g., higher concentration of Hispanic or African-American population), the alpha bag will be elongated along the axis of that variable.

Figure 4.1 shows Harris County, Texas biplots for the four years delineating the three decade study period. Changes in the shape of the alpha bag and the relationship among the race/ethnicity and poverty axes are evident, particularly after 1990. Changes in the angles between the axes can be attributed to high growth of the Hispanic population and the increase of poverty in suburban census tracts, which was more concentrated in inner city areas prior to 1990. Harris County has experienced enormous growth in its Hispanic population in recent decades. In 1980 the Hispanic population was only about 15% but increased to 22% in 1990 and then rapidly climbed to 33% and 40% in 2000 and 2010 respectively¹⁸. The black or African American population remained consistent at about 18 to 20% of the total population at all three decades, and although absolute numbers increased, their percentage remained at a significantly lower rate than the Hispanic population. As a result, in 1980 and 1990 (when the Hispanic and African-American population shares were not that different) there were high correlations among poverty rate and minority population, as indicated by smaller angles among these axes (Figure 4.1). In 2000 however, when the Hispanic population jumped to 33%, a sizable number of census tracts exhibited a higher share of Hispanic population and this reduced the historically high correlation between concentration of Hispanic and black population. With growth of Hispanic population, this pattern continued in 2010 when the axes showed angles similar to those observed for 2000. Angles between the axis of poverty rate and race/ethnicity increased after 1990 primarily due to a greater negative correlation between the Hispanic and black populations in 2000 and 2010. Since the areas with higher minority populations continued to hold higher percentages of the population living in poverty, the axis for poverty rate is located in almost equal distance from these two axes, indicating a positive correlation with both the Hispanic and black population percentage. In 1980, mobile homes comprised a higher percentage of the housing stock in some census tracts of Harris County along with a higher percentage of the population employed in manufacturing jobs. This is the reason for the 1980 alpha bag having a second spike besides the one associated with high poverty and minority population (Figure 4.1, top left box). This distortion gradually decreased in subsequent decades and for 2000 and 2010 the alpha bags

¹⁸ Calculated from 1980, 1990, 2000 decadal census data and 2008-2012 ACS data (considered as representation of 2010). All values are rounded to nearest whole number for easies interpretation.

indicated two spikes for census tracts with a higher share of black population and Hispanic population. This indicates that for Harris County, poverty and minority populations are the dominant social vulnerability factors in recent decades, whereas other indicators (e.g. percentage of mobile homes) were dominant in previous decades.



Figure 4.1: Correlation biplots (with alpha bag) of Harris County with poverty rate, % black population and % Hispanic population highlighted

Unlike Harris County, which experienced consistently high population growth, Orleans Parish, Louisiana saw a gradual decrease in population over time-specifically, it lost about 145,000 people between 2000 and 2010¹⁹. Hurricane Katrina in 2005 is one of the key factors behind this population loss as the storm displaced thousands of people and the parish has yet to fully recover from that devastation. In terms of racial/ethnic composition, Orleans Parish did not experience any significant shift in contrast to Harris County. The black or African American population was the largest racial group in the parish in 2000 (about 55%) and remained so in 2010 (about 60%), while the Hispanic or Latino population continued to constitute a small minority (about 3% in 1980 to only 5% in 2010). These trends are also reflected in the biplots of Orleans Parish presented in Figure 4.2. In this case, the poverty rate has a higher correlation with the black population percentage (as indicated by smaller angle between these two axes) in all time periods. The very low presence of Hispanic population also contributed to the negative correlation between Hispanic population percentage with poverty rate and Black population percentage. Here, none of the vulnerability dimensions are significantly influencing the shape of the alpha bag, in contrast to the Harris County case. Still, the poverty rate and black population percentage are the two key factors that are moving in tandem with other dimensions of vulnerability (particularly after 1990), as evidenced by the concentration of the axes of other variables near these two axes (Figure 4.2). Principal Components Analysis for calculating the Social Vulnerability Index, as discussed in next section, revisits and further examines those other dimensions of vulnerability that correlate with poverty and race in Orleans Parish.

¹⁹ In 2000 total population of Orleans Parish was 484,674, which decreased to 339,016 in 2010, as documented by Census 2000 and ACS 2008-2012 respectively.



Figure 4.2: Correlation biplots (with alpha bag) of Orleans County with poverty rate, % Black population and % Hispanic population highlighted

The vast majority of the population in Hillsborough County in Florida has historically been white. Although the county experienced gradual growth of its black and Hispanic populations in recent decades, it neither experienced a high growth rate (like the Hispanic population in Harris County) nor a high concentration of racial or ethnic minorities (like the black population in Orleans Parish). As shown in the biplots in Figure 4.3, in all three time periods (with the exception of 1980) the poverty rate and concentration of minority populations were highly correlated with each other in Hillsborough County (as indicated by the smaller angles between the three axes). The departure from this trend seen in 1980 is due to the fact that the Hispanic population was smaller compared to the black population (10% vs. 13% respectively), but by 1990 these populations were almost equal (both comprising about 13% of the total population)²⁰. In 2000 the Hispanic population surpassed the black population (18% vs. 16%) and the gap increased further by 2010 (25% Hispanic and 17% black population)²¹. Census tracts with a higher share of minority populations also tend to have higher poverty rates, as indicated by the small angles between axis of the poverty rate and Black and Hispanic population percentage variables (Figure 4.3). Besides poverty and minority population, the alpha bags for Hillsborough County indicate other emerging dimensions of social vulnerability in this county. While in 1980 and 1990, the shape of the alpha bags was primarily determined by poverty and minority population, in 2010 the percent elderly population is associated with another spike or elongation. This indicates that census tracts where the elderly comprise a larger percentage of the overall population also tend to exhibit higher social vulnerability.

²⁰ Calculated from 1980 and 1990 census report collected through Social Explorer

²¹ Calculated from 2000 census and 2008-2012 ACS data. All values are rounded to nearest whole number for easies interpretation.



Figure 4.3: Correlation biplots (with alpha bag) of Hillsborough County with poverty rate, % Black population and % Hispanic population highlighted

4.5 Results of PCA Analysis

The SoVI was calculated by applying Principal Component Analysis (PCA) to 26 variables (table 4.1) that have been established in the literature and were described in Chapter 2. After reducing the number of variables through PCA, the resulting component scores were aggregated using the additive approach pioneered by Cutter et al. (2003). The PCA results indicate how the prominent dimensions of vulnerability have evolved over time, as well as how the role of different variables used to construct the SoVI changed along with the shifting demographic composition of the counties. As mentioned in Chapter 3, this study used parallel analysis (Zwick & Velicer, 1986) rather than the Kaiser criterion applied in the original SoVI methodology of Cutter et al. (2003) to determine how many components to retain. Parallel analysis usually retains fewer components than the Kaiser criterion and is considered a superior alternative for determining the optimal number of components (Tate, 2012). After applying PCA, the selected components were interpreted and labeled based on the characteristics of the variables, specifically the loadings of the variables. Tables 4.2 to 4.4 show the selected vulnerability components along with the percentage of overall variance explained by those components for each of the three study counties. Appendix Tables A1 to A4 present a more detailed version of these tables and show which variables are contributing most to each of the components.

Among the three study counties, one common trend from the PCA was that the percentage of total variance explained by the SoVI components gradually decreased over time. Further, the parallel analysis approach retained a different number of components for each of the counties (and even in different years) and the variance explained by the individual components also changed across time periods. This suggests actual changes in the composition and significance of the underlying dimensions of social vulnerability in different time periods. For census tracts in Harris County (Table 4.2), the overall variance explained by the selected components decreased from 74.4% in 1990 to 72.5% in 2000 and 65.5% in 2010. In all four years, five components were consistently retained by the parallel analysis with *Race and socioeconomic status* (i.e., percent Black, poverty, unemployment, income, etc.) as the leading component in all years. While *Housing and labor force* was the second most important component in 1980 and 1990 (explaining 16.5% and 15.1% variance), it was replaced by the *Hispanic and foreign-born* component in 2000 and 2010 (explaining 17.6% and 13.9%

variance), which can be attributed to the rapid growth of the Hispanic population during this time period.

The PCA components explained 71% of the overall variance among the census tracts of Orleans Parish in 1980, 70.5% in 1990, 69.6% in 2000, and 67.7% in 2010. While four components were retained by the parallel analysis in 1980, 1990, and 2000, five components were retained in 2010. Race and socio-economic variables consistently emerged as the leading component, although the percentage of variance it explained decreased significantly in 2010. This can be attributed to the massive loss of population in Orleans County between 2000 and 2010 as a result of Hurricane Katrina, which may also have contributed to the retention of more components in 2010 (i.e., previously insignificant components became more prominent in the wake of population loss).

In the case of Hillsborough County, the PCA did not indicate any consistent trend (in terms of the percentage of variance explained or for the number of components retained) unlike Harris or Orleans County. In 1990, 73.6% of the variance among the census tracts of Hillsborough County were explained by five components, by 2000 74.3% was explained by six components, and in 2010 only 65.8% was explained by the five components retained. Race and socio-economic variables also comprised the leading component for all four decades here. A growing concentration of elderly population in certain census tracts (as mentioned in previous section) contributed to the significantly higher percentage of variance explained by the *Age and labor force* component in 2010 (increased from 13.3% in 2000 to 16.2%).

	1980	1990	2000	2010
% variance explained	71.8	74.4	72.5	65.5
No. of components	5	5	5	5
	Race and socio-	Race and socio-	Race and socio-	Race and socio-
	economic status (28.9)	economic status (29.4)	economic status (27.5)	economic status (23.4)
	Housing and labor	Housing and labor	Hispanic and foreign-	Hispanic and foreign-
Maior components	(c.01) to $c.01$	IOTCE (13.1)	porn (1/.6)	(13.9)
Major components	Hispanic and foreign-	Hispanic and foreign-	Employment and	Age and labor force
w variatice expranted	born (9.9)	born (12)	housing (11.4)	(11.4)
	Age and gender (8.9)	Age (10)	Age and labor force (10)	Employment, housing (9.2)
	Employment, mob.	Employment, mob.	Gender (6)	Gender (7.6)
	Homes (7.6)	Homes (7.9)		

ounty
Ŭ
rris
Haı
for
sults
V res
PCA
4.2:
ole 4
Tał

2010	67.7	5	Race and socio-	economic status (26.5)	Employment, gender	(12.7)	Hispanic and foreign-	born (10.9)	Age and gender (10.1)		Housing, employment	(C./)
2000	69.6	4	Race and socio-	economic status (38.7)	Hispanic and	employment (15)	Foreign-born and	Asian (8.7)	Age and gender (7.2)			
1990	70.5	4	Race and socio-	economic status (39.8)	Hispanic population	(11.9)	Foreign-born, Asian,	mob. Homes (10.5)	Employment and	Gender (8.3)		
1980	71	4	Race and socio-	economic status (40.3)	Age and Hispanic	(13.4)	Foreign-born and Asian	(8.7)	Employment, Gender	(8.6)		
	% variance explained	No. of components			Major components (%	variance explained)						

Parish
Orleans
for
results
PCA
4.3:
Table

	1980	1990	2000	2010
% variance explained	75.5	73.6	74.3	65.8
No. of components	9	5	9	5
	Race and socio-	Race and socio-	Race and socio-	Race and socio-
	economic status (31.1)	economic status (32.6)	economic status (27.6)	economic status (23.8)
	Age, labor force (11.5)	Age, labor force (13.9)	Age, labor force (13.3)	Age, labor force (16.2)
Major components (%	Household size, age	Household size,	Household size,	Hispanic, foreign-born
variance explained)	(9.1)	employment (10.7)	employment (10.9)	(10)
	Hispanic, foreign-born	Hispanic, foreign-born	Hispanic, foreign-born	Household size,
	(8.6)	(6)	(9.5)	housing (8.6)
	Employment, housing	Gender and Housing	Income, home value	Gender, employment
	(8.5)	(7.4)	(7.3)	(7.2)
	Gender (6.7)		Employment, housing	

>
County
orough
Hillsbc
for
results
PCA
4.4:
able

4.6 Spatial Changes in the SoVI

Component scores calculated through PCA were aggregated to derive the overall composite social vulnerability index (SoVI). Since the composition of the vulnerability factors varies from place to place and even over time in the same area (as the biplots in section 4.3 showed), the SoVI was calculated for each county and each decadal period separately. Figures 4.4 to 4.6 present the SoVI maps for the three study counties in all four decades. In these maps, the SoVI is classified based on standard deviations from the mean, which helps to better visualize comparative vulnerability of the census tracts within a county and longitudinally. In the case of Harris County, Texas (Figure 4.4) in 1980 and 1990, high concentrations of SoVI near the inner-city (surrounding the downtown) areas were evident. Between 2000 and 2010 social vulnerability decreased, particularly in the north-west and western parts of the downtown area, due to gentrification in the Heights, Memorial Park, and other surrounding neighborhoods consistent with the findings of prior studies (Podagrosi et al., 2011). An increase in vulnerability in the south and south-west part of the county, particularly in the Central Southwest, Sunnyside, and Southwest Houston neighborhoods is also evident during the same time period.

In Orleans Parish, Louisiana (Figure 4.5) changes in the SoVI between 2000 and 2010 are more apparent, which can be attributed to displacement after Hurricane Katrina, particularly in the Gentilly, Upper Ninth Ward, and Lower Ninth Ward neighborhoods in the north and northeast parts from the downtown area. For Hillsborough County, Florida (Figure 4.6) the SoVI appears to be less spatially concentrated in 2010 than in earlier years, particularly in the innercity areas of Tampa. During the same time period vulnerability has increased in southern part of county in the census tracts along Hillsborough Bay.













4.7 Cluster Analysis of the SoVI

While the SoVI maps presented in previous section show the spatial distribution of vulnerability within the counties, they do not indicate the extent to which census tracts with higher SoVI are clustered together and how this spatial distribution changed over time. To identify the patterns of similarity and dissimilarity in the clustering of social vulnerability, spatial autocorrelation of the SoVI among the census tracts was evaluated. Both global and local measures of spatial autocorrelation were calculated and when considered jointly, they reveal two distinct but complementary spatial characteristics of vulnerability. The global autocorrelation measure involves studying the entire map pattern and identifies whether the observed pattern is clustered, dispersed, or random (Cliff & Ord, 1981; Goodchild, 1986) and in this case, indicates whether overall social vulnerability within the county has become more or less concentrated over time. Local indicators of spatial of spatial autocorrelation or LISA statistics (Anselin, 1995), on the other hand, capture local variability and identify significant clusters of similar SoVI values (high or low social vulnerability). Tables 4.5 to 4.7 present the global Moran's I²² and number of census tracts within clusters based on local Moran's I of the SoVI for all three study counties, while Figures 4.7 to 4.9 present the corresponding LISA cluster maps for the three counties.

One important trend that emerged from the global analysis of SoVI is that the Moran's I value decreased over time for all of the counties. This suggests that, despite experiencing different patterns of population growth (as mentioned in section 4.1) and different compositions of vulnerability (as revealed through the biplots and PCA earlier), in all three study counties the overall concentration of social vulnerability has decreased over time. Although the positive values of Moran's I (all of which were statistically significant at p < 0.05) indicate that social vulnerability is highly concentrated in all of the counties, their decreasing values indicate that the level of concentration of vulnerability has lessened. In Harris County, Texas (Table 4.4), the global Moran's I was 0.63 in 1980, but gradually decreased to 0.46 in 2010 and the number of census tracts in high-high clusters (indicating concentration of high values of the SoVI) also decreased from 88 (12.34% of tracts) in 1980 to 70 (9% of tracts) in 2010.

 $^{^{22}}$ The Moran's I statistic is interpreted as follows: a value close to +1 represents strong similarity between the values of the SoVI (i.e. higher concentration); a value of -1 indicates dissimilarity (i.e. higher dispersion); while a value of zero represents a random pattern.

Global	<u> </u>		1990 0.62		200)0	2010 0.46	
Moran's I					0.5	8		
LISA categories	Count	%total	Count	%total	Count	%total	Count	%total
Significant local								
High-High	88	12.34	100	12.95	81	10.51	70	9.00
Low-Low	0	0.00	0	0.00	0	0.00	0	0.00
Spatial outliers								
Low-High	62	8.70	79	10.23	72	9.34	57	7.33
High-Low	1	0.14	2	0.26	0	0.00	2	0.26
No statistically significant spatial clustering								
Tracts	562	78.82	591	76.55	618	80.16	649	83.42
Total	713	100	772	100.00	771	100.00	778	100.00

Table 4.5: Spatial clustering statistics and LISA cluster categories for Harris County

In the case of Orleans Parish, Louisiana (Table 4.6) the global Moran's I was lower in earlier years (0.40 in 1980 and 0.37 1990), but increased in 2000 (0.47) and then decreased in 2010 (0.44). This suggests an overall decrease in the concentration of social vulnerability in recent years, but the number of tracts in high-high clusters barely changed. There were 10 census tracts in high-high clusters in 1980, which decreased in 1990 and 2000 only to climb back to 10 in 2010.

Global	198	30	1990		200)0	2010	
Moran's I	0.4	0	0.3	37	0.4	7	0.4	4
LISA categories	Count	%total	Count	%total	Count	%total	Count	%total
Significant loca	l spatial c	lusters						
High-High	10	5.99	8	4.85	9	5.42	10	6.21
Low-Low	0	0.00	0	0.00	0	0.00	0	0.00
Spatial outliers								
Low-High	5	2.99	5	3.03	5	3.01	14	8.70
High-Low	1	0.60	1	0.61	0	0.00	0	0.00
No statistically significant spatial clustering								
Tracts	151	90.42	151	91.52	152	91.57	137	85.09
Total	167	100.00	165	100.00	166	100.00	161	100.00

Table 4.6: Spatial clustering statistics and LISA cluster categories for Orleans Parish

For Hillsborough County in Florida (Table 4.7) the concentration of vulnerability (measured with global Moran's I) followed a consistent downward trend, similar to Harris County in Texas. Although Hillsborough County had a low concentration of vulnerability in 1980 (0.53), it increased in 1990 (0.62) before gradually decreasing in 2000 (0.57) and 2010 (0.42). For a number of tracts in high vulnerability clusters (i.e. high-high spatial clusters), however, the number of census tracts did not follow the downward trend of the global Moran's I. While there were 44 census tracts located in high vulnerability clusters in 1990, this number decreased to 33 in 2000, then increased to 46 in 2010. The larger number of census tracts (326 versus 300 in 2000) included in the 2010 analysis (which were excluded²³ in earlier years due to low population counts) may have had an effect, but this is not a sufficient explanation since even with a lower number of census tracts in 1990, a greater number of census tracts were part of high-high SoVI clusters than in 2000.

²³ Since both global and local spatial autocorrelations were measured separately for every year (i.e. with separate weight matrix), excluding any feature in one year shouldn't influence the result of another year. Although having different number of features may make the temporal comparison difficult, as the result shows here, number of features are not influencing the results. Also, in most cases excluded features are located nearby (i.e. tracts with fewer population clustered together), and as a result it can be expected that they will not affect the results in subsequent years.

Global	198	1980		990	20	00	2010	
Moran's I	0.53		0.62		0.5	57	0.42	
LISA categories	Count	% total	Count	% total	Count	% total	Count	%total
Significant loc	al spatial c	lusters						
High-High	24	11.37	44	17.89	33	11.00	46	14.11
Low-Low	0	0.00	0	0.00	0	0.00	1	0.31
Spatial outliers								
Low-High	17	8.06	6	2.44	27	9.00	23	7.06
High-Low	0	0.00	0	0.00	2	0.67	0	0.00
No statistically	significar	nt spatial cl	ustering					
Tracts	211	100.00	246	100.00	238	79.33	256	78.53
Total	252	119.43	296	120.33	300	100.00	326	100

Table 4.7: Spatial clustering statistics and LISA cluster categories for Hillsborough County

The LISA cluster maps in Figures 4.7 to 4.9 show the locations of clusters and spatial outliers within the study counties and how they changed over time. One trend that can be identified from these maps is, the high vulnerability clusters (i.e. high-high clusters) have become relatively less spatially concentrated in all three counties, and this is most evident in Harris County. In Harris County some of the high-high clusters have shifted from the immediate north and north-west vicinity of the downtown to further northern location. For Orleans Parish (Figure 4.8) some shifts in the high-high cluster near the downtown area can be identified between 2000 and 2010. In Hillsborough County (Figure 4.9) there is also evidence of fragmentation of established high-high clusters during the same time period. Compared to 2000, in 2010 there were fewer census tracts in the high vulnerability clusters located in northern and southern parts of Hillsborough County.












4.8 Temporal Changes in the SoVI

Measuring temporal change in social vulnerability is a challenging task due to shifts in the composition of vulnerability at the different time periods considered. Still, if we consider the SoVI as a relative measure of vulnerability within a county, it can be useful in examining how the degree of social vulnerability among all census tracts in the county has changed over time and for identifying areas that have experienced an increase or decrease in social vulnerability. Keeping this in mind, the approach taken by Cutter & Finch (2008) is applied here to identify historical trends of social vulnerability within the three study counties. Although Cutter & Finch (2008) focused on U.S. counties, this study adapted their methods for temporal trend analysis at the census tract level. Specifically, individual SoVI scores of census tracts were converted to zscores (based on county mean score per decade) in order to improve their comparability over time. If this transformed score consistently increased over time in a census tract, that tract can be viewed as an area experiencing a gradual increase in social vulnerability. Applying simple linear regression, a line of best fit from 1980 to 2010 was calculated for each of the census tracts (with their transformed SoVI score for all time periods considered). The slope of the line of best fit indicates the directionality of vulnerability over time and the resulting R^2 captures the strength of this best fit line in capturing the trend in the transformed SoVI score. A positive slope of the best fit line indicates an increasing trend of social vulnerability and a negative slope indicates decreasing social vulnerability in a census tract. For all regressions, F-statistics were used to determine the statistical significance of the best fit lines at a 0.05 alpha level. This is an inherently aspatial approach to evaluating longitudinal trends of social vulnerability since each of the census tract is analyzed as a separate entity without considering changes in vulnerability in surrounding regions, but it still helps to visually locate areas that experienced a consistent increase or decrease in social vulnerability over time. Another potential issue with this approach is that it is based on the vulnerability of a census tract with respect to the whole county. As a result, even if a census tract experienced an increase in vulnerability in a certain time period, a much higher increase or decrease of vulnerability in other areas may influence its transformed SoVI score. Also, census tracts having a high increase or decrease in vulnerability only in recent decades (with opposite trends in previous decades) may end up showing an insignificant overall trend. Accepting these limitations, this study adopted the Cutter & Finch (2008) approach for

evaluating longitudinal trends of social vulnerability, given its establishment in the literature and because there is no other widely accepted alternative for performing this kind of analysis.

SoVI trend maps for the three study counties are presented in Figures 4.10 to 4.12 and include the coefficient of the best fit trend line of SoVI scores for each of the tracts, outlining the tracts for which the trends were found to be statistically significant (at the 0.05 alpha level). As mentioned previously, here a positive slope indicates a consistent increase in social vulnerability and a negative slope indicates a consistent decrease in social vulnerability over time. From these maps one common trend can be identified for all three counties—the decrease of social vulnerability in inner city census tracts and increase of vulnerability in suburban census tracts. This can be attributed to the general phenomenon of suburbanization of poverty in major U.S. cities, which is widely discussed by the neighborhood change theorists (Dickens, 1999). Although these trends were statistically significant for only a few census tracts, these maps still provide an overall representation of the longitudinal trend of social vulnerability in the study counties.



Figure 4.10: Temporal trends of SoVI in Harris County, TX (1980-2010)



Figure 4.11: Temporal trends of SoVI in Orleans County, LA (1980-2010)





4.9 Challenges for Adaptation Planning

This chapter identified several commonalities in the observed pattern of social vulnerability change among the study counties. Despite having drastically different population growth trajectories and being located in different political and economic settings, in recent decades the spatial concentration of social vulnerability has gradually decreased in all of them. However, the composition of social vulnerability was different in each of the counties and also changed in a non-uniform way over time. For Harris County (Houston) in Texas, the high growth of the Hispanic and immigrant populations in recent decades elevated the importance of these factors in the SoVI, along with the percentage of black or African-American population and the poverty rate. This trend is consistent with the "urban restructuring" or "globalization" thesis of neighborhood change (Borja & Castells, 1997; Sassen, 2000; Soja et al., 1983) that attributes these demographic changes to capital and labor restructuring both at the global and local level. Demand for cheap labor has attracted sizeable immigrant and minority populations, which will make adaptation planning efforts more challenging because of the tendency of these groups to exhibit higher vulnerability. Ensuring equity and the legitimacy of adaptation efforts (Adger et al., 2005) would be problematic in this changed context when these new ethnic minority and immigrant population lack adequate political participation to have their voices heard in the planning process.

In Orleans Parish (New Orleans) Louisiana, displacement by Hurricane Katrina has significantly influenced the pattern of social vulnerability, which has become less concentrated and less dominated by race and socio-economic indicators. As the Katrina recovery process moves forward and planning for climate adaptation continues, this changed vulnerability pattern should be kept in mind both to ensure equitable outcomes from the planning process and to identify effective ways of avoiding future disasters. Hillsborough County (Tampa) in Florida exhibits two notable trends in its social vulnerability patterns, along with the decreased concentration of vulnerability found in the other two counties. In this county, gentrification in the inner city areas is pushing socially vulnerable populations (primarily minority and low income groups) to suburban and coastal census tracts and at the same time, some of the coastal locations are experiencing high growth of elderly populations due to the development of retirement communities there. These trends can be explained by urban 'growth machine' theories of neighborhood change (Logan & Molotch, 1987; Molotch, 1976) that emphasize how the

economic interests of local elites drive policy decisions, which are in turn changing the pattern of social vulnerability on the ground. Both of these trends will make adaptation planning more challenging in the future and reveal two priorities—first evaluating the climatic risks of census tracts being populated by low-income and minority population and finding safer places for them, and second ensuring that climatic uncertainties are considered when developing retirement communities and that steps are taken to prevent displacement of other vulnerable groups.

CHAPTER 5 SUBSIDIZED HOUSING AND SOCIAL VULNERABILITY IN A MULTI-HAZARD URBAN AREA

5.1 Study Area

Harris County, Texas is taken as a detailed case study based on its level of risk exposure to both natural and technological hazards (Cutter et al., 2003; EPA, 2008; Nicholls et al., 2008; Sexton et al., 2007) and its high racial/ethnic diversity. As reported in the 2012 American Community Survey (ACS), this county has more than 4 million residents with 41.5 percent identifying as Hispanic/Latino, 19.5 percent as Black/African American, and about 17.3 percent living below the poverty level. The population of this county is primarily centered in the city of Houston (Figure 5.1), the largest cultural and economic center of the Southwestern United States. Houston is also the only major city in the United States without zoning as an element of its land use planning and is often portrayed as an archetype of the free enterprise, capitalist, or laissezfaire city (Lamare, 1998; Lin, 1995). Despite a lack of zoning, local land use regulatory policies made by the municipality significantly influence urban development in Harris county (Qian, 2010), but the high natural and technological risk of this area makes it a complex environment for socially vulnerable populations to navigate. Efforts to minimize costs for potential investors as an economic development strategy in the metropolitan area have created less than ideal living conditions, especially for socially vulnerable groups (Vojnovic, 2003). Similar to other large U.S. cities, Houston is also faced with environmental justice problems as a result of disproportionately placing environmental hazards in areas occupied by lower income groups and non-whites (Bullard & Wright, 1993; Chakraborty et al., 2014; Pulido, 2000). While this issue is extensively studied in the environmental justice literature, the role of policies like low-income housing subsidies in placing marginalized groups in harm's way in a multi-hazard urban environment like Houston has not yet been explored. The spatial distribution of housing subsidies under the broader umbrella of two different programs (HCV and LIHTC) is explored here and ways in which housing subsidies influence neighborhood social vulnerability over time are examined through spatial econometric models.



Figure 5.1: Map of Harris County, Texas

5.2 Hazard Risk Exposure in Harris County

In order to portray the multi-hazard risk context of Harris County, both natural and technological risks are considered in this study. Specifically, natural hazard risk due to flood and hurricane storm surge and technological hazard risk due to the locations of TRI facilities, with Chapter 3 detailing the methodological steps followed for creating these multi-hazard risk measures. Figure 5.2 shows the natural hazard risks identified at the census tract level and in this case, risk is measured as the percentage of residential lots in each census tract that fall in either the 100 year flood plain or a Category-1 hurricane risk zone. Some census tracts are excluded from this analysis (shown in grey) that had very few people (less than 1,000 in both 2000 and 2010) or a high percentage (greater than 30%) of residents living in group quarter housing (i.e., jails, senior housing, university campuses, etc.). Although Houston has a vast amount of low-lying land in its south-east and eastern region, natural hazard exposure was not found to be significantly higher in these parts of that region (compared to other areas) due to the lower percentage of residential areas falling in the 100 year flood plain or Category 1 hurricane risk

zone. Still, the map (Figure 5.2) shows a high number of census tracts having at least 40 percent of their residential lots in the natural hazard risk zone, particularly concentrated in the southwest and northern part of the county.



Figure 5.2: Natural hazard risk (i.e. % of residential lots in flood and hurricane risk zone) of Harris County

As detailed in Chapter 3, technological hazard risk for Harris County is estimated using two different measures of distance decay which, produce significantly different hazard exposure patterns. Since there is no theoretical consensus on the ideal specification for distance decay or what the threshold distance for impacts from a facility should be, this study considers both approaches in order to capture as wide a spectrum of technological risk as possible. Since this study is more interested in relative hazard risk and how people locate within this hazard context, proximal exposure due to distance and total toxic emission from TRI facilities is considered more relevant than the actual health risks posed by a specific type of toxic chemical. Further, prior studies have extensively examined such health risks due to both point and non-point sources of emissions in Harris County or Houston (Chakraborty et al., 2014; Linder et al., 2008; Sexton et al., 2007). Considering the variability of emissions in different years, all reported TRI locations and their emissions from 2000 to 2010 were included in the hazard calculation and then an average level of hazard exposure is identified for the whole study period. Although it can be argued that new facilities appearing during this time period could in theory alter this risk measure, the data do not indicate that this is the case. Firstly, there was not a large increase in reported TRI facilities during this time (349 in 2001 increased to 367 in 2010) and secondly, the new facilities were located mostly in the areas which already had other facilities nearby. Due to these two factors, the relative risk measures should not change significantly, which is a key assumption of this study.

Figures 5.3 and 5.4 show quantile maps for risk measures calculated using the two distance decay approaches adopted for this study. Since it takes the average of exposure (for 2000 to 2010) weighted by the total emissions, measures using the power function were denoted AWPOWR, while measures based on the WCPE function were named AWCPE. Drastic differences in the hazard landscapes created under the two distance decay specifications are evident in Figures 5.3 and 5.4. Since the power function considers high exposure in the immediate vicinity of a facility with a rapid decrease in impact thereafter and does not consider any distance threshold or cutoff, it results in a relatively larger area identified as exposed to technological hazards. On the other hand, since the WCPE function imposes a distance threshold (Figure 5.4 shows a one mile threshold) it confines exposure to areas within that threshold distance and does not consider impacts for areas located far away from the facilities. Still, there are some common areas identified as high technological risk zones in both of the maps and in particular, near the south-eastern ship channel and along the north-west corridor (Highway 290).



Figure 5.3: Technological risk exposure (measured by power function) in Harris County



Figure 5.4: Technological risk exposure (measured by WCPE function, 1 mile distance threshold) in Harris County

5.3 Social Vulnerability in the Hazard Zones

The hazard risk measures discussed in previous section were used in a spatial regression analysis to understand the interaction between hazard risk and social vulnerability. This section discusses how the population in different racial/ethnic groups and poverty statuses are spatially distributed within this hazard context of Harris County in the year 2000 and 2010. Figures 5.5 to 5.7 show the race/ethnicity distribution within different levels of natural and technological hazard areas. Contrasting trends in the spatial distribution of Non-Hispanic White (henceforth mentioned as 'white') and Hispanic populations is evident here, while for the Non-Hispanic Black or African-American population (henceforth mentioned as 'black') this trend is unclear. With the growth of the Hispanic population in Harris County, the white population percentage has decreased significantly between 2000 and 2010, but within the identified hazard zones the relative distribution is consistent in both time periods. While the White population percentage is higher in safer areas (i.e., less risk from flood or hurricane storm surge), in hazardous areas the percentage of Hispanic population is higher consistently in both 2000 and 2010.



Low: Less than 20% residential area in hazard zone, Medium: 20%-40% area in hazard zone, High: More than 40% area in hazard zone

Figure 5.5: Race/ethnicity distribution in natural hazard areas

Considering the different hazard landscapes created by the power function and the WCPE function (Figures 5.3 and 5.4) it could be expected that the pattern of racial/ethnic distribution within hazard exposed areas would be different, but as shown in Figures 5.6 and 5.7 the patterns are actually similar in both 2000 and 2010. Since the power function assigns risk measures to all census tracts, its median value is used as the cutoff point for technological hazards (Figure 5.6) and in the case of the WCPE function, areas within the threshold distance of one mile is taken as the cutoff point (Figure 5.7). Despite these different approaches, both of the methods for technological risk measures indicate that the percent white is significantly lower and percent Hispanic is significantly higher in hazardous areas in both 2000 and 2010. For the black population however, these two approaches show different distributions. While the hazard zones from the power function do not show any consistent trend (Figure 5.6), from the WCPE function



(Figure 5.7) it appears that the black population percentage within technological hazard areas was higher compared to that in non-hazard areas.

Figure 5.6: Race/ethnicity distribution within technological hazard zones (power function)





Poverty rates within the natural and technological hazard areas are comparatively higher than the non-hazard areas and Figures 5.8 to 5.10 show this trend for both 2000 and 2010. Since the total poverty rate has increased in Harris County between these two time periods, the percentages of people under the poverty level has increased in all areas, but within hazardous areas, the concentration of poverty is consistently higher throughout the study period. All three figures (5.8 to 5.10) also show similar trends for female-headed families in poverty (as a percentage of all families with children). This indicator is considered here based on its enduring status in the principal component analysis as a significant contributor to social vulnerability. Although it can be expected to move in tandem with the findings for overall poverty rate, it is included here considering the higher degree of vulnerability experienced by female-headed households during any disaster event (Donner, 2003; Rodriguez & Russell, 2006). As Figures 5.8 to 5.10 indicate, among all the families with children living in different hazard zones, the presence of female-headed families living in poverty is significantly higher in natural or technological hazard areas.



Low: Less than 20% residential area in hazard zone, Medium: 20%-40% area in hazard zone, High: More than 40% area in hazard zone

Figure 5.8: Percentage of population under poverty and Percentage of female headed families (with children) under poverty by natural hazard zones



Figure 5.9: Percentage of population under poverty and Percentage of female headed families under poverty (with children) by technological hazard zones (power function)



Figure 5.10: Percentage of population under poverty and Percentage of female headed families under poverty (with children) by technological hazard zones (WCPE function)

5.4 Housing Subsidies in the Hazard Zones

While the above findings on race/ethnicity or poverty distribution in urban hazard zones is not surprising, there is yet to be any study that explores the location outcomes produced by housing subsidies within these multi-hazard contexts (Cutter et al., 2001; Houston et al., 2013). This section centers on how the two housing subsidy programs considered in this study are performing in terms of avoiding hazardous areas. Although the findings discussed in previous section were for the distribution of population groups within the hazard zones, in the interest of having a better reference for comparison, the population distribution throughout the county is compared to the distribution of housing subsidies among the hazard zones. If the distribution of housing subsidies follows the overall distribution of population, it can be argued that housing units provided under these programs are not concentrated within hazardous areas. However, considering the broader objective of poverty deconcentration associated with these programs, it can be expected that the housing subsidies would be located in less hazardous areas compared to the overall population. Since the two programs have different approaches for providing housing subsidies (demand based HCV and supply oriented LIHTC), different outcomes are to be expected.

Figures 5.11 to 5.13 indicate that for all hazard measures, subsidized housing units are not successfully avoiding hazardous areas; rather they are slightly more concentrated in hazard zones compared to the overall population distribution. Although both of the subsidy programs performed better in avoiding highly exposed natural hazard zones (i.e. census tracts with more than 40% in natural hazard zone), in the case of moderate exposure (i.e. census tracts with more than 20% in natural hazard zone) they exhibit a slightly higher presence compared to the overall population distribution in both 2000 and 2010 (Figure 5.11). Between the two programs, the HCV program appears to be performing marginally better than the LIHTC in avoiding moderate hazard zones, but in the case of high exposure areas, the LIHTC is performing better.



Low: Less than 20% residential area in hazard zone, Medium: 20%-40% area in hazard zone, High: More than 40% area in hazard zone



For technological hazards, both of the hazard measures indicate similar outcomes for the subsidy programs, although they vary in terms of the extent of their differences (between the programs). Following the trends of natural hazards, in the case of technological hazards both of the housing subsidy programs also failed to find safer areas and instead their placement in hazardous areas is marginally higher compared to the overall distribution of the population. For example, if we consider the hazard zones identified by the WCPE function (Figure 5.13), in 2010 about 30 percent of the population of Harris County lived in a hazardous area, but about 33 percent of HCV recipients and 40 percent of LIHTC units were found in those areas during the same time period. Between the two programs, despite performing better in terms of avoiding technological hazard areas in 2000 (33% LIHTC versus 36% HCV, Figure 5.13), in 2010 a higher proportion of LIHTC units were found in hazardous areas (40% versus 33% for HCV).



Figure 5.12: Comparing distribution of HCV recipients and LIHTC units to population distribution within technological hazard zones (power function) of Harris County



Figure 5.13: Comparing distribution of HCV recipients and LIHTC units to population distribution within technological hazard zones (WCPE function) of Harris County

5.5 Subsidized Housing in a Multi-hazard Area: Impacts on Social Vulnerability

The broad objective of this study is to explore the drivers of social vulnerability and examine how housing subsidy programs may act as a driver to locate more vulnerable population in harm's way. The previous sections advance this objective by clearly depicting the higher presence of vulnerable populations in hazardous areas and at the same time, how housing subsidies are failing to avoid the hazard zones of Harris County. This outcome can be attributed to limitations in the present mechanisms of subsidized housing provision (detailed in Chapter 6), while the political economic and environmental justice framing of vulnerability (discussed in Chapter 2) explains the location decisions of vulnerable populations in a contested urban space. Now the critical question to be asked is how the ostensibly progressive agenda of providing housing subsidies in a neoliberal political setting can influence and even exacerbate vulnerability in a multi-hazard urban area. This is where the idea of the production of urban vulnerability (Dooling, 2012), that examines how political economies of resource use and normative planning and management interventions influence which places and populations are made vulnerable (Collins, 2009; Orsi, 2004), reenters the discussion. Keeping this broader theoretical underpinning in mind, this study explores the impacts of placing subsidized housing in hazardous areas on the social vulnerability outcomes in Harris County. While the regression analysis discussed in the next section helps to empirically examine the relationship, a descriptive exploration is conducted before estimating the models.

Since social vulnerability is a multidimensional concept, it makes sense to explore it through an index (SoVI) as discussed in Chapter 4. But such an index is difficult to interpret for specific policy outcomes due to its changing composition from year to year. Keeping these limitations of the SoVI in mind, this study considers prominent indicators that contribute significantly to the overall social vulnerability index. As discussed in Chapter 4, poverty was found to be a consistent indicator that met this criterion for all study years. But it is also true that the experience of poverty is not the same across all population sub-groups. For example, analyzing the neighborhood crime rate of HCV recipients, Lens et al. (2011) found that black voucher holders lived in significantly safer areas than poor households of the same race, but that Hispanic and white voucher holders did not experience the same outcome. Considering this possible difference in the location patterns of those in poverty among population sub-groups, this study considers the poverty rate of racial and ethnic subgroups (e.g., Hispanic, Non-Hispanic

White, Non-Hispanic Black) along with the overall poverty rate. Before exploring the poverty rate for different subgroups, Table 5.1 shows population growth in different areas of Harris County categorized by the presence of natural hazard zones (greater than 40% residential land in natural hazard area) and increase in vouchers between 2000 and 2010. As this table shows, safer areas (away from natural hazards) that had an increase in HCV recipients experienced the highest population growth (32.34%), but it also shows significant population growth (7.46%) in hazardous areas with HCV growth between 2000 and 2010.

	Population				
Natural Hazard and HCV growth	2000	2010	Growth (%)		
Nat. Haz. Areas with increased HCV	246,399	264,782	7.46		
Other areas with increased HCV	1,878,958	2,486,683	32.34		
Nat. Haz. Areas without increased HCV	263,199	263,380	0.07		
Other areas without increased HCV	991,011	1,063,400	7.30		

Table 5.1: Population growth by location in natural hazard^{*} areas and HCV growth

*Areas with more than 40% residential land in natural hazard Area

While all the areas irrespective of hazard level or HCV presence experienced significant population growth, a question of interest for this study is how the poverty profile changed in those areas during the study period. Table 5.2 presents changes in vulnerability by the overall poverty rate and also by the poverty rate of major racial and ethnic population subgroups. As shown, the incidence of poverty for all groups was significantly higher in natural hazard areas that experience any growth in voucher households. These areas also witnessed an increase in poverty rate between 2000 and 2010.

	%population under		%population under		%population under			%population under				
Natural		poverty		pov	erty (Wh	ite)	pov	erty (B	lack)	pover	ty (Hispa	nic)
hazard and			(+/-)			(+/-)			(+/-)			(+/-)
HCV growth	2000	2010	%	2000	2010	%	2000	2010	%	2000	2010	%
Nat. Haz.												
Areas with	16 57	<u> </u>	675	5 99	12 40	6 67	5 25	6.64	1 30	7 97	12 01	6.04
increased	10.57	23.32	0.75	5.00	12.49	0.02	5.25	0.04	1.39	1.01	13.71	0.04
HCV												
Other areas												
with	12.00	16.42	2 27	5.24	0.02	2 00	2 71	4 57	0.96	5 00	0.50	2 (2
increased	13.06	16.43	3.37	5.24	8.23	2.99	3.71	4.57	0.86	5.88	8.52	2.63
HCV												
Nat. Haz.												
Areas without	19.01	22.24	2 12	7 70	12.09	5 20	4 1 2	4 1 4	0.01	11 54	15 07	2 72
increased	10.91	22.34	5.42	1.10	15.08	5.50	4.12	4.14	0.01	11.34	13.27	5.75
HCV												
Other areas												
without	17.05	10.00	1 77	7.01	11 41	4 40	4 07	2.96	0.06	0.46	10.05	2 70
increased	17.05	18.82	1.//	7.01	11.41	4.40	4.82	3.80	-0.96	9.40	12.23	2.19
HCV												

Table 5.2: Changes in vulnerability by location in natural hazard^{*} areas and HCV growth

*Areas with more than 40% residential land in natural hazard areas

In the same fashion, if we categorize all census tracts based on their natural hazard exposure and growth of LIHTC units, the results are the same. As Table 5.3 shows, natural hazard areas with any growth of LIHTC units had comparatively higher poverty rates in both 2000 and 2010. The overall poverty rate in those areas also increased significantly between 2000 and 2010 compared to other areas.

										%]	populatio	on
Natural	%pop	ulation u	nder	%рорі	ulation (V	White)	%рорі	ulation (I	Black)	(His	panic) u	nder
hazard and	1	poverty		uno	der pover	ty	und	der pove	rty		poverty	
change in			(+/-)			(+/-)			(+/-)			(+/-)
poverty %	2000	2010	%	2000	2010	%	2000	2010	%	2000	2010	%
Nat. Haz.												
Areas with	24.42	24.00	0.90	7 70	17.06	0.47	0.02	11.20	2.46	10.2	20.24	7.00
increased	24.42	54.22	9.80	1.19	17.20	9.47	0.05	11.29	2.40	12.5	20.54	7.99
TCU												
Other areas												
with	10.00	22.02	4.07	5.00	10.01	4.02	0.10	0.00	0.94	751	10.02	2 22
increased	18.00	22.93	4.27	5.98	10.01	4.03	8.12	8.96	0.84	7.51	10.83	3.32
TCU												
Nat. Haz.												
Areas without	17 40	22.21	4 7 4	6.92	10 55	5 72	1 10	5.07	0.50	0.65	14.00	4.62
increased	17.48	22.21	4.74	0.82	12.55	5.75	4.48	5.07	0.59	9.05	14.28	4.03
TCU												
Other areas												
without	12 55	15 70	2.22	5.02	0.00	2.17	2.25	2.07	0.02	7.04	0.25	0.20
increased	13.55	15.78	2.23	5.83	8.99	3.17	3.25	3.27	0.03	7.04	9.35	2.32
TCU												

Table 5.3: Changes in vulnerability by location in natural hazard^{*} areas and LIHTC units growth

*Areas with more than 40% residential land in natural hazard areas

Like natural hazards, areas with technological hazards showed a similar pattern of population growth and social vulnerability. Since both measures of technological hazards showed similar trends, only the results from the power function are presented here. Table 5.4 shows population growth of different areas categorized by their technological hazards (i.e., above the median value of AWPOWR) and increased presence of HCV households. The pattern of population growth observed in all areas is similar to that of natural hazards.

	Population					
			Growth			
Technological Hazard and population growth	2000	2010	(%)			
Tech. Haz. Areas with increased HCV	44,746,302	48,654,600	8.73			
Other areas with increased HCV	167,789,527	226,491,900	34.99			
Tech. Haz. Areas without increased HCV	40,005,169	40,249,300	0.61			
Other areas without increased HCV	85,415,935	92,428,700	8.21			

Table 5.4: Population growth (2000-2010) by location in technological hazards^{*} areas and HCV growth

*Areas with above median value of AWPOWR

In the case of vulnerability measures however, technological hazard areas yielded different results than natural hazard areas. Although hazardous areas with an increase in HCV households had comparatively higher poverty rates than other areas also experiencing HCV growth, the highest poverty rates were found in hazardous areas without any growth in HCV. This finding is consistent with the hypothesis that housing subsidies when placed in hazardous areas significantly increase neighborhood vulnerability compared to other non-hazardous areas also having such subsidies. Although for the overall population the change in poverty rate was higher in non-hazardous areas (4.03 versus 2.93) and the actual poverty rate in 2010 was higher in hazardous areas with HCV growth (21.42% versus 16.16%). However, for the white and Hispanic population living in poverty even the change in the rate of poverty is higher in hazardous areas with HCV growth compared to other areas also experiencing an increase in HCV households.

										%	populatio	on
	%pop	ulation u	ınder	%рорі	ulation (V	White)	%population (Black)			(His	panic) u	nder
Tech. hazard		poverty		un	der pover	ty	uno	ler pove	rty		poverty	
and change in			(+/-)			(+/-)			(+/-)			(+/-)
poverty%	2000	2010	%	2000	2010	%	2000	2010	%	2000	2010	%
Tech. Haz.												
Areas with	10.40	21 42	2.02	7 10	11 12	4.02	5 5 1	5 (0)	0.05	0.25	10 71	2 27
increased	18.49	21.42	2.95	7.10	11.15	4.03	5.54	5.00	0.05	9.55	12.71	5.57
HCV												
Other areas												
with	10.12	16.16	4.02	4.95	0.11	2.20	2 45	4.50	1 1 4	5.05	0.04	2.00
increased	12.13	16.16	4.03	4.85	8.11	3.26	3.45	4.59	1.14	5.25	8.24	2.99
HCV												
Tech. Haz.												
Areas												
without	23.90	26.22	2.32	9.93	15.62	5.69	5.87	5.39	-0.5	15.4	18.68	3.27
increased												
HCV												
Other areas	•											
without	14 42	16.61	2 10	5 00	10.00	4 1 0	4 1 2	2.20	0.0	7 22	10.22	2.00
increased	14.43	10.01	2.19	5.88	10.06	4.18	4.12	3.28	-0.8	1.53	10.32	2.99
HCV												

Table 5.5: Changes in vulnerability by location in technological hazard^{*} areas and HCV growth

*Areas with above median value of AWPOWR

As for the increase of LIHTC units in technological hazard areas, the findings also fit well with the hypothesis of this study. As Table 5.6 shows, hazardous areas with increased LIHTC units had significantly higher poverty rates in both 2000 and 2010 compared to other areas also receiving new LIHTC units during this time period. For the racial and ethnic population subgroups considered the same trend holds with the exception of the black population living in poverty.

					_					%	populatio	on
	%pop	ulation u	ınder	%рорі	ulation (V	White)	%population (Black)			(His	panic) u	nder
Tech. hazard		poverty		une	der pover	ty	uno	ler pove	rty		poverty	
and change in			(+/-)			(+/-)			(+/-)			(+/-)
poverty%	2000	2010	%	2000	2010	%	2000	2010	%	2000	2010	%
Tech. Haz.												
Areas with	21.04	26.66	5 67	7 69	12 49	4 80	7 62	<u> 9</u> 40	0.87	0.72	1471	4.00
increased	21.04	20.00	5.02	7.08	12.40	4.60	7.05	0.49	0.87	9.12	14./1	4.99
TCU												
Other areas	-											
with	18.00	22.20	4 20	5 11	0.64	4 20	0 25	0.21	0.86	6.04	10.15	3 20
increased	18.09	22.39	4.50	3.44	9.04	4.20	0.55	9.21	0.80	0.94	10.15	5.20
TCU												
Tech. Haz.	-											
Areas												
without	21.05	22.91	1.87	8.59	13.32	4.73	5.30	4.83	-0.5	12.7	15.57	2.85
increased												
TCU												
Other areas	-											
without	12.00	15.04	2.04	5 1 5	0 10	2 22	206	2 10	0.22	5 70	0 50	2 90
increased	12.00	13.04	5.04	5.15	0.48	5.55	2.00	5.18	0.55	3.18	0.38	2.80
TCU												

Table 5.6: Changes in vulnerability by location in technological hazard^{*} areas and LIHTC units growth

*Areas with above median value of AWCPE

5.6 Spatial Regression Model for Social Vulnerability

The exploration of changes in social vulnerability with respect to hazard exposure and housing subsidies in Harris County as presented in previous section appear to confirm the hypothesis of this study that housing subsidies exacerbate social vulnerability when located in hazardous areas, controlling for background or baseline rates of vulnerability attributable to population growth. This finding conflicts with the central objectives of the subsidy programs to revitalize depressed neighborhoods and to provide safer housing to low-income people. As this study posits, present attempts to achieve these objectives through the provision of market dependent housing subsidies is likely to fail, particularly in hazard exposed neighborhoods. To confirm these results, a spatial regression analysis was performed (as explained in Chapter 3). This kind of regression analysis not only controls for other factors that may contribute to social vulnerability, but also can account for spatial autocorrelation of the dependent variables. All four dependent variables considered in this study (Table 5.7) were found to exhibit significant spatial autocorrelation, warranting spatial regression analysis. In addition to standard ordinary least squares (OLS), spatial lag, spatial error, and spatial Durbin (SDM) models were estimated to compare their results. Each of these model results along with the Lagrange Multiplier (LM) diagnostics (Anselin, 1988; Anselin et al., 1996) for model selection are presented in Appendix C. Following the procedures suggested by Elhorst (2010), likelihood ratio (LR) tests were conducted to identify the model that best describes the data²⁴.

Identifying neighboring areas and assigning spatial weights for those neighboring areas is a key issue for spatial regression. The spatial weight matrix specifies which neighboring tracts are most important in defining the characteristics of an area and thereby formally articulates the spatial dependence relationships assumed by the regression analysis. Neighborhood change studies have documented how areas surrounding a neighborhood influences white flight (Crowder & South, 2008; Denton & Massey, 1991), minority composition (Denton & Massey, 1991; Massey & Mullan, 1984), housing values and appreciation rates (Burnell, 1988; Sampson et al., 1999), and loss of population (Morenoff & Sampson, 1997). Studies of subsidized housing have also shown spill-over effects on surrounding neighborhoods (Galster, 2013; Galster et al., 1999; Schill & Wachter, 1995). These studies indicate spatial dependence in the characteristics of a given neighborhoods on the characteristics of neighboring areas and the spatial matrix is designed to capture this influence. Since spatial dependence tends to decline with distance (Downey, 2006), an inverse distance weighting strategy is adopted for this study in which the influence of neighboring census tracts is assumed to be inversely related to distance. This distance-decay strategy defines the elements of the spatial weights matrix as $W_{ii} = 1/d_{ii}$ where d_{ii} is the geographic distance²⁵ between the centroid of tract i and the centroid of the neighboring tract *j*. In this case another critical factor is the distance beyond which a zero influence will be assumed (i.e., threshold distance for negligible influence). Voss & Chi (2006) tested multiple distance thresholds (and matrices) and selected the one that achieved a high coefficient of spatial

²⁴ See Chapter 3 section 3.3.2 for detail procedure followed in this study. Elhorst (2010) gives elaborate discussion.

²⁵ This distance can also be powered by two or more value.

autocorrelation along with a high level of statistical significance, while Crowder & South (2008), selected a much larger distance (100 miles) without a clear theoretical foundation²⁶. In this study, a smaller threshold distance (eight miles) is selected, which was found to be the minimum distance (for Harris County) to assign all census tracts at least one neighbor (i.e., eight miles is the maximum distance between all the tract centroids) and which also gives the highest coefficient of spatial autocorrelation²⁷. Since the traditional adjacency approach ignores tracts that do not share a boundary, a distance decay approach offers important theoretical and practical advantages. While, both approaches emphasize neighboring tracts, the distance-decay function produces a more realistic spatial pattern of neighborhood characteristics and residential location choice. This approach is consistent with the argument that householders consider conditions in a broad range of geographic areas when making their residential location choices (Crowder & South, 2008; Krysan, 2008).

Table 5.7 lists all variables used for the regression analysis. Change in social vulnerability (Δ SoVI) between 2000 and 2010 is one of the key dependent variables and is used to model how the independent variables are contributing to overall changes of vulnerability throughout Harris County. This variable is calculated from the SoVI values (as described in Chapter 4), by subtracting the standardized value of the 2000 SoVI from that of 2010. It should be noted that because the SoVI is a composite measure of social vulnerability, it may mask changes in specific dimensions of vulnerability. Similarly, as explained in Chapter 4 (section 4.7), temporal comparison of the SoVI can be problematic since the composition of vulnerability dimensions (and variables) may change over time. Considering these limitations of the SoVI, individual dimensions of social vulnerability are also explored here alongside changes in the SoVI. The poverty rate is one of the key measures of social vulnerability (as also found through the PCA results discussed in Chapter 4) and change in the poverty rate between 2000 and 2010 $(\Delta poverty_rate)$ is evaluated to determine how subsidized housing and hazard exposure interacted with the overall poverty rate at the census tract level. Since the experience of poverty is also not the same across all population subgroups (Lens et al., 2011), the change in the percentage of the population who are black and living in poverty (Δ black_poverty) as well as

²⁶ Spatial weights by inverse distance are quite small beyond distance of about 10 miles, as contended by Crowder & South (2008) in support of taking 100 mile distance threshold.

²⁷ It is quite obvious that higher distance threshold will produce lower spatial autocorrelation, but in this case the minimum distance was taken that would assign all neighborhood with at least one neighborhood and would better help to capture the spatial spill over patterns of neighborhood characteristics through distance decay function.

Hispanic and living in poverty (Δ hispanic_poverty) is also examined to determine if poor people within these subgroups are experiencing differential outcomes.

	Table 5.7: Variables used for the models
Dependent Variables	
ΔSoVI	Change in standardized values of SoVI (2000-2010)
∆poverty_rate	Change in % of population under poverty (2000-2010)
Δ black_poverty	Change in % of population who are Non-Hispanic Black and under poverty (2000-2010)
∆hispanic_poverty	Change in % of population who are Hispanic and under poverty (2000-2010)
Independent Variables	
zpowr*pchcvgr	Standardized value of technological exposure from power function*change in % of households with HCV
zpowr*pctcugr	Standardized value of technological exposure from power function*change in % of households in LIHTC units
zcpe1*pchcvgr	Standardized value of technological exposure from CPE function* change in % of households with HCV
zcpe1*pctcugr	Standardized value of technological exposure from CPE function* change in % of households in LIHTC units
zfldrt*pchcvgr	Standardized value of % residential lot in natural hazard area* change in % of households with HCV
zfldrt*pctcugr	Standardized value of % residential lot in natural hazard area* standardized growth of LIHTC units
pchcvgr	change in % of households with HCV (2000-2010)
pctcugr	change in % of households in LIHTC units (2000-2010)
zfldrt	Standardized value of % residential lot in natural hazard area
zpowr	Standardized value of technological exposure from power function
zcpe1	Standardized value of technological exposure from CPE function
Control variables	
Popgr	% growth of population (2000-2010)
Blkgr	Change in % of Black population (2000-2010)
Hspgr	Change in % of Hispanic population (2000-2010)
pov2000	% of population under poverty in 2000
hsp2000	% of Hispanic population in 2000
blk2000	% of Non-Hispanic Black population in 2000
mdrnt2000	Median rent of renter-occupied units in 2000
cbd_dist	Distance from CBD in miles

Among the independent variables, the key variables of interest for this study are the interaction terms for hazard exposure and housing subsidy. If those variables are statistically significant and carry a positive coefficient, this would confirm the hypothesis of this study. Each of the hazard exposure and housing subsidy variables were standardized (using the z-score transformation²⁸) before incorporating them in the regression analysis to enhance the interpretability of the regression coefficients (Schielzeth, 2010). Centering variables through standardization is important when the interaction of two continuous variables is included in a model, because without centering the input variables will result in an interaction predictor that is collinear with the main effects (Schielzeth, 2010). Standardization of the input variables before estimating the model will largely eliminate this problem of correlation (Aiken & West, 1991; Neter et al., 1999) and also allows tests for interaction effects²⁹ (Schielzeth, 2010). For the present analysis, the estimated coefficient for the interaction term will indicate if the combination of hazard risk and housing subsidy significantly impacts social vulnerability beyond the individual effect of these two variables that are estimated as main effects. Since this study adopted two different measures of technological hazards (i.e., the power function and WCPE function), all the models are estimated for both of these measures separately.

Several control variables were included in the models that may influence the changes of social vulnerability indicators of an area. The growth of the total population and growth of different racial and ethnic groups are included, assuming that they may influence the percentage of the population under poverty in an area. Base year poverty (pov2000) and racial/ethnic composition (blk2000 and hsp2000) are included in the model because prior studies have shown that poor people usually find it easier to move into areas that already have high poverty rates and high minority concentrations (Galster et al., 2008; Logan & Zhang, 2010). The median rent of housing units in 2000 (mdrnt2000) represents the affordability of an area, while distance from the central business district (cbd_dist) indicates accessibility to employment opportunities, which can also be important decision factors whether people living in poverty will move to a certain area (Brueckner & Rosenthal, 2009). After controlling for all these relevant factors, it can be expected that the independent variables will capture their net effects on the dependent variables. Procedures for preparing all the data used in the models are discussed in detail in Chapter 3.

²⁸ The z-score is derived as follows: $Z_{Tract} = (Score_{Tract} - Mean_{county}) / Standard Deviation_{county})$

²⁹ For example, if combinations of the two variables produce differential responses in addition to what is explained by the sum of the main effects.

5.7 Model Results Discussion

Model estimates for change in social vulnerability (Δ SoVI) in the census tracts between 2000 and 2010 are presented in Table 5.8. The Lagrange Multiplier (LM) diagnostics of the OLS models (Appendix Table C2 and C5, for both measures of technological hazard) indicated that both a spatial lag and error model specification can be selected for this variable, but the Likelihood Ratio (LR) test (Table C3 and C6) recommended by Elhorst (2010) indicated the Spatial Durbin Model (SDM) would be a better choice. As a result, OLS and SDM estimates are presented in Table 5.8 for both measures of technological hazard. As the OLS model results indicate, tracts with an increase in the percentage of households with HCV (pchcvgr) have experienced an increase in social vulnerability between 2000 and 2010. Although the SDM using the power function did not find a significant relationship, the SDM using the WCPE function did and for all models the interaction terms did not indicate any significant influence on changes in social vulnerability at the census tract level. As discussed in previous section, since the SoVI is a composite measure of vulnerability, it may mask the impacts of specific dimensions of social vulnerability. While the models for change in SoVI (Table 5.8) indicate the outcomes for the overall vulnerability pattern, it may also have suppressed underlying heterogeneity for different dimensions of vulnerability.

		(/		
	Power function for	tech. hazard	WCPE function	for tech. hazard
Variables	OLS	Spatial Durbin	OLS	Spatial Durbin
(Intercept)	-0.472 (0.194)*	-0.192 (0.506)	-0.492 (0.195)*	-0.277 (0.503)
zfldrt:pchcvgr	-0.013 (0.018)	-0.007 (0.017)	-0.013 (0.018)	-0.009 (0.017)
zfldrt:pctcugr	0.004 (0.007)	0 (0.007)	0.004 (0.007)	0.002 (0.007)
pchcvgr:zpowr	-0.007 (0.011)	-0.012 (0.011)		
pctcugr:zpowr	-0.008 (0.01)	-0.014 (0.009)		
pchcvgr:zcpe1			-0.016 (0.011)	-0.019 (0.011)
pctcugr:zcpe1			-0.015 (0.012)	-0.02 (0.012)
Pchcvgr	0.046 (0.015)**	0.028 (0.014)	0.048 (0.015)**	0.03 (0.014)*
Pctcugr	0.002 (0.005)	0.001 (0.005)	0.001 (0.005)	0 (0.005)
zfldrt	-0.012 (0.029)	-0.036 (0.033)	-0.01 (0.029)	-0.028 (0.033)
zpowr	-0.075 (0.036)*	-0.005 (0.044)		
zcpe1			-0.005 (0.036)	0.062 (0.039)
Popgr	0 (0)	0 (0)	0 (0)	0 (0)
Blkgr	0.03 (0.004)***	0.026 (0.004)***	0.03 (0.004)***	0.026 (0.004)***

Table 5.8: Models estimating change in standardized value of SoVI between 2000 and 2010 $(\Delta SoVI)$

	Power function	for tech. hazard	WCPE function	for tech. hazard
Variables	OLS	Spatial Durbin	Variables	OLS
Hspgr	0.033 (0.003)***	0.029 (0.003)***	0.033 (0.003)***	0.029 (0.003)***
mdrnt2000	0 (0)	0 (0)*	0 (0)	0 (0)*
pov2000	-0.021 (0.005)***	-0.018 (0.005)***	-0.022 (0.005)***	-0.017 (0.005)**
hsp2000	0.01 (0.002)***	0.012 (0.003)***	0.009 (0.002)***	0.013 (0.003)***
shrblk2000	0.01 (0.002)***	0.015 (0.003)***	0.01 (0.002)***	0.015 (0.003)***
cbd_dist	0.01 (0.006)	-0.026 (0.045)	0.01 (0.006)	-0.033 (0.045)
lag.pchcvgr		- 0.006 (0.061)		0.009 (0.06)
lag.pctcugr		0.019 (0.027)		0.008 (0.028)
lag.zfldrt		0.161 (0.097)		0.123 (0.097)
lag.zpowr		0.022 (0.152)		
lag.zcpe1				-0.193 (0.187)
lag.popgr		0 (0.001)		0 (0.001)
lag.blkgr		0.015 (0.013)		0.016 (0.013)
lag.hspgr		-0.003 (0.009)		-0.008 (0.009)
lag.mdrnt2000		0.001 (0.001)		0.001 (0)
lag.pov2000		0.012 (0.016)		0.009 (0.016)
lag.hsp2000		-0.01 (0.006)		-0.008 (0.006)
lag.shrblk2000		- 0.017 (0.006)**		-0.014 (0.006)*
lag.cbd_dist		0.013 (0.05)		0.021 (0.05)
lag.zfldrt:pchcvgr		0.087 (0.072)		0.088 (0.071)
lag.zfldrt:pctcugr		0.002 (0.033)		0.019 (0.031)
lag.pchcvgr:zpowr		0.003 (0.061)		
lag.pctcugr:zpowr		-0.062 (0.061)		
lag.pchcvgr:zcpe1				0.024 (0.072)
lag.pctcugr: zcpe1				-0.094 (0.091)
rho/lambda		0.516 (0.076)***		0.518 (0.075)***
Adj.R2	0.2896		0.2834	
AIC	1780.967	1719.9	1787.633	1717.7
Log likelihood	-872.4837	-824.9665	-875.8167	-823.8262
Moran's I for	0.115444		0.105.000	
residuals $*n < 05 **n < 01$	$\frac{0.11}{^{***}}$		0.125***	

Table 5.8 (cont.)

OLS and SDM estimates for the change in poverty rate between 2000 and 2010 variable (Δ poverty_rate) are presented in Table 5.9 while the full model results including LM diagnostics and LR test results are presented in Appendix Tables C7 to C12. While the LM test favored the spatial error model specification, the LR test again identified the SDM as the preferred alternative. As the model results indicate (Table 5.9), for both measures of technological hazard

the interaction of tax credit units (i.e., LIHTC units) and technological hazards (pctcugr: zpowr and pctcugr: zcpe1) are statistically significant predictors for an increase in poverty rate. This effect is in addition to the positive effect of the growth of such subsidized units (pctcugr) considered individually (i.e., its main effect). However, for HCV households or natural hazard exposure, the models do not indicate any significant influence. On the other hand, while technological hazard (zcpe1 and zpowr) itself is not significantly related to change in poverty rate, when those hazardous areas experience an increase in the number of tax credit units (pctcugr: zpowr and pctcugr: zcpe1), poverty rates increase significantly in the neighborhood. This result confirms the key hypothesis of this study that subsidized housing contributes to increased social vulnerability in an area, although it was only found to be significant for tax credit units (i.e. LIHTC) and not for vouchers (i.e. HCV).

Table 5.9: Models estimating change in % of population under poverty between 2000 and 2010

	Power function	for tech. hazard	WCPE function	n for tech. hazard
Variables	OLS	Spatial Durbin	OLS	Spatial Durbin
(Intercept)	-0.146 (1.822)	2.308 (4.876)	-0.173 (1.821)	0.524 (4.846)
zfldrt:pchcvgr	-0.158 (0.172)	-0.181 (0.167)	-0.122 (0.172)	-0.159 (0.166)
zfldrt:pctcugr	0.029 (0.068)	0 (0.065)	-0.002 (0.067)	-0.013 (0.064)
pchcvgr:zpowr	-0.017 (0.104)	-0.12 (0.102)		
pctcugr:zpowr	0.278 (0.093)**	0.212 (0.089)*		
pchcvgr:zcpe1			-0.096 (0.1)	-0.105 (0.109)
pctcugr:zcpe1			0.38 (0.116)**	0.341 (0.115)**
Pchcvgr	0.343 (0.141)*	0.256 (0.139)	0.308 (0.141)*	0.239 (0.138)
Pctcugr	0.154 (0.051)**	0.142 (0.049)**	0.167 (0.051)**	0.155 (0.05)**
zfldrt	0.551 (0.272)*	0.304 (0.319)	0.559 (0.272)*	0.324 (0.32)
zpowr	-0.849 (0.341)*	0.219 (0.421)		
zcpe1			-0.181 (0.333)	0.359 (0.38)
Popgr	-0.01 (0.003)***	-0.01 (0.003)**	-0.01 (0.003)***	-0.01 (0.003)**
Blkgr	0.174 (0.037)***	0.22 (0.039)***	0.171 (0.037)***	0.217 (0.039)***
Hspgr	0.312 (0.027)**	0.334 (0.032)***	0.317 (0.028)***	0.336 (0.032)***
mdrnt2000	-0.003 (0.002)	-0.003 (0.002)	-0.002 (0.002)	-0.003 (0.002)
pov2000	-0.274 (0.046)***	-0.288 (0.051)***	-0.279 (0.046)***	-0.283 (0.051)***
hsp2000	0.139 (0.018)***	0.184 (0.027)***	0.135 (0.018)***	0.183 (0.028)***
shrblk2000	0.09 (0.018)***	0.166 (0.029)***	0.09 (0.018)***	0.163 (0.029)***
cbd_dist	0.062 (0.052)	-0.305 (0.439)	0.071 (0.052)	-0.256 (0.439)
lag.pchcvgr		0.068 (0.589)		0.324 (0.581)

 $(\Delta poverty_rate)$

	Power function	on for tech. hazard	WCPE func	tion for tech. hazard
Variables	OLS	Spatial Durbin	Variables	OLS
lag.pctcugr		-0.079 (0.261)		-0.084 (0.273)
lag.zfldrt		1.155 (0.938)		0.785 (0.938)
lag.zpowr		-1.308 (1.481)		
lag.zcpe1				-3.002 (1.815)
lag.popgr		0.01 (0.009)		0.012 (0.009)
lag.blkgr		-0.123 (0.12)		-0.092 (0.121)
lag.hspgr		-0.061 (0.091)		-0.083 (0.092)
lag.mdrnt2000		0.002 (0.005)		0.003 (0.005)
lag.pov2000		0.206 (0.152)		0.19 (0.152)
lag.hsp2000		-0.137 (0.062)*		-0.122 (0.061)*
lag.shrblk2000		-0.209 (0.062)***		-0.182 (0.061)**
lag.cbd_dist		0.233 (0.481)		0.179 (0.479)
lag.zfldrt:pchcvgr		1.581 (0.694)*		1.616 (0.691)*
lag.zfldrt:pctcugr		-0.067 (0.319)		0.126 (0.3)
lag.pchcvgr:zpowr		-0.136 (0.588)		
lag.pctcugr:zpowr		-0.565 (0.59)		
lag.pchcvgr:zcpe1				0.252 (0.701)
lag.pctcugr: zcpe1				0.204 (0.88)
Rho		0.284 (0.091)**		0.316 (0.089)***
Adj.R2	0.2872		0.2869	
AIC	5222.408	5198.1	5222.682	5198.1
Log likelihood	-2593.204	-2564.032	-2593.341	-2564.033
Moran's I for residuals	0.069***		0.075***	

Table 5.9 (cont.)

p*<.05, *p*<.01, ****p*<.001

Since the experience of poverty is not the same across population subgroups (Lens et al., 2011), separate models were estimated for change in the percentage of population who are black and living in poverty (Δ black_poverty) as well as Hispanic and living in poverty rate for technological hazard and LIHTC units, for the black population (under poverty), as Table 5.10 shows, no such influence was found. This measure in particular follows the overall change in social vulnerability, indicating that an increase of voucher recipient households (pchcvgr) in a census tract also increases the percentage of poor black residents there. This pattern persists even after controlling for the overall trends of black population (blkgr) growth during the same period and similar results were found for both measures of technological hazards. The full model

results, including LM diagnostics and LR test results for this variable are presented in Appendix Table C13 to C18.

	Power function	for tech. hazard	WCPE function	for tech. hazard
Variables	OLS	Spatial Durbin	OLS	Spatial Durbin
(Intercept)	0.162 (0.886)	2.481 (2.365)	0.181 (0.885)	2.407 (2.346)
zfldrt:pchcvgr	0.072 (0.084)	0.069 (0.081)	0.08 (0.083)	0.082 (0.08)
zfldrt:pctcugr	-0.037 (0.033)	-0.045 (0.032)	-0.043 (0.032)	-0.046 (0.031)
pchcvgr:zpowr	-0.016 (0.05)	-0.054 (0.05)		
pctcugr:zpowr	0.054 (0.045)	0.038 (0.043)		
pchcvgr:zcpe1			-0.01 (0.049)	-0.045 (0.053)
pctcugr:zcpe1			0.094 (0.056)	0.069 (0.056)
pchcvgr	0.37 (0.069)***	0.296 (0.067)***	0.362 (0.068)***	0.296 (0.067)***
pctcugr	0.026 (0.025)	0.025 (0.024)	0.031 (0.025)	0.026 (0.024)
zfldrt	0.272 (0.132)*	-0.049 (0.155)	0.271 (0.132)*	-0.046 (0.155)
zpowr	-0.044 (0.166)	0.328 (0.204)		
zcpe1			-0.006 (0.162)	0.283 (0.184)
popgr	-0.003 (0.001)*	-0.003 (0.001)	-0.003 (0.001)*	-0.003 (0.001)
blkgr	0.255 (0.018)***	0.273 (0.019)***	0.254 (0.018)***	0.272 (0.019)***
hspgr	0.003 (0.013)	0.003 (0.016)	0.004 (0.013)	0.004 (0.015)
mdrnt2000	-0.001 (0.001)	-0.001 (0.001)	-0.001 (0.001)	-0.001 (0.001)
pov2000	-0.135 (0.022)***	-0.113 (0.025)***	-0.135 (0.022)***	-0.109 (0.025)***
hsp2000	0.053 (0.009)***	0.04 (0.013)**	0.052 (0.009)***	0.042 (0.013)**
shrblk2000	0.043 (0.009)***	0.064 (0.014)***	0.043 (0.009)***	0.064 (0.014)***
cbd_dist	0.003 (0.025)	-0.069 (0.213)	0.004 (0.025)	-0.104 (0.213)
lag.pchcvgr		0.079 (0.288)		0.161 (0.284)
lag.pctcugr		-0.069 (0.126)		-0.16 (0.132)
lag.zfldrt		1.09 (0.455)*		0.988 (0.456)*
lag.zpowr		-0.363 (0.711)		
lag.zcpe1				-0.363 (0.872)
lag.popgr		0.006 (0.004)		0.007 (0.004)
lag.blkgr		-0.173 (0.061)**		-0.17 (0.061)**
lag.hspgr		-0.024 (0.041)		-0.04 (0.042)
lag.mdrnt2000		-0.001 (0.002)		-0.001 (0.002)
lag.pov2000		-0.025 (0.074)		-0.038 (0.074)
lag.hsp2000		0.014 (0.03)		0.017 (0.03)
lag.shrblk2000		-0.07 (0.03)*		-0.065 (0.029)*
lag.cbd_dist		0.039 (0.233)		0.072 (0.232)
lag.zfldrt:pchcvgr		-0.086 (0.336)		-0.01 (0.334)
lag.zfldrt:pctcugr		0.128 (0.154)		0.183 (0.145)

Table 5.10: Models estimating change in % of population Black and under poverty between 2000 and 2010 (Δ black_poverty)
Table 5.10 (Cont.)							
	Power function for tech. hazard		WCPE f	WCPE function for tech. hazard			
Variables	OLS	Spatial Durbin	OLS	Spatial Durbin			
lag.pchcvgr:zpowr		-0.146 (0.285)					
lag.pctcugr:zpowr		-0.295 (0.285)					
lag.pchcvgr:zcpe1				-0.137 (0.339)			
lag.pctcugr: zcpe1				-0.789 (0.424)			
rho/lambda		0.395 (0.085)***		0.4 (0)***			
Adj.R2	0.3327		0.3338				
AIC	4113.338	4087	4112.064	4085.1			
Log likelihood	-2038.669	-2008.487	-2038.032	-2007.56			
Moran's I for							
residuals	0.07***		0.069***				
*n < 05 **n < 01	***n < 0.01						

*p < .05, **p < .01, ***p < .001

In the case of change in the percentage of the poor Hispanic population (Δ hispanic_poverty), the two measures of technological hazard exhibited different patterns of relationships with the independent variables. As Table 5.11 shows, for interaction between technological hazard and an increase in LIHTC units the power function did not indicate any significant contribution (pctcugr:zpowr), but the WCPE function (pctcugr:zcpe1) model specification suggested a positive relationship between LIHTC units and poor Hispanic residents. This result persists after controlling for the growth of the Hispanic population (hspgr) during the same time period and given the main effect of LIHTC growth (pctcugr) indicating a significant results from the two interaction terms (pctcugr:zpowr and pctcugr:zcpe1) are primarily due to the different hazard measures produced by the two models and how poor Hispanic people have located themselves with high growth of Hispanic population in Harris County. All the model results, including LM diagnostics and LR test results for this variable are presented in Appendix Table C19 to C24. In this case too, although LM tests favored a spatial error model specification, the LR test identified SDM as a better alternative.

	Power function for tech. hazard		WCPE function for tech. hazard	
Variables	OLS	Spatial Durbin	OLS	Spatial Durbin
(Intercept)	-1.812 (1.401)	-0.432 (3.794)	-1.887 (1.402)	-2.05 (3.771)
zfldrt:pchcvgr	-0.2 (0.132)	-0.242 (0.13)	-0.175 (0.132)	-0.229 (0.129)
zfldrt:pctcugr	0.043 (0.052)	0.022 (0.051)	0.023 (0.051)	0.015 (0.05)
pchcvgr:zpowr	-0.028 (0.08)	-0.099 (0.079)		
pctcugr:zpowr	0.182 (0.071)*	0.135 (0.069)		
pchcvgr:zcpe1			-0.092 (0.077)	-0.075 (0.084)
pctcugr:zcpe1			0.258 (0.089)**	0.249 (0.09)**
Pchcvgr	0.004 (0.109)	0 (0.108)	-0.013 (0.108)	-0.008 (0.107)
Pctcugr	0.097 (0.039)*	0.084 (0.038)*	0.107 (0.039)**	0.097 (0.038)*
Zfldrt	0.404 (0.209)	0.458 (0.248)	0.408 (0.209)	0.475 (0.248)
Zpowr	-0.772 (0.262)**	-0.042 (0.327)		
zcpe1			-0.273 (0.256)	0.073 (0.295)
Popgr	-0.005 (0.002)*	-0.006 (0.002)*	-0.005 (0.002)*	- 0.006 (0.002)*
Blkgr	-0.017 (0.028)	0.018 (0.03)	-0.019 (0.028)	0.016 (0.03)
Hspgr	0.387 (0.021)***	0.402 (0.025)***	0.39 (0.021)***	0.403 (0.025)***
mdrnt2000	-0.001 (0.001)	-0.001 (0.001)	-0.001 (0.001)	-0.001 (0.001)
pov2000	-0.077 (0.035)*	-0.107 (0.039)**	-0.082 (0.035)*	-0.106 (0.039)**
hsp2000	0.091 (0.014)***	0.136 (0.021)***	0.089 (0.014)***	0.134 (0.021)***
shrblk2000	0.038 (0.014)**	0.085 (0.022)***	0.039 (0.014)**	0.083 (0.022)***
cbd_dist	0.011 (0.04)	-0.072 (0.341)	0.018 (0.04)	-0.007 (0.34)
lag.pchcvgr		-0.074 (0.455)		0.138 (0.448)
lag.pctcugr		-0.048 (0.203)		0.012 (0.212)
lag.zfldrt		-0.208 (0.726)		-0.454 (0.724)
lag.zpowr		-1.166 (1.151)		
lag.zcpe1				-2.99 (1.414)*
lag.popgr		0.005 (0.007)		0.007 (0.007)
lag.blkgr		-0.053 (0.093)		0.022 (0.093)
lag.hspgr		-0.032 (0.077)		-0.039 (0.077)
lag.mdrnt2000		0.001 (0.004)		0.003 (0.004)
lag.pov2000		0.2 (0.117)		0.192 (0.117)
lag.hsp2000		-0.13 (0.048)**		-0.117 (0.047)*
lag.shrblk2000		-0.131 (0.048)**		-0.109 (0.047)*
lag.cbd_dist		-0.007 (0.373)		-0.076 (0.372)
lag.zfldrt:pchcvgr		1.69 (0.54)**		1.653 (0.536)**
lag.zfldrt:pctcugr		-0.103 (0.247)		0.028 (0.232)
lag.pchcvgr:zpowr		-0.093 (0.457)		

Table 5.11: Models estimating change in % of population Hispanic and under poverty between 2000 and 2010 (Δhispanic_poverty)

	Power function for tech. hazard		WCPE function for tech. hazard	
Variables	OLS	Spatial Durbin	Variables	OLS
lag.pctcugr:zpowr		-0.145 (0.458)		
lag.pchcvgr:zcpe1				0.357 (0.544)
lag.pctcugr: zcpe1				0.917 (0.684)
rho/lambda		0.227 (0.095)*		0.243 (0.093)**
Adj.R2	0.4099		0.4081	
AIC	4818.134	4807.5	4820.529	4804.3
Log likelihood	-2391.067	-2368.733	-2392.265	-2367.154
Moran's I for				
residuals	0.059***		0.052***	
$k_{n} < 05 + k_{n} < 01$	***n < 001			

Table 5.11 (cont.)

p*<.05, *p*<.01, ****p*<.001

5.8 Summary of Findings

Although the descriptive analyses at the beginning of this chapter suggested that an increase of both forms of housing subsidy (HCV and LIHTC) in both natural and technological hazard zones contributed to increased social vulnerability, the spatial regression found this relationship to be statistically significant only for an increase of LIHTC units in technological hazard zones. Also, further nuance was discovered when the outcomes for different racial and ethnic subgroups of the population living in poverty were examined. For tracts experiencing an increase in the share of black population living in poverty, none of the interactions (between hazards and housing subsidies) were found to be significant, but for the Hispanic population a significant relationship was found when technological hazards are measured using the WCPE function. The different approaches for providing affordable housing exemplified by the HCV and LIHTC programs may have also contributed to the different outcomes identified in this study. One of the key aspects of the LIHTC program is that it is project based and thereby, tends to concentrate low-income units in a neighborhood. Exploration of the spatial distribution of subsidized housing units also revealed a higher percentage of LIHTC units in the hazard zones. Taken together, these two factors help to explain the positive relationship between social vulnerability and an increase of LIHTC units in technological hazard zones.

For natural hazards, the locations of subsidized housing units did not deviate very much from the distribution of the overall population. Although Harris County exhibits a high degree of social vulnerability in natural hazard areas, because those areas did not experience a significant increase in subsidized housing units, this study also did not find any evidence that the HCV or LIHTC programs are contributing to social vulnerability in natural hazard areas for the time period considered. These findings suggest that while tax credit properties significantly increase social vulnerability when they are located in technological hazard areas, since there were not many new LIHTC developments in natural hazard zones between 2000 and 2010, their contribution to vulnerability in those areas was not found to be significant. The results of this analysis argue for scrutinizing the location requirements of the LIHTC program in terms of neighborhood environmental quality and proximity to hazardous facilities. Chapter 6 discusses existing environmental location requirements of housing subsidy programs, first in general for all HUD assisted housing, and then specifically for the HCV and LIHTC programs. Interviews of officials at the Houston Housing Authority and selected examples of LIHTC developments are also presented in Chapter 6 to further contextualize the findings presented in this chapter.

CHAPTER 6

HOUSING POLICY FOR A MULTI-HAZARD ENVIRONMENT

6.1 Introduction

Housing is more than just a shelter. Besides providing a place to sleep and to store belongings, it plays a vital role in the health and wellbeing of individuals and communities. Housing is an integral part of human identity and shapes the way that current residents and future generations interact with others. Further, housing affordability is one of the key determinants of how individuals and households locate themselves within the housing market, which in turn, shapes a variety of life outcomes. Low-income and marginalized groups have a more difficult time finding and retaining decent quality housing, which would better enable them to overcome the cycle of social vulnerability. Since the 1949 Housing Act when the U.S. Congress envisioned ensuring "a decent home in a suitable living environment for every American family", the federal government has helped to fund the construction and rehabilitation of more than 5 million housing units for low-income households and provided rental vouchers to nearly 2 million additional families (Schwartz, 2010). But these programs have undergone significant changes over time, starting with public housing type direct subsidy provision to the present market dependent approaches of the Low Income Housing Tax Credit (LIHTC) program and Housing Choice Voucher (HCV) program (McClure & Johnson, 2014; Schwartz, 2010). Recently the LIHTC and HCV programs have grown and currently rank as the largest components of the federal portfolio of assisted households and units. The two programs now serve in excess of 2 million households each³⁰ and considering the overall focus of these subsidy programs, prior studies have evaluated their outcomes in terms of whether housing units and households are being placed in low poverty, low crime neighborhoods with better opportunities (Baum-Snow & Marion, 2009; Freeman, 2004; Galster, 2013; Hollar & Usowki, 2007; Lang, 2012; Oakley, 2008). However, very few studies have explored their location outcomes in terms of environmental justice. As discussed in the previous chapter (through the case of Harris County, Texas) these programs are placing proportionally more subsidized housing units and households

³⁰ It is commonplace to add up all of the federal housing programs to generate a total, but this process exaggerates the count somewhat because there is some unknown amount of overlap between the two programs in that voucher households may lease units in LIHTC developments (O'Regan & Horn, 2013). The overlap may be 16% or more of all vouchers (Williamson et al., 2009).

(compared to overall population distribution) in environmentally hazardous areas and are therefore, failing to achieve their desired neighborhood impact (of providing safer places to lowincome people) and in some cases increasing social vulnerability. This chapter argues that these failures are due to oversights in the locational provisions of these programs and discusses policy changes that should be considered to address these issues. It begins with a discussion of lowincome housing and its implications for environmental justice, then explores what the environmental quality provisions of the Department of Housing and Urban Development (HUD) are and why those provisions are not being applied in the LIHTC and HCV programs. Finally, the housing subsidy program administered by Houston Housing Authority is discussed and selected cases of LIHTC properties in Houston that underscore these points are presented.

6.2 Environmental Justice and Low-income Housing

Although the concerns of environmental justice (EJ) have been widely researched, the present provisions for low-income housing and their possible EJ outcomes have garnered less attention. This section briefly discusses EJ theories and then explores why low-income housing subsidies need to consider EJ issues more explicitly. Broadly speaking, EJ theories emphasize social justice aspects of disproportionate environmental quality and illustrate how certain communities face greater environmental risks than others. In the United States, early studies on EJ linked exposure to environmental risks and pollution to both class and race, and explicitly showed that it was not only poor communities that bear disproportionate environmental burdens, but communities of color as well (Buzzelli, 2007; Schlosberg, 2013). Studies have widely documented the nature and extent of disproportionate exposures to health hazards, ranging from toxic waste sites and air pollution to the landfill siting process, and how it varies by population subgroups (Chakraborty & Armstrong, 2001; Crowder & Downey, 2010; Hamilton, 1995; Maantay, 2001). However, EJ scholarship has also moved beyond the simple description and documentation of inequity into a thorough analysis of the underlying reasons for that injustice and specifically, the question of why those communities were devalued in the first place (Schlosberg, 2013). As discussed in Chapter 2 (section 2.4), the racial income inequality thesis (Downey, 2005; Oakes et al., 1996) and the residential discrimination thesis (Bullard, 1999; Mohai & Bryant, 1998) both answer this question following two different perspectives, emphasizing the lower affordability thresholds of minority people and discriminatory housing practices respectively. The location outcomes of housing subsidy programs can be explained by both theses. First, since non-white households are overrepresented in the universe of low-income households, they are more likely to receive a housing subsidy. Second, real estate agents and local governments, even without any intent to discriminate, will locate housing subsidies in low rent and hazardous areas since this is where resistance is likely to be lowest (Mohai et al., 2009). Seicshnaydre (2010) documented this problem in post-Katrina New Orleans recovery process.

One issue that is still being debated in the EJ literature is the establishment of causation through the question of "which came first" (Boone et al., 2014)—whether the polluting facility was already there and then minority population moved in, or whether the areas was already minority dominated before the facility located there. In both cases the issue of EJ holdsminority people moving in near the polluting industry can still be an EJ problem, since the tight housing market plus discrimination and unfair housing practices, might diminish opportunities for minorities to find safer places (Boone et al., 2014). If we allow housing subsidies to be placed in polluted areas then EJ concerns will be intensified, and in this case through an institutional process. Another debated issue in EJ scholarship is identifying the evidence of harm (Bowen, 2002). To ensure the health and wellbeing of nearby residents, determining what a safe distance from the hazardous facility is can be an important piece of information (Brulle & Pellow, 2006; Hynes & Lopez, 2012). While this distance can be measured through a complex epidemiological study and applied for evaluating EJ (Osiecki et al., 2013), there are several strong arguments against this approach (Boone et al., 2014). First, waiting for the scientific demonstration to establish the causal link can be perilous for residents living near a polluting facility and this can also be a time consuming process for planning decision making (e.g., permitting process). Second, the proof of harm should not rest with victims, as is typically the case, but with the polluters themselves. So, there should be defined guidelines for the polluters (or for the developers, who propose to develop any housing project) to show that the polluting facility will not harm the residents. As shown in the next section, HUD already has such regulations for assisted housing, but those are not applicable for the particular case of the HCV or LIHTC programs.

Section 3604(b) of the Fair Housing Act (Title VIII of the Civil Rights Act of 1968) makes it unlawful to "discriminate against any person in the terms, conditions, or privileges of sale or rental of a dwelling, or in the provision of services or facilities in connection therewith."

For the last fifteen years or so, Title VIII has been an area of some commentary in the environmental-justice movement (Kaswan, 1997). Although it has been widely applied and debated in the cases for locating new public housing or during post-disaster recovery process (Rajotte, 2007; Seicshnaydre, 2010), its appropriateness as a general provision for the HCV and LIHTC programs, the problems of EJ, or how these programs may violate section 3604(b) has not yet to received much attention. Since this study does not intend to give an extensive review of legal provisions, some limitations of these popular housing subsidy programs that need further scrutiny are highlighted here without going into specifics of the legislation.

6.3 Environmental Quality Requirements for HUD Assisted Housing

Considering the possible impacts of placing low-income people in hazardous areas, HUD has established detailed guidelines for any HUD assisted housing through *Title 24 Code of Federal Regulations* (CFR), which outlines all the rules and regulation to be followed by HUD officials and for any HUD assisted activity (i.e., those receiving HUD funding). The discussion here is limited to those rules that focus on the environmental requirements of subsidized housing and that should play significant roles in preventing any housing from being placed in natural or technological hazard areas. As it is shown in this section and later (section 6.4 and 6.5), most of these regulations do not apply for the HCV or LIHTC programs since they do not fall under the umbrella of typical HUD assisted housing projects. Despite this fact, these regulations are described to give a better sense of the problems associated with moving to market dependent housing subsidies.

6.3.1 Technological Hazards

The Environmental quality requirements of HUD assisted housing are outlined in 24 CFR §50.3 (2014), which states, "[i]t is the policy of the Department to reject proposals which have significant **adverse environmental impacts** and to encourage the modification of projects in order to enhance environmental quality and **minimize environmental harm**." While it is more concerned with specific impacts of the housing, it also addresses the environmental quality of housing locations in §50.3(i):

"(1) It is HUD policy that all property proposed for use in HUD programs be free of hazardous materials, contamination, toxic chemicals and gasses, and radioactive

substances, where a hazard could **affect the health and safety of occupants** or conflict with the intended utilization of the property.

(2) HUD environmental review of multifamily and non-residential properties shall include evaluation of previous uses of the site and other **evidence of contamination on or near the site**, to assure that occupants of proposed sites are not adversely affected by the hazards listed in paragraph (i)(1) of this section.

(3) Particular attention should be given to any proposed site on or in the general proximity of such areas as dumps, landfills, **industrial sites** or other locations that contain hazardous wastes.

(4) HUD shall require the use of current techniques by qualified professionals to undertake investigations determined necessary."

Particularly, §50.3(i)(3) is specifically mentioning the proximity to hazardous facilities for any HUD assisted housing. Besides specific HUD regulations (discussed later), 24 CFR §50.3(b) gives a list of related federal laws and authority that also guide the HUD assisted housing. Particularly, this section mentions that HUD housing should comply with Flood Disaster Protection Act, The Coastal Zone Management Act, The Safe Drinking Water Act, The Clean Air Act, etc.

24 CFR §51 (Environmental Criteria and Standard) gives the detailed guidelines for environmental quality of HUD assisted housing. Subpart B of §51 outlines regulations for noise abatement and control, while Subpart C gives regulations regarding proximity to technological hazards. In these regulations (Subpart C) the acceptable distance from specific, stationary, hazardous operations (which store, handle, or process hazardous substances) is termed as acceptable separation distances or ASD³¹. Subpart C gives detailed technical guidance for identifying hazard operations, evaluating the anticipated degree of danger, and based on these factors, how to measure ASD for any HUD assisted housing. For calculating ASD, §51.203 (of 24 CFR) dictates:

"The following standards shall be used in determining the acceptable separation distance of a proposed HUD-assisted project from a hazard:

³¹ ASD is defined as "the distance beyond which the explosion or combustion of a hazard is not likely to cause structures or individuals to be subjected to blast overpressure or thermal radiation flux levels in excess of the safety standards in § 51.203. The ASD is determined by applying the safety standards established by this subpart C to the guidance set forth in HUD Guidebook, "Siting of HUD-Assisted Projects Near Hazardous Facilities."

(a) Thermal Radiation Safety Standard. Projects shall be located so that:

(1) The allowable thermal radiation flux level at the building shall not exceed 10,000 BTU/sq. ft. per hr.;

(2) The allowable thermal radiation flux level for outdoor, unprotected facilities or areas of congregation shall not exceed 450 BTU/sq. ft. per hour.

(b) Blast Overpressure Safety Standard. Projects shall be located so that the maximum allowable blast overpressure at both buildings and outdoor, unprotected facilities or areas shall not exceed 0.5 psi.

(c) If a hazardous substance constitutes both a thermal radiation and blast overpressure hazard, the ASD for each hazard shall be calculated, and the larger of the two ASDs shall be used to determine compliance with this subpart.

(d) Background information on the standards and the logarithmic thermal radiation and blast overpressure charts that provide assistance in determining acceptable separation distances are contained in appendix II to this subpart C."

Appendix I and II of subpart C (of 24 CFR §51) gives list of hazardous liquids and gases, and detailed calculation methods for determining ASD. Appendix D of this dissertation presents those appendices that can be used by planners for determining the acceptable distance from hazardous facilities (as required by HUD) for locating any assisted housing.

Two limitations in the approach for calculating ASD can be identified. The first involves the amount of hazardous material stored at certain point in time and the type of material. As the TRI data shows, the amount and type of hazardous material for certain facility can change over time and 24 CFR §51.203 (Safety Standards) does not give any guidance on how to calculate ASD in such a case. Second, it only considers the location of hazardous facilities, not the access road to the facility (although 'road' can come in as part of noise level mentioned in Subpart B of 24 CFR §51). Carrying hazardous materials to the facility can pose threat to the assisted housing if the access road is located nearby. These regulations are detailed enough to be followed in case of location decisions for any HUD assisted housing, but are most often applied to HUD assisted public housing or any other project-based developments and not for tax credit properties or vouchers (as discussed in section 6.4 and 6.5).

6.3.2 Flood and Natural Hazards

HUD regulations are also specific enough to strictly control any development in flood or natural hazard areas. 24 CFR §55 gives the required guidelines for developing and maintaining HUD assisted housing located in floodplains. Particularly, §55.1(b) instructs:

"(b) Under section 202(a) of the Flood Disaster Protection Act of 1973, 42 U.S.C. 4106(a), proposed HUD financial assistance (including mortgage insurance) for acquisition or construction purposes in any "area having special flood hazards" (a flood zone designated by the Federal Emergency Management Agency (FEMA)) shall not be approved in communities identified by FEMA as eligible for flood insurance but which are not participating in the National Flood Insurance Program...."

So, instead of categorically denying any development in flood plain areas, this legislation mandates denial of any subsidized housing in areas which are identified by FEMA to be located in floodplains, but are not participating in the National Flood Insurance Program (NFIP). Instead of giving its own regulation (like technological hazard), in this case the regulations mostly refer to FEMA guidelines. For example, 24 CFR §55.1I(3) instructs to deny any subsidy in these cases:

"(3) Any non-critical action located in a coastal high hazard area, unless the action is designed for location in a coastal high hazard area or is a functionally dependent use. An action will be considered to be designed for location in a coastal high hazard area if:

- (i) In the case of new construction or substantial improvement, the work meets the current standards for V zones in FEMA regulations (44 CFR 60.3(e)) and, if applicable, the Minimum Property Standards for such construction in 24 CFR 200.926dI(4)(iii); or
- (ii) In the case of existing construction (including any minor improvements):

(A) The work met FEMA elevation and construction standards for a coastal high hazard area (or if such a zone or such standards were not designated, the 100-year floodplain) applicable at the time the original improvements were constructed; or

(B) If the original improvements were constructed before FEMA standards for the 100-year floodplain became effective or before FEMA designated the location of the action as within the 100-year floodplain, the work would meet at least the earliest FEMA standards for construction in the 100-year floodplain".

Although this section clarifies that any development in a floodway or coastal high hazard area will not get approval for HUD assistance or will need to meet necessary construction standards (in case of 100-year floodplain), through 24 CFR §55.12(c) it excludes such requirements for any housing voucher. As §55.12(c) mentions:

"(c) This part shall not apply to the following categories of proposed HUD actions:

.

(11) Issuance or use of Housing Vouchers, Certificates under the Section 8 Existing Housing Program, or other forms of rental subsidy where HUD, the awarding community, or the public housing agency that administers the contract awards rental subsidies that are not project-based (i.e., do not involve site-specific subsidies);"

So, the above mentioned regulations regarding flood hazards are mainly for public housing or other project-based HUD assistance and not for housing vouchers or any other development not directly assisted by HUD (like the LIHTC program).

6.4 Environmental Requirements for the HCV Program

Since its inception in the mid-1970s, the Section 8 housing voucher program (later renamed Housing Choice Voucher or HCV program) has grown from a small pilot project to become one of the primary programs for providing housing assistance to low-income households in the United States (Carlson et al., 2012). HUD spends about 19 billion dollars³² every year through the Housing Choice Voucher program to provide rental assistance to over two million households. Moving away from the concept of public housing, this program is premised on the notion that the existing private market will provide an adequate number of quality units and that households will use their rental assistance to locate in quality neighborhoods (Winnick, 1995). Since it is dependent on the private market, this program does not have much control over the location outcomes of voucher recipients. The general expectation in this case is that the voucher will increase the rental affordability of low-income people and will enable them to find better

³² Based on Department of Housing and Urban Development budget outlays by program Comparative summary, fiscal years 2013-2015, available at: http://portal.hud.gov/hudportal/documents/huddoc?id=fy15cj_bdgt_otly_tbl.pdf

places that require payment of higher rent. Through this program, HUD allocates certificates to participating Public Housing Authorities (PHA), which maintains waiting lists of eligible households, ordered according to both local and federal preference criteria. The PHAs allocate the vouchers following the waiting list and the voucher recipient households negotiate a lease with a landlord in the private market under the condition that the rent must be reasonable, the unit must be appropriately sized for the household, and the unit must pass a physical inspection (McClure & Johnson, 2014). In this case the unit rent cannot be more than the fair market rent and the tenant pays between 30 to 40 percent of their annual income for rent.

Since HCV is a rental support program, for determining eligible housing units (to be supported by HCV) it focuses more on the quality of the housing unit itself rather than the neighborhood quality or the distance of the unit from any kind of hazard. 24 CFR Section 982.401- Housing Quality Standard (HQS) defines "standard housing" and establishes the minimum criteria necessary for the health and safety of program participants. It gives detailed requirements for sanitary facilities, space and security, thermal environment, water supply, etc. that must be checked before considering any unit eligible for accepting HCV recipients. HUD developed an inspection manual for conducting this physical inspection for use by the PHA and also provides the necessary Inspection Form (HUD-52580) and Inspection Checklist (form HUD 52580-A). While CFR 24§982.401 gives detailed guidelines for HQS and HUD provides supporting checklists for the physical inspection, the regulations for site and neighborhood quality are comparatively lax. As part of the HQS, §982.401(l) gives this guideline for neighborhood quality of the HCV unit:

"(l) Site and Neighborhood—

(1) Performance requirement. The site and neighborhood must be **reasonably free** from disturbing noises and reverberations and other dangers to the health, safety, and general welfare of the occupants.

(2) Acceptability criteria. The **site and neighborhood** may not be subject to serious adverse environmental conditions, natural or manmade, such as dangerous walks or steps; instability; flooding, poor drainage, septic tank back-ups or sewage hazards; mudslides; abnormal air pollution, smoke or dust; excessive noise, vibration or vehicular traffic; excessive accumulations of trash; vermin or rodent infestation; or fire hazards."

There is considerable room for subjectivity in the evaluation of site quality compared to the detailed guidelines provided for any project-based HUD assistance (as discussed in previous section). HUD can allow PHAs to accommodate variations in the acceptability criteria outlined by HQS to make it compatible with the standards of local housing codes or other codes adopted by the PHA or because of local climatic or geographic conditions. The Houston Housing Authority (HHA) could use this option to impose more restrictions on neighborhood quality requirements considering the density of hazardous facilities in certain areas of this city (as discussed in Chapter 5), but as the administrative plan for HCV³³ adopted by the HHA shows, they have adopted some additional criteria (on top of HUD requirements) but those are also only for unit characteristics³⁴ rather than the site or neighborhood quality of the unit.

As this study showed (Chapter 5), in Harris County, Texas (primarily served by the HHA for housing vouchers), compared to the overall population distribution, a higher proportion of HCV recipients are living in hazardous areas. With high population growth and an increasingly tight housing market, more HCV recipients may have to move into hazardous areas in the absence of any safeguards against such movement. Considering the key focus of HCV program to provide "decent, safe, and sanitary³⁵" affordable housing to low-income people, the focus on the quality of the HCV unit is understandable, but in terms of 'safe' housing this program is failing to fully achieve its objective. One study exploring crime patterns in the neighborhoods of HCV households found that although they tend to live in safer neighborhoods than households in supply-side subsidized housing, these households still lag behind the general population in terms of neighborhood safety (Lens, 2013). Van Zandt & Mhatre (2013) also found that due to a lack of adequate voucher accepting units in safer areas, HCV households tend to live in higher crime neighborhoods. Comparing the locations of voucher households to other households living in project-based housing subsidies (like LIHTC), another study found that schools near voucher holders have lower performing students than the schools near other poor households without a housing subsidy (Horn et al., 2014). Apart from these findings, in general, existing research on the neighborhoods of voucher holders indicates that, on average, voucher holders live in slightly less disadvantaged neighborhoods than other poor households (Pendall, 2000; Wood et al.,

³³ Administrative Plan for Section 8 Housing Programs, Houston Housing Authority (2014), available at: http://www.housingforhouston.com/voucher-program/the-administrative-plan.aspx

³⁴ Page 34 of the Administrative Plan. for Section 8 Housing Programs, Houston Housing Authority (2014)

³⁵ Page 5 of the Administrative Plan. for Section 8 Housing Programs, Houston Housing Authority (2014)

2008), but, as this study shows, if we consider the hazard levels of different neighborhoods, in case of Harris County, Texas, HCV households are not in a significantly better position.

6.5 Environmental Requirements for the LIHTC Program

The LIHTC program was created by the Tax Reform Act of 1986 and is administered by the Department of the Treasury, as opposed to HUD. The portfolio of LIHTC properties grew dramatically over the last 2 decades and the program now supports more than 39,000 projects and 2.4 million units (HUD, 2014). Despite suffering a significant setback with the financial crisis of 2008–2010, the LIHTC program has now become the primary program that adds to the supply of rental housing for low- and moderate-income households (McClure & Johnson, 2014). Under this program, the federal government grants tax credits to states on a per capita basis and the states distribute the tax credit to developers (proposing projects) on a competitive basis. For distributing the tax credits, each state publishes a Qualified Allocation Plan (QAP), which details the criteria for evaluation of the proposed projects. Proposed projects should designate a minimum share of "affordable" units³⁶ within the development and projects that obtain the highest score as per the criteria outlined in QAP to receive the tax credits. After completion of the project, the developer receives a tax credit over a 10-year period, but the project must remain in low- or moderate-income occupancy for at least 15 years (McClure & Johnson, 2014). Developers awarded the credits sell them to investors to pay for some of the project's total development costs³⁷. The tax credit amounts are generally 9% of the non-land development costs and in case it is financed through bonds with interest that is exempt from federal income taxes, it decreases to 4%. To encourage affordable rental developments in difficult development areas (a location with particularly high development costs relative to incomes) or in a qualified census tract (a census tract with a high incidence of low-income households), the tax credit amounts are boosted by an additional 30% if it is located in those areas. Following the scoring criteria of the QAP, developers generally try to maximize the amount of subsidy and as a result they are attracted to sites located in difficult development areas and qualified census tracts, which are often badly deteriorated neighborhoods (Oakley, 2008).

 $^{^{36}}$ To be eligible to apply for the program, developments must have a minimum of 20% of the units affordable to households earning 50% of the metropolitan area's median family income or, a minimum of 40% of the units are affordable to households earning 60% of the metropolitan area's median family income (Schwartz, 2010)s.

³⁷ Additional costs are usually paid via a combination of debt and other subsidy programs. (McClure & Johnson, 2014)

Since the LIHTC program is primarily guided by the QAP of state housing financing agency, to get better idea of their location requirements the state's regulations need to be evaluated. In the case of Texas (where the primary case study Harris County is located), the Texas Department of Housing and Community Affairs (TDHCA) is assigned the responsibility of awarding and allocation of housing tax credits. Following the general guidelines for the LIHTC program discussed above, TDHCA also publishes a QAP every year³⁸ to establish the procedures and requirements relating to an award and allocation of housing tax credits. Besides outlining the procedures for administering the program, it details the scoring criteria upon which the proposed housing developments will be selected for awarding tax credits. As per the 2015 QAP, there are criteria like size and quality of units (15 points), income level of tenants (16 points), rent level of tenants (13 points), tenant services (11 points), and other criteria that would promote community support and engagement. While this scoring criteria favors development in low poverty area (giving 7 points for development in census tracts with poverty rate below 15 percent), it does not have any criteria that would encourage development in safer areas or discourage development in hazardous areas. The only criteria that talk about such hazards are under the broad criteria for community support and engagement, and actually gives additional points (6 points) when the development is located in a hazardous area and there is a community revitalization plan for that area. One of the factors for such revitalization plan is mentioned as:

"adverse environmental conditions, natural or manmade, that are material in nature and are inconsistent with the general quality of life in typical average income neighborhoods. By way of example, such conditions might include significant and recurring flooding, presence of hazardous waste sites or ongoing localized emissions not under appropriate remediation, nearby heavy industrial uses, or uses presenting significant safety or noise concerns such as major thoroughfares, nearby active railways (other than commuter trains), or landing strips; significant and widespread (e.g. not localized to a small number of businesses or other buildings) rodent or vermin infestation acknowledged to present health risks requiring a concerted effort; or fire hazards;"³⁹

This encompasses both natural and technological hazards and supports any revitalization initiative taken by a municipality or county for those hazardous areas. Other than this criterion,

³⁸ 2014 and 2015 QAP of TDHCA are available at: http://www.tdhca.state.tx.us/multifamily/nofas-rules.htm

³⁹ Page 25 of 2015 QAP, available at: http://www.tdhca.state.tx.us/multifamily/docs/15-QAP.pdf

which indirectly supports any clean-up or hazard mitigation activity, the QAP does not give any specific provision that would disincentivize development in the floodplain or near hazardous facilities.

6.6 Subsidized Housing in Houston, Texas

Besides analyzing subsidized housing data and exploring its location in the multi-hazard environment of Harris County (as discussed in Chapter 5), the author visited Houston in Spring 2014 to gain a better sense of the social vulnerability scenarios there and how subsidized housing is being managed, particularly in the city of Houston. Several neighborhoods were visited, both declining and gentrifying, and officials with the Houston Housing Authority (HHA)—the public housing agency that manages housing vouchers in Houston and also maintains some public housing and tax credit properties—were interviewed. The key objective of this visit was to relate the findings of the spatial analysis (presented in Chapter 5) to actual developments on the ground in Houston. Due to time and resource constraints, extensive surveys or interviews could not be conducted, but short visits to selected sites were able to provide a deeper understanding of how the location of housing subsidies interacts with the multi-hazard context of Houston. This section briefly discusses gentrifying and declining neighborhoods in Houston and then describes the key informant interviews with HHA officials focusing on their provision of housing vouchers. Finally, the chapter examines selected housing projects funded through the LIHTC program.

6.6.1 Gentrification and Low-Income Housing

As identified in Chapter 4, certain neighborhoods of Houston have gradually gentrified over time, while others have consistently been characterized by high poverty and high minority concentration (i.e., high social vulnerability). The juxtaposition of neighborhoods with vastly different levels of affordability combined with strong population growth means that low-income households are likely to encounter difficulty in obtaining quality housing in safe neighborhoods. Midtown is an example of a gentrifying neighborhood located just west of Houston Downtown (Figure 6.1) and shown in Photo 6.1. The gentrification process has advanced by demolishing old houses and developing new upscale condominiums on several blocks. In 2010, about 57%⁴⁰ of the population (25 years and above) in this neighborhood had a bachelor's degree or higher

⁴⁰ Data collected from City-Data.com; http://www.city-data.com/neighborhood/Midtown-Houston-TX.html

education, a poverty rate of 17.4% (compared to 23.8% in Houston), and was comprised of about 50% non-Hispanic white population. Compared to the Midtown neighborhood, the Park Place neighborhood located southwest of the downtown (Figure 6.2) is a depressed neighborhood with high concentration of poverty and minority population and also mostly contains dilapidated housing structures (Photo 6.2). In 2010, this neighborhood had only about 27% of its population (25 years or above) with a bachelor's degree or higher education, about 34% population living under the poverty line, and more than 75% Hispanic population.



Photo 6.1: Midtown Neighborhood (near Gray St. and Bagby St.)



Figure 6.1: Map of Midtown neighborhood



Photo 6.2: Park Place Neighborhood (near Broadway St. and Detroit St.)



Figure 6.2: Map of Park Place Neighborhood, Houston

Due to the differences in their physical and socio-economic conditions, these two neighborhoods also exhibit different levels of affordability. In 2011, the median rent in the Midtown neighborhood was \$936, and at the same time the median rent in the Park Place neighborhood was \$575⁴¹. Given these different levels of affordability, poor people coming into the city find it more affordable to move into low rent neighborhoods like Park Place and if the present trend of gentrification continues (which can be expected with population growth), poor people may find it even more difficult to rent places in better neighborhoods even with housing vouchers. These conclusions are echoed in the interviews⁴² conducted with HHA officials.

6.6.2 Housing Vouchers and the HHA

The Houston Housing Authority (HHA) is the public agency in Houston responsible for administering and managing the Housing Choice Voucher (HCV), Moderate Rehabilitation,

⁴¹ Data collected from city-data.com

⁴² Necessary approval from Institutional Review Board (IRB) at the University of Illinois was obtained prior to conducting these interviews.

Single Room Occupancy (SRO), Disaster Voucher, and other Houston Housing Authority Section 8 special programs. In order to learn about the HCV program and how the voucher award process shapes housing outcomes, interviews were conducted with two HHA officials— Senior Policy Advisor Brian Gage, and Policy Analyst Ross LaFour. As they informed in the interviews, HHA currently supports about 17,000 households through its HCV program and owns about 3,000 project-based units (combination of public housing, tax credit projects, and Section 8 projects). The waiting list for awarding HCVs was initiated about two years ago when roughly 80,000 people applied. Through a lottery they selected 20,000 households to support and currently they have about 13,000 households on the waitlist to receive vouchers. The same procedures outlined by HUD (see section 6.4) are used to identify housing units eligible for selection by HCV households. In response to the question of whether HHA is concerned about vouchers being used primarily in low-income neighborhoods and whether HHA takes any measure to recruit landlords in better neighborhoods, the interviewees acknowledged that although they try do this, funding is a substantial obstacle. According to Brian:

"One of the things we have been looking into implementing is, how do we go out to these areas where we have very few voucher holders now and get landlords interested in the program..., but the honest truth of the matter is our funding level for administering the program is declining sharply. In the past three to five years or so they (i.e., HUD) only give us 75 cents on a dollar what is actually cost to run the program. So, doing those extras make it much more difficult to fit in budget-wise. You know, there are lots of things that we know we can do if we had the administrative funding to go out and build our relationships with landlords."

As this response conveys, although the HHA knows about the problem of poverty concentration, budgetary constraints do not allow them to make a concerted effort to find housing units in higher rent areas that would accept housing vouchers. In response to the question of whether vouchers allow or perhaps encourage people to move to better neighborhoods, as suggested in literature (Carlson et al., 2012), the interviewees have not seen significant evidence of this in Houston. As Brian said:

"Yes, we have people that want to stay in their neighborhood, they don't want to move into neighborhood they are not familiar with, because of their churches, kids go to school, because their cousin lives around the corner... because they are comfortable with that. We did a major relocation initiative down here (after demolishing public housing units)... and... I want to say about half stayed in the same zip code although they had the choice to go anywhere in the city. And, there were some who moved, even moved to California, cause you can move anywhere in the country, but they are just one or two individuals..."

According to Brian and Ross, currently the HHA is identifying areas which have fewer vouchers and planning to offer the maximum allowable subsidy (110% of Fair Market Rent) to households moving into those areas. Still, they expressed their concern that such a task will be hard to accomplish since Houston is a majority minority city and there simply are not very many areas with low minority populations and where the vouchers can be allocated.

The Housing Quality Standard (HQS) checklist required by HUD is the basis for evaluating the eligibility of any unit (for giving vouchers), but Brian found this approach too restrictive and he thinks that more flexible inspection protocol would allow more landlords to participate in the program. As per this current checklist, any minor problem may make an otherwise good housing unit ineligible to accept voucher households. However, Brian also acknowledged the fact that due to this strict checklist, people are moving into higher quality units overall. In terms of the location requirement of the HCV housing unit, as mentioned in Section 6.4, the HHA does not follow any particular provision for avoiding hazardous areas, since this is not required by HUD.

6.6.3 Selected Examples of LIHTC Developments

In addition to interviewing HHA officials, a selected number of tax credit developments were visited to get a better sense of such developments and how their neighborhoods fit within the broader context of Houston. During the interviews at the HHA, Brian Gage noted that developing gated communities through the LIHTC is particularly popular in Houston since it gives a sense of security to the prospective tenants. One of the aims of the visits to a few of the LIHTC projects was to see whether that is the case and to observe the degree of integration between those developments and their surrounding neighborhoods. South Houston and Pasadena were chosen since the analysis of natural and technological hazards (presented in Chapter 5) found this part of Houston to be significantly exposed to both kinds of hazards. Most of the census tracts in this part of the city also have higher poverty rates and larger minority

populations. Of the eight LIHTC developments visited, six were gated housing projects. A brief overview of three of those tax credit developments is presented here and the question of how those developments may have been impacting their neighborhoods is considered later. Data on these developments were collected from the Texas Department of Housing and Community Affairs (TDHCA).

Vista Bonita Apartments

Vista Bonita Apartments (Photo 6.3) is a gated community of 118 tax credit units located in South Houston (9313 Tallyho Road, Houston, TX). Due to its location just east of the Gulf Freeway (I-45) it has easy access to Houston downtown and Hobby Airport (Figure 6.3). Approved by the TDHCA in 2008, it is a competitive (9%) Housing Tax Credit (HTC)⁴³ development. The site visit revealed that this project is not assimilating very much with its surrounding neighborhood. There is another gated community (non-LIHTC) just opposite of this development on Tallyho road and other than that, there are few residential developments in its immediate vicinity. The surrounding blocks are mostly commercial and light industrial uses (storage, construction yards, motel, etc.) with some scattered houses, but being a recent development, the units here are in very good condition.

Granada Terrace Apartments

Granada Terrace Apartments (Photo 6.4) is an older LIHTC project also located in South Houston (1301 Avenue A, Houston, TX). The TDHCA awarded tax credit status for this development in 1991 and it has a total of 154 tax credit units. Unlike Vista Bonita Apartments, it is not gated and mostly surrounded by residential development. As an older development, the physical condition of the units does not seem to be very high and most of the houses in the surrounding neighborhood are in a shabby state, suggesting that they are mostly occupied by low-income people.

⁴³ For 9% HTC the tax credit amount is 9% of the non-land development cost



Photo 6.3: Vista Bonita Apartments



Figure 6.3: Location of Vista Bonita Apartments



Photo 6.4: Granada Terrace Apartments



Figure 6.4: Location of Granada Terrace Apartments

Gardenview Apartments

Gardenview Apartments (Photo 6.5) is located in Pasadena (2730 Lafferty Road, Pasadena, TX) and about two miles away from Granada Terrace Apartments. It is also an older development (tax credit status awarded in 1993), but unlike Granada Terrace it is a gated community. It has a total of 309 tax credit units and despite being an older development, most of the units here seem to be in a good condition. A park and an elementary school isolate this project from the single family housing in its neighborhood, which appears to be home to mostly middle-income households. Although this development is somewhat detached from rest of the neighborhood and there are few similar residential developments in its vicinity, there is a mobile home park just south of this project, indicating a concentration of low-income households. So while this project is located in a middle-income neighborhood, the fact that it is a gated community and separated by physical barriers (park and school) means that it is not well integrated with its neighborhood.



Photo 6.5: Gardenview Apartments



Figure 6.5: Location of Gardenview Apartments

The preceding cases focus on one recent LIHTC project and two older projects (one gated and another non-gated). While the recent project was not found to be well integrated with its surrounding neighborhood, the older two projects present interesting examples of how they may be influencing their neighborhoods. The older non-gated community (Granada Terrace) is located in a low-income neighborhood and may have contributed to the changes there over time. Similarly, the older gated community (Grandview Apartments), despite being located in a middle-income neighborhood, may have contributed to the emergence of a low-income community in its immediate vicinity. More detailed case studies may have helped to understand these specific developments, but they also indicate the heterogeneity of location and patterns of tax-credit developments, which makes it harder to isolate the influences of these developments. Still, the popularity of developing gated communities through the LIHTC program invites further scrutiny. For more than a decade scholars have debated the consequences of the socio-economic differentiation and particularly the residential segregation created by the gated communities (Atkinson & Blandy, 2006; Blakely & Snyder, 1997; Gordon, 2004; Le Goix, 2005; Vesselinov, 2008). In terms of property values, while some studies indicate that such gated residential developments can play an instrumental role in avoiding decay and other externalities in a neighborhood (LaCour-Little & Malpezzi, 2001), others have argued that sometimes such development can be detrimental to property values in non-gated developments nearby (Le Goix & Vesselinov, 2013). While this study found that LIHTC developments are associated with increasing vulnerability in hazardous areas, the role played by gated developments in the particular case of Houston warrants further research.

6.4 Opportunities for Policy Intervention

Although HUD has specific environmental regulations in place that would better ensure the safety (i.e., areas less exposed to natural or technological hazards) of low income households, those regulations are not applicable for the two most popular low-income housing programs—the HCV and LIHTC programs. Given that these are market driven approaches operating within a neoliberal policy environment, this lack of regulatory safeguards against natural or technological hazards increases the possibility that they will concentrate low-income people in hazardous areas of the city (as this study found was particularly true for the LIHTC program). As in the case of Houston, this can be more problematic at times when the city is experiencing high rates of population growth alongside an influx of low-income and minority residents. An increasingly competitive housing market may push these socially vulnerable people into hazardous areas and thereby create a breeding ground for catastrophic disaster, whether by natural hazards exemplified by Hurricanes Katrina or Ike, or via a technological hazard like the fertilizer plant explosion in West, Texas⁴⁴. While mitigation of those hazards can be one approach for avoiding such a catastrophic disaster, at the same time we need to examine how our plans and policies are influencing people's location choices within this multi-hazard environment. Several recommendations have emerged from the analysis presented in this dissertation that if implemented, should help to ensure that we are not placing socially vulnerable people in harm's way.

First, despite being a HUD assisted program, the HCV program does not require that the location criteria applicable to other HUD assisted project-based developments be followed. This is an administrative challenge because applying the present checklists for Housing Quality

⁴⁴ The explosion killed 15 and wounded another 226 people in West, Texas. More on this explosion here: http://www.cnn.com/2014/04/22/us/west-texas-fertilizer-plant-explosion-investigation/

Standard (HQS) to identify eligible housing units is already a difficult and time-consuming task for public housing officials (as confirmed by the HHA interviewees). Particularly, conducting case by case inspection of housing units, while ensuring good quality housing for HCV households can become problematic if environmental quality or the hazard exposure of the units must also be considered. Instead, it can be suggested that for such measurement of environmental quality and hazard exposure, public housing authorities can make use of the already available data from U.S. Environmental Protection Agency (EPA) and Federal Emergency Management Agency (FEMA). The EPA provides data on air quality and the location of contaminated sites or facilities handling toxic materials (i.e., TRI sites) while FEMA maintains and distributes data on floodplains and coastal hazard zones. As previously noted, HUD regulations already mandate the use of these data sources for identifying the location of housing projects, which is done on a case by case basis. The issue is that while such a case by case evaluation can be financially and administratively feasible for multi-family units, it is much more difficult for housing vouchers which are more flexible in terms of potential locations and more widely scattered across a broad region. In this case, housing officials can look at neighborhood or census tract data, much like they already do for poverty level or minority concentration, but also incorporate EPA and FEMA data to simultaneously evaluate the environmental quality and hazard exposure of areas under consideration. During the process of allocating vouchers, housing officials can check whether the candidate housing units are located in those census tracts and thereby determine the eligibility of the unit for accepting vouchers.

Second, since the LIHTC program is not administered or assisted by HUD, these developments are not required to follow the regulations like HUD assisted housing projects currently do, but because this program also targets low- and medium-income households, there is no reason why the same environmental regulations cannot be applicable to LIHTC projects as well. It is true that in the absence of any strict guidelines from the federal level the state housing finance agency (e.g., TDHCA in Texas) enjoys much freedom and flexibility to adjust the program requirements to meet local needs. But as it turns out, despite being a state frequently ravaged by different natural and technological hazards, there are not clear and detailed guidelines for locating tax credit properties in this state. The current scoring criteria outlined by the QAP of TDHCA gives scores based on neighborhood socio-economic condition, but there is no specific scoring mechanism that would disincentivize development in hazardous areas. In this case, it can

be suggested that the TDHCA (and for that matter any other housing finance agency in states with a higher frequency of hazards) should take cues from HUD for the environmental requirements of the tax credit properties. Like HUD assisted projects, they can adopt acceptable separation distances (ASD) for locating tax credit properties near hazardous sites and can ask developers to not propose projects (or give negative scores) in areas not participating in the National Flood Insurance Program (NFIP).

Third, the problems inherent in these market dependent housing subsidy approaches need to be reevaluated at the national level, particularly in the present context of high population growth in coastal metro areas which are also experiencing increases in poverty and the spatial concentration of minority populations. Gentrification and increasingly tight housing markets will make it gradually more difficult for HCV households to find safe and secure places to live. While the growth of LIHTC is vulnerable to economic downturns because of its dependence on the tax credit market, in terms of location outcomes it can become concentrated in depressed neighborhood to maximize the tax credit benefits. With the increased price of tax credits, LIHTC units may be able to gradually enter the suburbs (McClure, 2006), but it also indicates the limitation of this approach. How the location of low-income housing will turn out is totally dependent on the market's profitability from such investments. This outcome needs to be analyzed in the broader context of environmental justice to determine if this market dependent subsidy approach will push more low-income minority people in hazardous areas of the city. As the results of this study suggest, this is precisely what is happening in the case of Houston.

With the increased emphasis on climate change and adaptation planning, particularly in coastal metro areas, the provision of low-income housing subsidies needs to be further scrutinized. When these market-dependent approaches fail to provide safe housing to socially vulnerable population groups, any climate adaptation effort that do not consider their housing needs (or pose any threat to existing stock of affordable housing) may turn out to be a mal-adaptive approach in the long term.

CHAPTER 7

CONCLUSION

With recent enthusiasm for urban sustainability and climate change adaptation (Blanco et al., 2009), planners need to be more careful and attentive to the equity outcomes in this policy environment. Climate change is expected to increase the frequency and severity of natural hazards and in the near future more disaster events are a likely outcome. In a multi-hazard urban environment a clear challenge for planners is to ensure that socially marginalized populations are not disproportionately exposed to hazards. Differential hazard exposure and disaster outcomes for vulnerable low-income and minority population groups are already well documented (Crowder & Downey, 2010; Finch et al., 2010; Van Zandt et al., 2012), but how the present pattern of vulnerability has evolved over time and how political-economy of cities may play a role in this process has not been extensively studied. Moving beyond the conventional notion of vulnerability as static and exogenous to urban development and politics, this study demonstrates how social vulnerability has changed over time in three coastal cities and calls for planning approaches that are more responsive to prevailing trends in vulnerability. Through a vulnerability production framing (Dooling & Simon, 2012) this study further explored how our present provision of subsidized low-income housing is failing to provide safer housing, and in some cases, is increasing vulnerability.

Present scholarship on social vulnerability, while highlighting the underlying cause of differential disaster outcomes and rightly identifying the social factors that need to be emphasized, paints an incomplete picture of how vulnerability evolves over time. By integrating neighborhood change theories and social vulnerability theories, this study explores three different coastal cities to explain the dynamics behind the changing patterns of vulnerability observed there. As this study found, despite having drastically different population growth trajectories and being located in different political and economic settings, in recent decades the spatial concentration of social vulnerability has gradually decreased in the study cities. However, in terms of the composition of social vulnerability, the study cities exhibited different trends. In the case of Harris County (Houston) in Texas, the high growth of the Hispanic and immigrant populations in recent decades made these the key factors in social vulnerability, along with the percentage of Black or African-American population and poverty rate. In the case of Orleans

Parish (New Orleans) Louisiana, displacement by Hurricane Katrina has significantly influenced the pattern of social vulnerability, which has become less concentrated and less dominated by race and socio-economic indicators. For Hillsborough County (Tampa) in Florida, gentrification in the inner city areas is pushing socially vulnerable populations to suburban and coastal census tracts and at the same time, some of the coastal locations are experiencing high rates of growth of elderly populations due to the development of retirement communities there. These trends imply drastically different adaptation challenges in these counties.

Analyzing the location outcomes of subsidized housing in Houston, this study found that among the two most popular housing subsidy programs (HCV and LIHTC), the supply-side based subsidy provision of LIHTC program significantly increases neighborhood social vulnerability when it is located in technological hazard areas. In comparison to the total population distribution however, it was found that both of these subsidies have a proportionally higher presence in both natural and technological hazard areas. A potential explanation for the significant contribution of the LIHTC program can be that it is a project based approach and necessarily concentrates low-income units within a given neighborhood. On the other hand, it was found that a substantial number of these projects are gated communities, which may also exert negative neighborhood impacts by devaluing adjacent property values (i.e., spillover effects).

Evaluating the environmental requirements of different subsidized housing programs, this study identified limitations that may have contributed to unfavorable location outcomes from the HCV and LIHTC programs. While HUD has explicit regulations regarding the placement of assisted projects in natural or technological hazard zones, those provisions are not applicable for these two subsidy programs. Since the HCV program is not project based and the LIHTC is administered by the U.S. Treasury Department, the location requirements for traditional public housing established by HUD cannot be directly applied to these programs. Because these are market driven approaches operating within a neoliberal policy environment, this lack of regulatory safeguards increases the possibility that they will concentrate low-income people in hazardous areas of the city (as this study found was particularly true for the LIHTC program). While this study emphasizes reevaluating the provisions of these market-based programs, policy modifications were also proposed that should help to ensure that housing subsidies are not placed in natural or technological hazard zones if implemented.

REFERENCES

- Abt Associates, I. (2006). Are States Using the Low Income Housing Tax Credit to Enable Families with Children to Live in Low Poverty and Racially Integrated Neighborhoods?. Cambridge, MA.
- Adger, W. N. (2006). Vulnerability. *Global Environmental Change*, 16(3), 268–281. doi:10.1016/j.gloenvcha.2006.02.006
- Adger, W. N., Arnell, N. W., & Tompkins, E. L. (2005). Successful adaptation to climate change across scales. *Global Environmental Change*, 15, 77–86. doi:10.1016/j.gloenvcha.2004.12.005
- Aiken, L. S., & West, S. G. (1991). *Multiple regression: Testin and interpreting interactions*. *Multiple regression: Testing and interpreting interactions*.
- Anderson, W. A. (2005). Bringing Children into Focus on the Social Science Disaster Research Agenda. *International Journal of Mass Emergencies and Disasters*, 23(3), 159–175.
- Andrey, J., & Jones, B. (2008). The dynamic nature of social disadvantage: Implications for hazard exposure and vulnerability in Greater Vancouver. *Canadian Geographer*, 52(2), 146–168. doi:10.1111/j.1541-0064.2008.00206.x
- Angotti, T. (2008). *New York for Sale: Community Planning Confronts Global Real Estate*. Cambridge, MA: Massachusetts Institute of Technology.
- Anselin, L. (1988). Spatial Econometrics: Methods and Models. Operational Regional Science Series (p. 289). Retrieved from http://www.amazon.com/Spatial-Econometrics-Methods-Operational-Regional/dp/9024737354
- Anselin, L. (1995). Local indicators of spatial association LISA. *Geographical Analysis*, 27(2), 93–115. doi:10.1111/j.1538-4632.1995.tb00338.x
- Anselin, L. (2003). Spatial Externalities, Spatial Multipliers, And Spatial Econometrics. *International Regional Science Review*, 26, 153–166. doi:10.1177/0160017602250972
- Anselin, L., Bera, A. K., Florax, R., & Yoon, M. J. (1996). Simple diagnostic tests for spatial dependence. *Regional Science and Urban Economics*, 26, 77–104. doi:10.1016/0166-0462(95)02111-6
- Atkinson, R., & Blandy, S. (2006). Introduction: International Perspectives on The New Enclavism and the Rise of Gated Communities. *Housing Studies*, 20(2), 177–186. doi:10.1080/0267303042000331718
- Barnett, J., & O'Neill, S. (2010). Maladaptation. *Global Environmental Change*, 20, 211–213. doi:10.1016/j.gloenvcha.2009.11.004

- Baum-Snow, N., & Marion, J. (2009). The effects of low income housing tax credit developments on neighborhoods. *Journal of Public Economics*, 93(5-6), 654–666. doi:10.1016/j.jpubeco.2009.01.001
- Beer, C., & Riedl, A. (2012). Modelling spatial externalities in panel data: The Spatial Durbin model revisited. *Papers in Regional Science*, 91(2), 299–318. doi:10.1111/j.1435-5957.2011.00394.x
- Bivand, R., & Gebhardt, A. (2000). Implementing functions for spatial statistical analysis using the R language. *Journal of Geographical Systems*, 2, 307–317. doi:10.1007/PL00011460
- Blaikie, P., Cannon, T., Davis, I., & Wisner, B. (1994). At Risk: Natural Hazards, People's Vulnerability and Disasters (1st ed.). London: Routledge.
- Blakely, E. J., & Snyder, M. G. (1997). *Fortress America: Gated communities in the United States*. Washington, DC: Brookings Institution Press.
- Blanchard-Boehm, R. D. (1998). Understanding Public Response to Increased Risk from Natural Hazards. *International Journals of Mass Emergencies and Disasters*, *16*(3), 247–278.
- Blanco, H., Alberti, M., Olshansky, R., Chang, S., Wheeler, S. M., Randolph, J., ... Watson, V. (2009). Shaken, shrinking, hot, impoverished and informal: Emerging research agendas in planning. *Progress in Planning*, 72(4), 195–250. doi:10.1016/j.progress.2009.09.001
- Bohle, H. G., Downing, T. E., & Watts, M. J. (1994). Climate change and social vulnerability: Toward a sociology and geography of food insecurity. *Global Environmental Change*, 4(1), 37–48. doi:http://dx.doi.org/10.1016/0959-3780(94)90020-5
- Bolin, B., & Stanford, L. (1998). *The Northridge earthquake: Vulnerability and disaster*. London: Routledge.
- Boone, C. G., & Fragkias, M. (2013). Connecting Environmental Justice, Sustainability, and Vulnerability. In C. G. Boone & M. Fragkias (Eds.), *Urbanization and Sustainability: Linking Urban Ecology, Environmental Justice and Global Environmental Change* (pp. 49– 59). Dordrecht: Springer Netherlands. doi:10.1007/978-94-007-5666-3
- Boone, C. G., Fragkias, M., Buckley, G. L., & Grove, J. M. (2014). A long view of polluting industry and environmental justice in Baltimore. *Cities*, 36, 41–49. doi:10.1016/j.cities.2013.09.004
- Borja, J., & Castells, M. (1997). *Local and Global: Management of Cities in the Information Age*. London, UK: Earthscan Publications.
- Bowen, W. (2002). An analytical review of environmental justice research: What do we really know? *Environmental Management*, 29(1), 3–15. doi:10.1007/s00267-001-0037-8

- Briggs, X. (1997). Moving up versus moving out: Neighborhood effects in housing mobility programs. *Housing Policy Debate*, 8(1), 195–234.
- Brown, J., Florax, R., & McNamara, K. (2009). Determinants of Investment Flows in U.S. Manufacturing. *The Review of Regional Studies*, *39*(3), 269–286.
- Brueckner, J. K., & Rosenthal, S. S. (2009). Gentrification and Neighborhood Housing Cycles: Will America's Future Downtowns Be Rich? *Review of Economics and Statistics*, 91(4), 725–743. doi:10.1162/rest.91.4.725
- Brulle, R. J., & Pellow, D. N. (2006). Environmental justice: human health and environmental inequalities. *Annual Review of Public Health*, 27(102), 103–124. doi:10.1146/annurev.publhealth.27.021405.102124
- Bullard, R. D. (1999). Dismantling Environmental Racism in the USA. *Local Environment*, 4(1), 5–19. doi:10.1080/13549839908725577
- Bullard, R. D., & Wright, B. H. (1993). Environmental justice for all: community perspectives on health and research needs. *Toxicology & Industrial Health.*, 9, 821–841.
- Burby, R. J. (1998). *Cooperating with nature: Confronting natural hazards with land use planning for sustainable communities.* Washington, DC: Joseph Henry Press.
- Burge, G. (2011). Do tenants capture the benefits from the low-income housing tax credit program? *Real Estate Economics*, *39*(1), 71–96.
- Burgess, E. W. (2008). The growth of the city: An introduction to a research project. In *Urban Ecology: An International Perspective on the Interaction Between Humans and Nature* (pp. 71–78). doi:10.1007/978-0-387-73412-5_5
- Burnell, J. D. (1988). Crime and Racial Composition in Contiguous Communities as Negative Externalities: Prejudiced Households' Evaluation of Crime Rate and Segregation Nearby Reduces Housing Values and Tax Revenues. American Journal of Economics & Sociology, 47, 177–193.
- Buzzelli, M. (2007). Bourdieu does environmental justice? Probing the linkages between population health and air pollution epidemiology. *Health and Place*, *13*, 3–13. doi:10.1016/j.healthplace.2005.09.008
- Cannon, T., & Müller-Mahn, D. (2010). Vulnerability, resilience and development discourses in context of climate change. *Natural Hazards*, 55(3), 621–635. doi:10.1007/s11069-010-9499-4
- Carlson, D., Haveman, R., Kaplan, T., & Wolfe, B. (2012). Long-term effects of public lowincome housing vouchers on neighborhood quality and household composition. *Journal of Housing Economics*, 21, 101–120. doi:10.1016/j.jhe.2012.04.004

- Carter, W., Schill, M. H., & Wachter, S. M. (1998). Polarisation, Public Housing and Racial Minorities in US Cities. *Urban Studies*, *35*(10), 1889–1911. doi:10.1080/0042098984204
- Castells, M. (1983). The City and the Grassroots. Berkeley: University of California Press.
- Chakraborty, J., & Armstrong, M. P. (2001). Assessing the Impact of Airborne Toxic Releases on Populations with Special Needs. *The Professional Geographer*, 53, 119–131. doi:10.1111/0033-0124.00274
- Chakraborty, J., Collins, T. W., Grineski, S. E., Montgomery, M. C., & Hernandez, M. (2014). Comparing Disproportionate Exposure to Acute and Chronic Pollution Risks: A Case Study in Houston, Texas. *Risk Analysis*. doi:10.1111/risa.12224
- Chakraborty, J., & Maantay, J. A. (2011). Proximity analysis for exposure assessment in environmental health justice research. In J. A. Maantay & S. McLafferty (Eds.), *Geospatial* analysis of environmental health (pp. 111–138). Springer Netherlands. doi:10.1007/978-94-007-0329-2
- Cliff, A. D., & Ord, J. K. (1981). *Spatial Processes: Models & Applications* (Vol. 44). London, UK: Pine Forge Press.
- Collins, T. W. (2009). The production of unequal risk in hazardscapes: An explanatory frame applied to disaster at the US-Mexico border. *Geoforum*, 40(4), 589–601. doi:10.1016/j.geoforum.2009.04.009
- Collins, T. W. (2010). Marginalization, facilitation, and the production of unequal risk: The 2006 Paso del Norte floods. *Antipode*, 42(2), 258–288. doi:10.1111/j.1467-8330.2009.00755.x
- Collins, T. W., & Jimenez, A. M. (2012). The Neoliberal Production of Vulnerability and Unequal Risk. In S. Dooling & G. Simon (Eds.), *Cities, Nature and Development: The Politics and Production of Urban Vulnerabilities* (pp. 49–67). Burlington, VT: Ashgate Publishing, Ltd.
- Conley, J. F. (2011). Estimation of exposure to toxic releases using spatial interaction modeling. *International Journal of Health Geographics*, *10*(1), 20. doi:10.1186/1476-072X-10-20
- Crowder, K., & Downey, L. (2010). Interneighborhood migration, race, and environmental hazards: modeling microlevel processes of environmental inequality. AJS; American Journal of Sociology, 115, 1110–1149. doi:10.1086/649576
- Crowder, K., & South, S. J. (2008). Spatial Dynamics of White Flight: the Effects of Local and Extralocal Racial Conditions on Neighborhood Out-Migration. *American Sociological Review*, 73(5), 792–812. doi:10.1177/000312240807300505
- Crump, J. (2002). Deconcentration by demolition: Public housing, poverty, and urban policy. *Environment and Planning D: Society and Space*, 20(5), 581–596. doi:10.1068/d306
- Cutter, S. L. (1996). Vulnerability to environmental hazards. *Progress in Human Geography*, 20, 529–539. doi:10.1177/030913259602000407
- Cutter, S. L. (2003). American hazardscapes. Washington, DC: Joseph Henry Press.
- Cutter, S. L., Boruff, B. J., & Shirley, W. L. (2003). Social Vulnerability to Environmental Hazards. *Social Science Quarterly*, 84(2), 242–261.
- Cutter, S. L., & Finch, C. (2008). Temporal and spatial changes in social vulnerability to natural hazards. *Proceedings of the National Academy of Sciences of the United States of America*, 105(7), 2301–2306. Retrieved from http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=2268131&tool=pmcentrez&ren dertype=abstract
- Cutter, S. L., Hodgson, M. E., & Dow, K. (2001). Subsidized Inequities: the Spatial Patterning of Environmental Risks and Federally Assisted Housing. *Urban Geography*, 22(1), 29–53. doi:10.2747/0272-3638.22.1.29
- Cutter, S. L., Johnson, L. A., Finch, C., & Berry, M. (2007). The U.S. Hurricane Coasts: Increasingly Vulnerable? *Environment: Science and Policy for Sustainable Development*, 49(7), 8–21. doi:10.3200/ENVT.49.7.8-21
- Davis, M. (1998). *Ecology of fear: Los Angeles and the imagination of disaster*. New York: Metropolitan Books.
- Dawkins, C. J. (2004). Recent evidence on the continuing causes of black-white residential segregation. *Journal of Urban Affairs*, 26, 379–400. doi:10.1111/j.0735-2166.2004.00205.x
- Deng, L. (2011). Low-Income Housing Tax Credit Developments and Neighborhood Change: A Case Study of Miami-Dade County. *Housing Studies*, 26(6), 867–895. doi:10.1080/02673037.2011.593130
- Denton, N. A., & Massey, D. S. (1991). Patterns of neighborhood transition in a multiethnic world: U.S. metropolitan areas, 1970-1980. *Demography*, 28, 41–63. doi:10.2307/2061335
- Deyle, R. E., Chapin, T. S., & Baker, E. J. (2008). The Proof of the Planning Is in the Platting: An Evaluation of Florida's Hurricane Exposure Mitigation Planning Mandate. *Journal of the American Planning Association*, 74(3), 349–370. doi:10.1080/01944360802229612
- Dickens, W. (1999). Rebuilding Urban Labor Markets: What Community Development Can Accomplish. In R. Ferguson & W. Dickens (Eds.), Urban Problems and Community Development (pp. 381–435). Washington, D.C.: Brookings Institution Press.
- Donner, W. R. (2003). *The Human Ecological Model and Its Application to Severe Weather Phenomena: Predicting Tornado Morbidity*. University of Delaware, Newark, DE.

- Dooling, S. (2012). Sustainability Planning, Ecological Gentrification and the Production. In S. Dooling & G. Simon (Eds.), *Cities, Nature and Development: The Politics and Production of Urban Vulnerabilities*. Ashgate Publishing, Ltd.
- Dooling, S., & Simon, G. (2012). Cities, Nature and Development: The Politics and Production of Urban Vulnerability. In S. Dooling & G. Simon (Eds.), *Cities, Nature and Development: The Politics and Production of Urban Vulnerabilities* (pp. 3–21). Ashgate Publishing, Ltd.
- Downey, L. (2005). The Unintended Significance of Race: Environmental Racial Inequality in Detroit. *Social Forces*, 83(3), 971–1007. doi:10.1353/sof.2005.0026
- Downey, L. (2006). Using geographic information systems to reconceptualize spatial relationships and ecological context. *American Journal of Sociology*, *112*, 567–612. Retrieved from http://www.scopus.com/scopus/inward/record.url?eid=2-s2.0-33845472683&partnerID=40&rel=R6.5.0
- Downs, A. (1981). *Neighborhoods and Urban Development*. Washington, DC: The Brookings Institution.
- Eakin, H., & Luers, A. L. (2006). Assessing the Vulnerability of Social-Environmental Systems. *Annual Review of Environment and Resources*, *31*(1), 365–394. doi:10.1146/annurev.energy.30.050504.144352
- Elhorst, J. P. (2010). Applied Spatial Econometrics: Raising the Bar. *Spatial Economic Analysis*, 5(1), 9–28. doi:10.1080/17421770903541772
- Enarson, E., Fothergill, A., & Peek, L. (2007). Gender and Disaster : Foundations and Directions. In E. L. Quarantelli, R. Dynes, & H. Rodriguez (Eds.), *Handbook of Disaster Research* (pp. 130–146). Springer New York. doi:10.1007/978-0-387-32353-4_8
- Enarson, E., & Morrow, B. H. (1998). *The gendered terrain of disaster: through women's eyes*. Miami, FL: Florida International University. Retrieved from http://www.ops.org.bo/textocompleto/ide18245.pdf
- EPA. (2008). *Toxic Release Inventory national analysis brochure 2008*. Retrieved from http://www.epa.gov/tri/tridata/tri08/national_analysis/brochure/brochure.htm
- Eriksen, S., & Kelly, P. (2007). Developing credible vulnerability indicators for climate adaptation policy assessment. *Mitigation and Adaptation Strategies for Global ..., 12*(4), 495–524. doi:10.1007/s11027-006-3460-6
- Fainstein, S. S., & Fainstein, N. I. (1985). Economic Restructuring and the Rise of Urban Social Movements. Urban Affairs Quarterly, 21, 187–206. doi:10.1177/004208168502100204

- Feins, J. D., & Patterson, R. (2005). Geographic Mobility in the Housing Choice Voucher Program : A Study of Families Entering the Program , 1995 – 2002. *Cityscape*, 8(2), 21–47. doi:10.2307/1166662
- Finch, C., Emrich, C. T., & Cutter, S. L. (2010). Disaster disparities and differential recovery in New Orleans. *Population and Environment*, 31(4), 179–202. doi:10.1007/s11111-009-0099-8
- Findlay, A. M. (2005). Editorial: Vulnerable spatialities. *Population, Space and Place*, 11, 429–439. doi:10.1002/psp.399
- Firey, W. (1945). Sentiment and Symbolism as Ecological Variables. *American Sociological Review*, *10*(2), 140–148. doi:10.2307/2085629
- Fothergill, A. (2004). *Heads Above Water: Gender, Class and Family in the Grand Forks Flood.* Albany, NY: State University of New York Press,.
- Fothergill, A., Maestas, E. G., & Darlington, J. D. (1999). Race, ethnicity and disasters in the United States: a review of the literature. *Disasters*, 23(2), 156–173. doi:10.1111/1467-7717.00111
- Fothergill, A., & Peek, L. (2004). Poverty and disasters in the United States: A review of recent sociological findings. *Natural Hazards*, *32*(1), 89–110.
- Freeman, L. (2003). The impact of assisted housing developments on concentrated poverty. *Housing Policy Debate*, *14*(1-2), 103–141. doi:10.1080/10511482.2003.9521470
- Freeman, L. (2004). Siting affordable housing: Location and neighborhood trends of low income housing tax credit developments in the 1990s (pp. 1–20). Washington, DC.
- Füssel, H. M., & Klein, R. J. T. (2006). Climate change vulnerability assessments: An evolution of conceptual thinking. *Climatic Change*, 75, 301–329. doi:10.1007/s10584-006-0329-3
- Gabriel, K. R. (1971). The biplot graphic display of matrices with application to principal component analysis. *Biometrika*, *58*, 453–467. doi:10.1093/biomet/58.3.453
- Galster, G. C. (2013). U.S. Assisted Housing Programs and Poverty Deconcentration: A Critical Geographic Review. In D. Manley, M. van Ham, N. Bailey, L. Simpson, & D. Maclennan (Eds.), *Neighbourhood Effects or Neighbourhood Based Problems?* (pp. 215–249). New York, NY: Springer. doi:10.1007/978-94-007-6695-2
- Galster, G. C., Cutsinger, J. M., & Malega, R. (2008). The costs of Concentrated Poverty: Neighborhood Property Markets and the Dynamics of Decline. In N. P. Retsinas & E. S. Belsky (Eds.), *Revisiting Rental Housing: Policies, Programs, and Priorities* (p. 93143). Washington, DC: Joint Center for Housing Studies, Harvard University; Brookings Institution Press.

- Galster, G. C., & Godfrey, E. (2005). By Words and Deeds: Racial Steering by Real Estate Agents in the U.S. in 2000. *Journal of the American Planning Association.*, 71(3), 251–268. doi:10.1080/01944360508976697
- Galster, G. C., & Keeney, W. M. (1988). Race, Residence, Discrimination, and Economic Opportunity: Modeling the Nexus of Urban Racial Phenomena. Urban Affairs Review, 24, 87–117. doi:10.1177/004208168802400106
- Galster, G. C., Mincy, R., & Tobin, M. (1997). The Disparate Racial Neighborhood Impacts of Metropolitan Economic Restructuring. Urban Affairs Review, 32(6), 797–824. doi:10.1177/107808749703200603
- Galster, G. C., Tatian, P., & Smith, R. (1999). The impact of neighbors who use Section 8 certificates on property values. *Housing Policy Debate*, *10*(4), 879–917. doi:10.1080/10511482.1999.9521354
- Gans, H. (1962). *The Urban Villagers: Group and Class Life of Italian Americans*. New York, NY: The Free Press.
- Gardner, S. (2001). Extensions of biplot methodology to discriminant analysis with applications of non-parametric principal components. University of Stellenbosch, Stellenbosch, South Africa.
- Gardner, S., Le Roux, N. J., & Aldrich, C. (2005). Process data visualisation with biplots. *Minerals Engineering*, 18(9), 955–968. doi:10.1016/j.mineng.2004.12.010
- Glaeser, E. L., Kahn, M. E., & Rappaport, J. (2008). Why do the poor live in cities? The role of public transportation. *Journal of Urban Economics*, 63, 1–24. doi:10.1016/j.jue.2006.12.004
- Godschalk, D. R., Beatly, T., Berke, P., Brower, D. J., & Kaiser, E. J. (1999). *Natural hazard mitigation: Recasting disaster policy and planning*. Washington, DC: Island Press.
- Goering, J., Feins, J. D., & Richardson, T. M. (2002). A Cross-Site Analysis of Initial Moving to Opportunity Demonstration Results. *Journal of Housing Research*, 13(1), 1–30.
- Goetz, E. G. (2005). Comment: Public Housing Demolition and the Benefits to Low-Income Families. *Journal of the American Planning Association*, 71(4), 407–410.
- Goodchild, M. F. (1986). Spatial Autocorrelation (Vol. 47). Norwich: Geo Books.
- Gordon, T. (2004). Moving up by moving out? planned developments and residential segregation in California. *Urban Studies*, *41*(2), 441–461. doi:10.1080/0042098032000165334

Gower, J., & Hand, D. (1996). Biplots. London, UK: Chapman & Hall.

- Green, R. K., Malpezzi, S., & Seah, K. (2002). Low income housing tax credit housing developments and property values.
- Hackworth, J., & Smith, N. (2001). The changing state of gentrification. *Tijdschrift Voor Economische En Sociale Geografie Tijdschrift Voor Economische En Sociale Geografie*, 92, 464–477. doi:10.1111/1467-9663.00172
- Hamilton, J. T. (1995). Testing for Environmental Racism: Prejudice, Profits, Political Power? *Journal of Policy Analysis and Management*, 14(1), 107–132. doi:10.2307/3325435
- Handmer, J. W., Dovers, S., & Downing, T. E. (1999). Societal Vulnerability to Climate Change and Variability. *Mitigation and Adaptation Strategies for Global Change*, 4, 267–281. doi:10.1023/a:1009611621048
- Hardoy, J. E., Mitlin, D., & Satterthwaite, D. (2001). Environmental problems in an urbanizing world: Finding solutions for cities in Africa, Asia and Latin America. London: Earthscan Publications.
- Harvey, D. (1973). Social justice and the city. Oxford, UK: Blackwell Publishing.
- Harvey, D. (2005). A Brief History of Neoliberalism. New York: Oxford University Press.
- Hewitt, K. (1983). *Interpretations of calamity From the viewpoint of human ecology*. (K. Hewitt, Ed.). Boston: Allen and Unwin.
- Hewitt, K. (1997). *Regions of risk : a geographical introduction to disasters. Themes in resource management.* (p. xx, 389 p.).
- Heynen, N. C., Kaika, M., & Swyngedouw, E. (2006). In the nature of cities: urban political ecology and the politics of urban metabolism. (N. C. Heynen, M. Kaika, & E. Swyngedouw, Eds.). New York, NY: Routledge.
- Highfield, W. E., Peacock, W. G., & Van Zandt, S. (2014). Mitigation Planning: Why Hazard Exposure, Structural Vulnerability, and Social Vulnerability Matter. *Journal of Planning Education and Research*, 34, 287–300. doi:10.1177/0739456X14531828
- Hogan, D. J. (1996). Consequences, Scattered-site public housing: Characteristics and consequences. Washington, D.C: US Department of Housing and Urban Development.
- Hogan, D. J., & Marandola, E. (2005). Towards an interdisciplinary conceptualisation of vulnerability. *Population, Space and Place*, 11(6), 455–471. doi:10.1002/psp.401
- Hollar, M., & Usowki, K. (2007). Low-income housing tax credit qualified census tracts. *Cityscape*, *9*(3), 153–160.

- Holloway, S. R., Bryan, D., Chabot, R., Rogers, D. M., & Rulli, J. (1998). Exploring the Effect of Public Housing on the Concentration of Poverty in Columbus, Ohio. *Urban Affairs Review*, 33(6), 767–789. doi:10.1177/107808749803300603
- Hoover, E. M., & Vernon, R. (1959). Anatomy of a Metropolis. Cambridge, MA: Harvard U Press, 1959. xvi+345 pp. Cambridge, MA: Harvard Univ. Press.
- Horn, K. M., Ellen, I. G., & Schwartz, A. E. (2014). Do Housing Choice Voucher holders live near good schools? *Journal of Housing Economics*, 23, 28–40.
- Houston, D., Basolo, V., & Yang, D. (2013). Walkability, transit access, and traffic exposure for low-income residents with subsidized housing. *American Journal of Public Health*, 103(4), 673–8. doi:10.2105/AJPH.2012.300734
- Hoyt, H. (1939). *The Structure and Growth of Residential Neighborhoods in American Cities*. Washington, DC: Federal Housing Administration.
- HUD. (1993). Worst Case Needs for Housing Assistance in the United States in 1990 and 1991. Washington, DC.
- HUD. (2007). Affordable Housing Needs 2005: Report to Congress. Washington, DC.
- HUD. (2014). Low-Income Housing Tax Credits: About the LIHTC database, users note. Retrieved February 08, 2015, from http://www.huduser.org/portal/datasets/lihtc.html
- Hynes, H., & Lopez, R. (2012). Cumulative risk and a call for action in environmental justice communities. *Journal of Health Disparities*, 1(2), 29–57.
- Ionescu, C., Klein, R. J. T., Hinkel, J., Kumar, K. S. k., & Klein, R. (2009). Towards a Formal Framework of Vulnerability to Climate Change. *Environmental Modeling & Assessment*, 14(1), 1–16. doi:10.1007/s10666-008-9179-x
- Kamel, N. (2012). The actualization of neoliberal space and the loss of housing affordability in Santa Monica, California. *Geoforum*, 43(3), 453–463. doi:10.1016/j.geoforum.2011.10.005
- Kasperson, R. E., Dow, K., Archer, E., Caceres, D., Downing, T., Elmqvist, T., ... Ziervogel, G. (2005). Vulnerable Peoples and Places. In R. Hassan, R. Scholes, & N. Ash (Eds.), *Ecosystems and Human Wellbeing: Current State and Trends* (p. Vol. 1 143–164). Washington, DC: Island Press.
- Kaswan, A. (1997). Environmental justice: bridging the gap between environmental laws and "justice." *The American University Law Review*, 47, 222–300.
- Khadduri, J., Climaco, C., & Burnett, K. (2012). What Happens to Low-Income Housing Tax Credit Properties at Year 15 and Beyond?.

- Krysan, M. (2008). Does race matter in the search for housing? An exploratory study of search strategies, experiences, and locations. *Social Science Research*, 37, 581–603. doi:10.1016/j.ssresearch.2007.06.001
- Kucheva, Y. A. (2013). Subsidized Housing and the Concentration of Poverty, 1977-2008: A Comparison of Eight U.S. Metropolitan Areas. *City & Community*, 12(2), 113–133. doi:10.1111/cico.12014
- La Grange, A., Le Roux, N., & Gardner-Lubbe, S. (2009). BiplotGUI : Interactive Biplots in R. *Journal Of Statistical Software*, 30(12), 1–37. doi:10.1002/wics.10
- LaCour-Little, M., & Malpezzi, S. (2001). *Gated communities and property values. Research report.* Madison, WI.
- Lamare, J. (1998). Texas Politics: Economics, Power and Policy. New York: West Pub. Co.
- Landis, J. D., & McClure, K. (2010). Rethinking federal housing policy. *Journal of the American Planning Association*, 76(3), 319–348. doi:10.1080/01944363.2010.484793
- Lang, B. J. (2012). Location incentives in the low-income housing tax credit: Are qualified census tracts necessary? *Journal of Housing Economics*, 21(2), 142–150. doi:10.1016/j.jhe.2012.04.002
- Laska, S., & Morrow, B. H. (2006). Social Vulnerabilities and Hurricane Katrina: An Unnatural Disaster in New Orleans. *Marine Technology Society Journal*, 40(4), 16–26. doi:10.4031/002533206787353123
- Le Goix, R. (2005). Gated Communities: Sprawl and Social Segregation in Southern California. *Housing Studies*, 20(2), 323–343. doi:10.1080/026730303042000331808
- Le Goix, R., & Vesselinov, E. (2013). Gated communities and house prices: Suburban change in southern california, 1980-2008. *International Journal of Urban and Regional Research*, 37(6), 2129–2151. doi:10.1111/j.1468-2427.2012.01139.x
- Le Roux, N. J., & Gardner, S. (2005). Analysing your Multivariate data as a pictorial: A case for applying biplot methodology? *International Statistical Review*, 73(3), 365–387. doi:10.1111/j.1751-5823.2005.tb00154.x
- Lees, L. (2003). Super-gentrification: The case of Brooklyn Heights, New York City. *Urban Studies*, 40(12), 2487–2509. doi:10.1080/0042098032000136174
- Lens, M. C. (2013). Safe, but Could Be Safer: Why Do HCVP Households Live in Higher Crime Neighborhoods? *Cityscape: A Journal of Policy Development and Research*, 15(3), 131– 152.

- Lens, M. C., Ellen, G., & O'Regan, K. (2011). Do Vouchers Help Low-Income Households Live in Safer Neighborhoods ? Evidence on the Housing Choice Voucher Program. *Cityscape*, 13(3), 135–159.
- LeSage, J., & Pace, R. (2009). *Introduction to spatial econometrics*. Boca Raton, FL: Taylor & Francis.
- Li, Y., & Morrow-Jones, H. A. (2010). The Impact of Residential Mortgage Foreclosure on Neighborhood Change and Succession. *Journal of Planning Education and Research*, 30(1), 22–39. doi:10.1177/0739456X10375305
- Lin, J. (1995). Ethnic places, postmodernism, and urban change in Houston. *Sociological Quarterly*, *36*, 629–647. doi:10.1111/j.1533-8525.1995.tb00457.x
- Linder, S. H., Marko, D., & Ken, S. (2008). Cumulative cancer risk from air pollution in houston: Disparities in risk burden and social disadvantage. *Environmental Science and Technology*, 42, 4312–4322. doi:10.1021/es072042u
- Logan, J. R., & Molotch, H. (1987). *Urban fortunes: The political economy of place*. Berkeley: University of California Press.
- Logan, J. R., Whaley, R. B., & Crowder, K. (1997). The Character and Consequences of Growth Regimes: An Assessment of 20 Years of Research. Urban Affairs Review, 32(5), 603–630. doi:10.1177/107808749703200501
- Logan, J. R., Xu, Z., & Stults, B. J. (2014). Interpolating U.S. Decennial Census Tract Data from as Early as 1970 to 2010: A Longitudinal Tract Database. *The Professional Geographer*, 66(3), 1–9. doi:10.1080/00330124.2014.905156
- Logan, J. R., & Zhang, C. (2010). Global Neighborhoods: New Pathways to Diversity and Separation1. American Journal of Sociology. doi:10.1086/649498
- Long, A. P. (2007). Poverty is the new prostitution: race, poverty, and public housing in post-Katrina New Orleans. *The Journal of American History*, 94(3), 795–803. doi:doi: 10.2307/25095141
- Maantay, J. A. (2001). Zoning, equity, and public health. *American Journal of Public Health*, *91*(7), 1033–1041. doi:10.2105/AJPH.91.7.1033
- Maantay, J. A., & Maroko, A. (2009a). Mapping urban risk: Flood hazards, race, & environmental justice in New York. *Applied Geography*, 29, 111–124. doi:10.1016/j.apgeog.2008.08.002
- Maantay, J. A., & Maroko, A. (2009b). Mapping Urban Risk: Flood Hazards, Race, & Environmental Justice In New York". *Applied Geography (Sevenoaks, England)*, 29(1), 111–124. doi:10.1016/j.apgeog.2008.08.002

- Maantay, J. A., Maroko, A. R., & Herrmann, C. (2007). Mapping Population Distribution in the Urban Environment: The Cadastral-based Expert Dasymetric System (CEDS). *Cartography* and Geographic Information Science, 34(2), 77–102. doi:10.1559/152304007781002190
- Macintosh, A. (2013). Coastal climate hazards and urban planning: How planning responses can lead to maladaptation. *Mitigation and Adaptation Strategies for Global Change*, *18*, 1035–1055. doi:10.1007/s11027-012-9406-2
- Malpass, P. (2003). Housing policy and the disabling of local authorities. In J. Birchall (Ed.), *Housing Policy in the 1990s* (pp. 10–28). New York, NY: Routledge.
- Manly, B. F. J. (1986). *Multivariate statistical methods: a primer. Multivariate Statistical Methods A Primer* (Vol. 6, p. 159). Retrieved from http://books.google.be/books?id=bLSEeMjvD9oC
- Massey, D. S., & Denton, N. (1993). *American apartheid: Segregation and the making of the underclass*. Cambridge, MA: Harvard University Press.
- Massey, D. S., & Kanaiaupuni, S. M. (1993). Public Housing and Concentration of Poverty. *Social Science Quarterly*, 74(1), 109–122.
- Massey, D. S., & Mullan, B. P. (1984). Processes of Hispanic and Black Spatial Assimilation. *American Journal of Sociology*, 89, 836–873.
- McClure, K. (2006). The low-income housing tax credit program goes mainstream and moves to the suburbs. *Housing Policy Debate*, *17*(3), 419–446. doi:10.1080/10511482.2006.9521576
- McClure, K., & Johnson, B. (2014). Housing Programs Fail to Deliver on Neighborhood Quality, Reexamined. *Housing Policy Debate*. doi:10.1080/10511482.2014.944201
- McLaughlin, P., & Dietz, T. (2008). Structure, agency and environment: Toward an integrated perspective on vulnerability. *Global Environmental Change*, 18, 99–111. doi:10.1016/j.gloenvcha.2007.05.003
- Mileti, D. S., & Darlington, J. D. (1997). The Role of Searching in Shaping Reactions to Earthquake Risk Information. *Social Problems*. doi:10.1525/sp.1997.44.1.03x0214f
- Miller, L. M., Antonio, R. J., & Bonanno, A. (2011). Hazards of neoliberalism: delayed electric power restoration after Hurricane Ike. *The British Journal of Sociology*, 62(3), 504–22. doi:10.1111/j.1468-4446.2011.01376.x
- Mohai, P., & Bryant, B. (1998). Is There a "Race" Effect on Concern for Environmental Quality? *The Public Opinion Quarterly*, 62, 475–505. doi:10.2307/2749675
- Mohai, P., Pellow, D., & Roberts, J. T. (2009). Environmental Justice. *Annual Review of Environment and Resources*, *34*, 405–430. doi:10.1146/annurev-environ-082508-094348

- Mohai, P., & Saha, R. (2006). Reassessing racial and socioeconomic disparities in environmental justice research. *Demography*, 43(2), 383–399. doi:10.1353/dem.2006.0017
- Molotch, H. (1976). The City as Growth Machine: Toward a Political Economy of Place. *American Journal of Sociology*, 82, 309–330.
- Morenoff, J. D., & Sampson, R. J. (1997). Violent Crime and The Spatial Dynamics of Neighborhood Transition: Chicago, 1970-1990. Social Forces, 76, 31–64. doi:10.2307/2580317
- Moser, C. O. N. (1998). The asset vulnerability framework: Reassessing urban poverty reduction strategies. *World Development*, 26(1), 1–19. doi:10.1016/S0305-750X(97)10015-8
- Mustafa, D. (2005). The production of an urban hazardscape in Pakistan: Modernity, vulnerability, and the range of choice. *Annals of the Association of American Geographers*, 95(3), 566–586. doi:10.1111/j.1467-8306.2005.00475.x
- Neter, J., Kutner, M. H., Nachtsheim, C. J., & Wasserman, W. (1999). *Applied Linear Statistical Models. Irwin series in statistics* (Vol. 1). Chicago, IL: Irwin. Retrieved from http://www.mendeley.com/research/applied-linear-statistical-models/
- Newman, K., & Ashton, P. (2004). Neoliberal urban policy and new paths of neighborhood change in the American inner city. *Environment and Planning A*, *36*(7), 1151–1172. doi:10.1068/a36229
- Nicholls, R. J., Hanson, S., Herweijer, C., Patmore, N., Hallegatte, S., Corfee-Morlot, J., ... Wood, R. M. (2008). *Ranking Port Cities with High Exposure and Vulnerability to Climate Extremes: Exposure Estimates* (No. 1).
- O'Connor, B. P. (2000). SPSS and SAS programs for determining the number of components using parallel analysis and velicer's MAP test. *Behavior Research Methods, Instruments, & Computers, 32*(3), 396–402. doi:10.3758/BF03200807
- O'Regan, K. M., & Horn, K. M. (2013). What can we learn about the Low-Income Housing Tax Credit program by looking at the tenants? *Housing Policy Debate*, 23, 597–613.
- Oakes, J. M., Anderton, D. L., & Anderson, A. B. (1996). A Longitudinal Analysis of Communities with Hazardous Waste Facilities. *Social Science Research*, 25(2), 125 – 148.
- Oakley, D. (2008). Locational Patterns of Low-Income Housing Tax Credit Developments: A Sociospatial Analysis of Four Metropolitan Areas. *Urban Affairs Review*, *43*(5), 599–628. doi:10.1177/1078087407309432
- Ord, K. (1975). Estimation methods for models of spatial interaction. *Journal of the American Statistical Association*, 70(349), 120–126. doi:10.1080/01621459.1975.10480272

- Orlove, B. (2009). The past, the present and some possible futures of adaptation. In W. N. Adger, I. Lorenzoni, & K. O'Brien (Eds.), *Adapting to Climate Change Thresholds Values Governance* (pp. 131–163). Cambridge, MA: Cambridge University Press.
- Orsi, J. (2004). *Hazardous metropolis: flooding and urban ecology in Los Angeles*. Berkeley: University of California Press.
- Osiecki, K. M., Kim, S., Chukwudozie, I. B., & Calhoun, E. A. (2013). Utilizing Exploratory Spatial Data Analysis to Examine Health and Environmental Disparities in Disadvantaged Neighborhoods. *Environmental Justice*, *6*(3), 81–87. doi:10.1089/env.2013.0010
- Parry, M. L., Canziani, O. F., Palutikof, J. P., van der Linden, P. J., & Hanson, C. E. (2007). Climate Change 2007: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change (Vol. 37, p. 2407). Cambridge, MA: Cambridge University Press. doi:10.2134/jeq2008.0015br
- Pastor, M., Sadd, J. I. M., & Hipp, J. (2001). Which Came First? Toxic Facilities, Minority Move-in, and Environmental Justice. *Journal of Urban Affairs*, 23(1), 1–21. doi:10.1111/0735-2166.00072
- Patil, V. H., Singh, S. N., Mishra, S., & Todd Donavan, D. (2008). Efficient theory development and factor retention criteria: Abandon the "eigenvalue greater than one" criterion. *Journal of Business Research*, *61*(2), 162–170. doi:10.1016/j.jbusres.2007.05.008
- Peacock, W. G. (2003). Hurricane Mitigation Status and Factors Influencing Mitigation Status among Florida's Single-Family Homeowners. *Natural Hazards Review*. doi:10.1061/(ASCE)1527-6988(2003)4:3(149)
- Peacock, W. G., Dash, N., & Zhang, Y. (2007). Sheltering and Housing Recovery Following Disaster. In *Handbook of Disaster Research* (pp. 258–274). doi:10.1007/978-0-387-32353-4_15
- Peacock, W. G., Morrow, B., & Gladwin, H. (1997). *Hurricane Andrew: Ethnicity, Gender and the Sociology of Disasters*. New York, NY: Routledge.
- Pearce, D. M. (1979). Gatekeepers and Homeseekers: Institutional Patterns in Racial Steering. *Social Problems*, 26(3), 325–342. doi:10.1525/sp.1979.26.3.03a00080
- Peet, R., & Watts, M. (1996). Liberation Ecologies Environment, Development, Social Movements. In ... (Eds) Liberation Ecologies: Environment, development ... (pp. 1–45). New York, NY: Routledge.
- Pelling, M. (2003). Toward a political ecology of urban environmental risk. In K. S. Zimmerer & T. J. Bassett (Eds.), *Political Ecology: an Integrative Approach to Geography and Environment-Development Studies* (pp. 73–93). New York, NY: Guilford.

- Pendall, R. (2000). Why voucher and certificate users live in distressed neighborhoods. *Housing Policy Debate*, 11(4), 881–910. doi:10.1080/10511482.2000.9521391
- Perkins, H. A. (2012). The Production of Urban Vulnerability through Market-based Parks Governance. In S. Dooling & G. Simon (Eds.), *Cities, Nature and Development: The Politics and Production of Urban Vulnerabilities* (pp. 69–82). Ashgate Publishing, Ltd.
- Perry, R. W., & Mushkatel, E. (1986). *Minority Citizens in Disaster*. Athens GA: University of Georgia Press.
- Perry, R. W., & Nelson, L. (1991). Ethnicity and hazard information dissemination. *Environmental Management*, 15(4), 581–587. doi:10.1007/BF02394747
- Peterman, W. (2000). Neighborhood planning and community-based development : the potential and limits of grassroots action. Cities & planning series (p. xiii, 191 p.). doi:10.4135/9781452232508
- Phillips, B., Thomas, D. S., Fothergill, A., & Blinn-Pike, L. (2009). *Social vulnerability to disasters*. London: Taylor & Francis.
- Podagrosi, A., Vojnovic, I., & Pigozzi, B. (2011). The diversity of gentrification in Houston's urban renaissance: from cleansing the urban poor to supergentrification. *Environment and Planning A*, 43(8), 1910–1929. doi:10.1068/a43526
- Popkin, S. J., Katz, B., Cunningham, M. K., Brown, K. D., Gustafon, J., & Turner, M. A. (2004). *A Decade of HOPE VI : Research Findings and Policy Challenges. Urban Institute.*
- Pulido, L. (2000). Rethinking Environmental Racism: White Privilege and Urban Development in Southern California. *Annals of the Association of American Geographers*, 90, 12–40. doi:10.1111/0004-5608.00182
- Qian, Z. (2010). Without zoning: Urban development and land use controls in Houston. *Cities*, 27, 31–41. doi:10.1016/j.cities.2009.11.006
- Rajotte, B. (2007). Environmental Justice in New Orleans: A New Lease on Life for Title VIII? *Tulane Environmental Law Journal*, 21(1), 51–81.
- Reardon, S. F., & Bischoff, K. (2011). Income inequality and income segregation. American Journal of Sociology, 116(4), 1092–1153. doi:10.1086/657114
- Ribot, J. (2009). Vulnerability does not fall from the sky: Towards multi-scale, pro-poor climate policy. In R. Mearns & A. Norton (Eds.), *Social dimensions of climate change: equity and vulnerability in a warming world* (pp. 47–74). Washington, DC: The World Bank. doi:10.1088/1755-1307/6/34/342040

- Rodriguez, H., & Russell, C. N. (2006). Understanding Disasters: Vulnerability, Sustainable Development, and Resiliency. In J. Blau & K. Lyall-Smith (Eds.), *Public Sociologies Reader* (pp. 93–211). Rowman & Littlefield.
- Rosenthal, S. S. (2008). Old homes, externalities, and poor neighborhoods. A model of urban decline and renewal. *Journal of Urban Economics*, 63(3), 816–840. doi:10.1016/j.jue.2007.06.003
- Ross, S. L., & Yinger, J. (2002). *The Color of Credit: Mortgage Discrimination, Research Methodology, and Fair-Lending Enforcement*. Cambridge: The MIT Press.
- Russell, L. A., Goltz, J. D., & Bourque, L. B. (1995). Preparedness and Hazard Mitigation Actions before and after Two Earthquakes. *Environment and Behavior*. doi:10.1177/0013916595276002
- Sampson, R. J., Morenoff, J. D., & Earls, F. (1999). Beyond Social Capital: Spatial Dynamics of Collective Efficacy for Children. *American Sociological Review*, 64(5), 633. doi:10.2307/2657367
- Sanderson, D. (2000). Cities, Disasters and Livelihoods. *Environment and Urbanization*, 12(2), 93–102. doi:10.1057/palgrave.rm.8240068
- Sarewitz, D., Pielke, R., & Keykhah, M. (2003). Vulnerability and risk: Some thoughts from a political and policy perspective. *Risk Analysis*, 23(4), 805–810. doi:10.1111/1539-6924.00357
- Sassen, S. (2000). *Cities in a World Economy* (Second Edi). Thousand Oaks, CA: Pine Forge Press.
- Scally, C., & Koenig, R. (2012). Beyond NIMBY and poverty deconcentration: reframing the outcomes of affordable rental housing development. *Housing Policy Debate*, 22(3), 435– 461.
- Schielzeth, H. (2010). Simple means to improve the interpretability of regression coefficients. *Methods in Ecology and Evolution*, *1*, 103–113. doi:10.1111/j.2041-210X.2010.00012.x
- Schill, M. H., & Wachter, S. M. (1995). The Spatial Bias of Federal Housing Law and Policy: Concentrated Poverty in Urban America. University of Pennsylvania Law Review, 143, 1284–1349.
- Schiller, A. R. (2010). The impact of a storm surge on business establishments in the Houston MSA. *Natural Hazards*, 56(1), 331–346. doi:10.1007/s11069-010-9572-z
- Schlosberg, D. (2013). Theorising environmental justice: the expanding sphere of a discourse. *Environmental Politics*, 22(1), 37–55. doi:10.1080/09644016.2013.755387

- Schmidtlein, M. C., Deutsch, R. C., Piegorsch, W. W., & Cutter, S. L. (2008). A sensitivity analysis of the social vulnerability index. *Risk Analysis*, 28(4), 1099–1114. doi:10.1111/j.1539-6924.2008.01072.x
- Schwartz, A. F. (2010). Housing policy in the United States (p. 368). New York, NY: Routledge.
- Seicshnaydre, S. E. (2010). How Government Housing Perpetuates Racial Segregation: Lessons from Post-Katrina New Orleans. *Catholic University Law Review*, 60, 660–718.
- Sexton, K., Linder, S. H., Marko, D., Bethel, H., & Lupo, P. J. (2007). Comparative assessment of air pollution-related health risks in Houston. *Environmental Health Perspectives*, 115, 1388–1393. doi:10.1289/ehp.10043
- Shlay, A. B., & Rossi, P. H. (1981). Keeping up the Neighborhood: Estimating Net Effects of Zoning. *American Sociological Review*, 46(6), 703–719. doi:10.2307/2095075
- Shultz, S. D., & Fridgen, P. M. (2001). Floodplains and housing values: Implications for flood mitigation projects. *Journal of the American Water Resources Association*, 37, 595–603.
- Smit, B., & Wandel, J. (2006). Adaptation, adaptive capacity and vulnerability. *Global Environmental Change*, *16*, 282–292. doi:10.1016/j.gloenvcha.2006.03.008
- Smith, S. M., Tremethick, M. J., Johnson, P., & Gorski, J. (2009). Disaster planning and response: considering the needs of the frail elderly. *International Journal of Emergency Management*, 6(1), 1–13. doi:10.1504/IJEM.2009.025170
- Soja, E. (2000). Postmetropolis. Oxford, UK: Blackwell Publishers.
- Soja, E., Morales, R., & Wolff, G. (1983). Urban Restructuring: An Analysis of Social and Spatial Change in Los Angeles. *Economic Geography*, 59(2), 195–230.
- South, S. J., & Crowder, K. D. (1998). Leaving the 'Hood: Residential Mobility between Black, White, and Integrated Neighborhoods. *American Sociological Review*, 63(1), 17–26. doi:10.2307/2657474
- Speyrer, J. F., & Ragas, W. R. (1991). Housing prices and flood risk: An examination using spline regression. *The Journal of Real Estate Finance and Economics*, 4, 395–407. doi:10.1007/BF00219506
- Stearns, L. B., & Logan, J. R. (1986). The Racial Structuring of the Housing Market and Segregation in Suburban Areas. *Social Forces*, 65(1), 28–42. doi:10.2307/2578934
- Susman, P., O'Keefe, P., & Wisner, B. (1983). Global disasters, a radical interpretation. In K. Hewitt (Ed.), *Interpretations of Calamity from the Viewpoint of Human Ecology* (pp. 263–283). Boston: Allen and Unwin.

- Suttles, G. (1972). *The Social Construction of Communities*. Chicago, IL: University of Chicago Press.
- Tate, E. (2012). Social vulnerability indices: a comparative assessment using uncertainty and sensitivity analysis. *Natural Hazards*, *63*(2), 325–347. doi:10.1007/s11069-012-0152-2
- Tate, E., Cutter, S. L., & Berry, M. (2010). Integrated multihazard mapping. *Environment and Planning B: Planning and Design*, *37*(4), 646–663. doi:10.1068/b35157
- Temkin, K., & Rohe, W. (1996). Neighborhood Change and Urban Policy. *Journal of Planning Education and Research*, *15*(3), 159–170. doi:10.1177/0739456X9601500301
- Turner, B. L., Kasperson, R. E., Matson, P. A., McCarthy, J. J., Corell, R. W., Christensen, L., ... Schiller, A. (2003). A framework for vulnerability analysis in sustainability science. *Proceedings of the National Academy of Sciences of the United States of America*, 100, 8074–8079. doi:10.1073/pnas.1231335100
- Turner, M. A. (1998). Moving out of poverty: Expanding mobility and choice through tenant-based housing assistance. *Housing Policy Debate*, 9(2), 373–394. doi:10.1080/10511482.1998.9521299
- Van Zandt, S., & Mhatre, P. C. (2009). Growing Pains: Perpetuating Inequality Through the Production of Low-Income Housing in the Dallas/Fort Worth Metroplex. Urban Geography, 30(5), 490–513. doi:10.2747/0272-3638.30.5.490
- Van Zandt, S., & Mhatre, P. C. (2013). The Effect of Housing Choice Voucher Households on Neighborhood Crime: Longitudinal Evidence From Dallas. *Poverty & Public Policy*, 5(3), 229–249. doi:10.1002/pop4.36
- Van Zandt, S., Peacock, W. G., Henry, D. W., Grover, H., Highfield, W. E., & Brody, S. D. (2012). Mapping social vulnerability to enhance housing and neighborhood resilience. *Housing Policy Debate*, 22(1), 29–55. doi:10.1080/10511482.2011.624528
- Vesselinov, E. (2008). Members only: Gated communities and residential segregation in the metropolitan united states. *Sociological Forum*, *23*(3), 536–555. doi:10.1111/j.1573-7861.2008.00075.x
- Vojnovic, I. (2003). Governance in Houston: Growth theories and urban pressures. *Journal of Urban Affairs*, 25, 589–624. doi:10.1111/j.1467-9906.2003.00004.x
- Voss, P. R., & Chi, G. (2006). Highways and Population Change. Rural Sociology, 71(1), 33-58.
- Walters, I. S., & Le Roux, N. J. (2008). Monitoring gender remuneration inequalities in academia using biplots. *ORiON: The Journal of ORSSA*, 24(1), 49–73.

- Walters, I. S., & Roux, N. le. (2008). Monitoring gender remuneration inequalities in academia using biplots. *ORiON: The Journal of ORSSA*, 24(1), 49–73.
- Williamson, A. R., Smith, M. T., & Strambi-Kramer, M. (2009). Housing Choice Vouchers, the Low-Income Housing Tax Credit, and the Federal Poverty Deconcentration Goal. Urban Affairs Review, 45, 119–132. doi:10.1177/1078087409336529
- Wilson, W. J. (1987). The Truly Disadvantaged: The Inner-City, the Underclass, and Public Policy. Ethics (Vol. 101). Chicago, IL: University of Chicago Press. doi:10.1086/293318
- Winnick, L. (1995). The triumph of housing allowance programs: How a fundamental policy conflict was resolved. *Cityscape*, *1*, 95–121.
- Wisner, B., Blaikie, P., Cannon, T., & Davis, I. (2004). At risk: natural hazards, people's vulnerability, and disasters (2nd ed.). London: Routledge.
- Wisner, B., Westgate, K., & O'Keefe, P. (1976). Poverty and disaster. New Society, 9, 546-548.
- Wood, M., Turnham, J., & Mills, G. (2008). Housing affordability and family well-being: Results from the housing voucher evaluation. *Housing Policy Debate*, 19(2), 367–412. doi:10.1080/10511482.2008.9521639
- Wyly, E., & DeFilippis, J. (2010). Mapping Public Housing: The Case of New York City. *City & Community*, 9(1), 61–86. doi:10.1111/j.1540-6040.2009.01306.x
- Yinger, J. (1995). Closed Doors, Opportunities Lost: The Continuing Costs of Housing Discrimination. Contemporary Sociology (Vol. 26). New York: Russell Sage Foundation. doi:10.2307/2076750
- Zahran, S., Brody, S. D., Peacock, W. G., Vedlitz, A., & Grover, H. (2008). Social vulnerability and the natural and built environment: A model of flood casualties in Texas. *Disasters*, *32*(4), 537–560. doi:10.1111/j.1467-7717.2008.01054.x
- Zhang, Y. (2010). Residential Housing Choice in a Multihazard Environment: Implications for Natural Hazards Mitigation and Community Environmental Justice. *Journal of Planning Education and Research*, 30(2), 117–131. doi:10.1177/0739456X10381386
- Zhang, Y., & Peacock, W. G. (2009). Planning for Housing Recovery? Lessons Learned From Hurricane Andrew. *Journal of the American Planning Association*, 76(1), 5–24. doi:10.1080/01944360903294556
- Zwick, W. R., & Velicer, W. F. (1986). Comparison of five rules for determining the number of components to retain. *Psychological Bulletin*, 99, 432–442. doi:10.1037/0033-2909.99.3.432

APPENDIX A

RESULTS OF PCA ANALYSIS

Table A1: PCA results for Harris County

Major components	Major contributing variables
(% variance explained)	
1980 (total variance explained: 71.8%)	
Race and socio-economic status (28.9)	SHRBLK80, FFH80, TRVLPB80, EDUCUH80,
	UNEMPRT80, SERVOCC80, POVRAT80,
	WELFARE80, AVHHIN80, AVGRN180
Housing and labor force (16.5)	KIDS80, AVHHSZ80, FEMLABR80, RNTUR80
Hispanic and foreign-born (9.9)	SHRHSP80, SHRFOR80
Age and gender (8.9)	SHRASN80, OLD80, FEMLABR80
Employment and mobile homes (7.6)	MANUF80, MOB80
1990 (total variance explained: 74.4%)	
Race and socio-economic status (29.4)	SHRBLK90, FFH90, TRVLPB90, EDUCUH90,
	UNEMPRT90, SERVOCC90, POVRAT90,
	WELFARE90, AVHHIN90, AVGRNT90
Housing and labor force (15.1)	AVHHSZ90, FEMLABR90, RNTUR90,
	LABPOP90, HODENT90
Hispanic and foreign-born (12)	SHRHSP90, SHRFOR90, FEMR90
Age (10)	KIDS90, OLD90, AVGVAL90
Employment and mobile homes (7.9)	SHRASN90, MANUF90, MOB90
2000 (total variance explained: 72.5%)	
Race and socio-economic status (27.5)	SHRBLK00, FFH00, TRVLPB00, EDUCUH00,
	UNEMPRT00, SERVOCC00, POVRAT00,
	WELFARE00, AVHHIN00, AVGVAL00,
	AVGRNT00
Hispanic and foreign-born (17.6)	SHRHSP00, KIDS00, SHRFOR00, AVHHSZ00,
	EDUCUH00
Employment and housing (11.4)	MANUF00, RNTUR00, MOB00, HODENT00
Age and labor force (10)	OLD00, FEMLABR00, LABPOP00
Gender (6)	GROUPQ00, FEMR00
2010 (total variance explained: 65.5%)	
Race and socio-economic status (23.4)	SHRBLK10, FFH10, TRVLPB10, EDUCUH10,
	UNEMPRT10, SERVOCC10, POVRAT10,
	WELFARE10, AVHHIN10, AVGVAL10,
	AVGRNT10
Hispanic and foreign-born (13.9)	SHRHSP10, SHRFOR10, AVHHSZ10, EDUCUH10
Age and labor force (11.4)	OLD10, FEMLABR10, WELFARE10
Employment and housing (9.2)	MANUF10, HODENT10
Gender (7.6)	GROUPQ10, FEMR10

Major components	Major contributing variables
(% variance explained)	
1980 (total variance explained: 71%)	
Race and socio-economic status (40.3)	SHRBLK80, FFH80, TRVLPB80, EDUCUH80,
	UNEMPRT80, SERVOCC80, POVRAT80,
	WELFARE80, AVHHIN80, RNTUR80,
	AVGVAL80, AVGRNT80, LABPOP80, HODENT80
Age and Hispanic population (13.4)	SHRHSP80, KIDS80, OLD80, AVHHSZ80
Foreign-born and Asian population	SHRASN80, SHRFOR80, MOB80
(8.7)	EENDOG EENILADDOG MANUEOO
Employment and Gender (8.6)	FEMR80, FEMLABR80, MANUF80
1990 (total variance explained: 70.5%)	
Race and socio-economic status (39.8)	SHRBLK90, FFH90, TRVLPB90, EDUCUH90,
	UNEMPRT90, SERVOCC90, POVRAT90,
	WELFARE90, AVHHIN90, RNTUR90,
	AVGRNT90, LABPOP90, HODENT90
Hispanic population (11.9)	SHRHSP90, SHRFOR90, AVHHSZ90
Foreign-born, Asian, mobile homes	SHRASN90, SHRFOR90, MOB90
(10.5)	
Employment and Gender (8.3)	FEMLABR90, AVHHIN90, AVGVAL90
2000 (total variance explained: 69.6%)	
Race and socio-economic status (38.7)	SHRBLK00, FFH00, TRVLPB00, EDUCUH00,
	UNEMPRT00, SERVOCC00, POVRAT00,
	WELFARE00, AVHHIN00, RNTUR00,
	AVGVAL00, AVGRNT00, LABPOP00, HODENT00
Hispanic population and employment	SHRHSP00, AVHHSZ00, FEMR00, MANUF00,
(15)	AVGVAL00, HODENT00
Foreign-born and Asian (8.7)	SHRASN00, SHRFOR00
Age and gender (7.2)	OLD00, GROUPQ00, FEMLABR00
2010 (total variance explained: 67.7%)	
Race and socio-economic status (26.5)	SHRBLK10, FFH10, TRVLPB10, EDUCUH10,
	UNEMPRT10, SERVOCC10, WELFARE10,
	AVHHIN10, RNTUR10, AVGVAL10, AVGRNT10,
	LABPOP10
Employment, gender (12.7)	KIDS10, AVHHSZ10, FFH10, LABPOP10,
	HODENT10
Hispanic and foreign-born (10.9)	SHRHSP10, SHRASN10, SHRFOR10
Age and gender (10.1)	OLD10, FEMLABR10, WELFARE10
Housing, employment (7.5)	GROUPQ10, MANUF10,

Table A2: PCA results for Orleans Parish

Major components	Major contributing variables
(% variance explained)	
1980 (total variance explained: 75.5%)	
Race and socio-economic status (31.1)	SHRBLK80, FFH80, TRVLPB80, EDUCUH80,
	UNEMPRT80, SERVOCC80, POVRAT80,
	WELFARE80, AVHHIN80, RNTUR80,
	AVGRNT80, HODENT80
Age, labor force (11.5)	OLD80, FEMLABR80, LABPOP80
Household size, age (9.1)	KIDS80, OLD80, AVHHSZ80
Hispanic, foreign-born (8.6)	SHRHSP80, SHRFOR80
Employment and housing (8.5)	GROUPQ80, EDUCUH80, MANUF80
Gender (6.7)	SHRIND80, FEMR80, MOB80
1990 (total variance explained: 73.6%)	
Race and socio-economic status (32.6)	SHRBLK90, FFH90, TRVLPB90, EDUCUH90,
	UNEMPRT90, SERVOCC90, POVRAT90,
	WELFARE90, AVHHIN90, RNTUR90,
	AVGRNT90, HODENT90
Age, labor force (13.9)	SHRASN90, OLD90, FEMLABR90, LABPOP90
Household size, employment (10.7)	AVHHSZ90, MOB90, HODENT90
Hispanic, foreign-born (9)	SHRHSP90, SHRFOR90
Gender and Housing (7.4)	GROUPQ90, FEMR90, MOB90
2000 (total variance explained: 74.3%)	
Race and socio-economic status (27.6)	SHRBLK00, FFH00, TRVLPB00, EDUCUH00,
	UNEMPRT00, SERVOCC00, POVRAT00,
	WELFARE00, AVHHIN00, RNTUR00,
	AVGRNT00
Age, labor force (13.3)	SHRASN00, OLD00, FEMLABR00, LABPOP00
Household size, employment (10.9)	AVHHSZ00, MANUF00, RNTUR00, HODENT00
Hispanic, foreign-born (9.5)	SHRHSP00, SHRFOR00
Income, home value (7.3)	SHRIND00, AVHHIN00, AVGVAL00
Employment and housing (5.7)	GROUPQ00, UNEMPRT00
2010 (total variance explained: 65.8%)	
Race and socio-economic status (23.8)	SHRBLK10, FFH10, TRVLPB10, EDUCUH10,
	UNEMPRT10, SERVOCC10, POVRAT10,
	WELFARE10, AVHHIN10, RNTUR10,
	AVGVAL10, AVGRNT10
Age, labor force (16.2)	OLD10, FEMLABR10, WELFARE10, LABPOP10
Hispanic, foreign-born (10)	SHRHSP10, SHRFOR10
Household size, housing (8.6)	KIDS10, AVHHSZ10
Gender and employment (7.2)	FEMR10, MANUF10, MOB10, HODENT10

Table A3: PCA results for Hillsborough County

APPENDIX B CORRELATION BIPLOTS



Figure B1: Correlation biplots (with alpha bag) of Harris County



Figure B2: Correlation biplots (with alpha bag) of Orleans Parish



Figure B3: Correlation biplots (with alpha bag) of Hillsborough County

APPENDIX C

MODEL RESULTS AND COMPARISONS

		8	,	
Variables	OLS	Spatial Lag	Spatial Durbin	Spatial Error
(Intercept)	-0.472 (0.194)*	-0.125 (0.188)	-0.192 (0.506)	-0.61 (0.25)*
zfldrt:pchcvgr	-0.013 (0.018)	-0.014 (0.017)	-0.007 (0.017)	-0.015 (0.017)
zfldrt:pctcugr	0.004 (0.007)	0.001 (0.007)	0 (0.007)	0 (0.007)
pchcvgr:zpowr	-0.007 (0.011)	-0.009 (0.01)	-0.012 (0.011)	-0.01 (0.01)
pctcugr:zpowr	-0.008 (0.01)	-0.009 (0.009)	-0.014 (0.009)	-0.009 (0.009)
Pchcvgr	0.046 (0.015)**	0.034 (0.014)*	0.028 (0.014)	0.034 (0.014)*
Pctcugr	0.002 (0.005)	0.001 (0.005)	0.001 (0.005)	0.001 (0.005)
Zfldrt	-0.012 (0.029)	-0.013 (0.027)	-0.036 (0.033)	-0.024 (0.031)
Zpowr	-0.075 (0.036)*	-0.033 (0.034)	-0.005 (0.044)	-0.027 (0.039)
Popgr	0 (0)	0 (0)	0 (0)	0 (0)
Blkgr	0.03 (0.004)***	0.022 (0.004)***	0.026 (0.004)***	0.025 (0.004)***
Hspgr	0.033 (0.003)***	0.025 (0.003)***	0.029 (0.003)***	0.03 (0.003)***
mdrnt2000	0 (0)	0 (0)*	0 (0)*	0 (0)*
pov2000	-0.021 (0.005)***	-0.015 (0.005)***	-0.018 (0.005)***	-0.019 (0.005)***
hsp2000	0.01 (0.002)***	0.006 (0.002)***	0.012 (0.003)***	0.01 (0.002)***
shrblk2000	0.01 (0.002)***	0.007 (0.002)***	0.015 (0.003)***	0.011 (0.002)***
cbd_dist	0.01 (0.006)	-0.002 (0.005)	-0.026 (0.045)	0.018 (0.011)
lag.pchcvgr			-0.006 (0.061)	
lag.pctcugr			0.019 (0.027)	
lag.zfldrt			0.161 (0.097)	
lag.zpowr			0.022 (0.152)	
lag.popgr			0 (0.001)	
lag.blkgr			0.015 (0.013)	
lag.hspgr			-0.003 (0.009)	
lag.mdrnt2000			0.001 (0.001)	
lag.pov2000			0.012 (0.016)	
lag.hsp2000			-0.01 (0.006)	
lag.shrblk2000			-0.017 (0.006)**	
lag.cbd_dist			0.013 (0.05)	
lag.zfldrt:pchcvgr			0.087 (0.072)	
lag.zfldrt:pctcugr			0.002 (0.033)	
lag.pchcvgr:zpowr			0.003 (0.061)	
lag.pctcugr:zpowr			-0.062 (0.061)	

Table C1: Models estimating change in standardized value of SoVI (using power function for technological hazards)

		Table C1 (cont.)		
rho/lambda		0.574 (0.061)***	0.516 (0.076)***	0.661 (0.064)***
Adj.R2	0.2896			
AIC	1780.967	1715.4	1719.9	1714.5
Log likelihood	-872.4837	-838.7081	-824.9665	-838.2343
* < 05 ** < 01 ***-	. < 001			

 Table C2: Lagrange Multiplier diagnostics for OLS model estimating change in standardized value of SoVI (using power function for technological hazards)

	Statistic	df	p-value
LM (error)	92.88	1	0.000
LM (lag)	95.908	1	0.000
Robust LM (error)	6.864	1	0.009
Robust LM (lag)	9.892	1	0.002
SARMA	102.772	2	0.000

Table C3: Likelihood Ratio (LR) test for spatial regression models estimating change in standardized value of SoVI (using power function for technological hazards)

	Log Likelihood	Likelihood ratio with SDM	
Spatial Durbin Model	-824.966		
Spatial Lag Model	-838.708	27.483*	
Spatial Error Model	-838.234	26.536*	
* <i>p</i> <.05, ** <i>p</i> <.01, *** <i>p</i> <.001; df=16			

Table C4: Models estimating change in standardized value of SoVI (using WCPE function for technological hazards)

		8	~/	
Variables	OLS	Spatial Lag	Spatial Durbin	Spatial Error
(Intercept)	-0.492 (0.195)*	-0.124 (0.188)	-0.277 (0.503)	-0.625 (0.254)*
zfldrt:pchcvgr	-0.013 (0.018)	-0.014 (0.017)	-0.009 (0.017)	-0.017 (0.017)
zfldrt:pctcugr	0.004 (0.007)	0.001 (0.007)	0.002 (0.007)	0 (0.007)
pchcvgr:zcpe1	-0.016 (0.011)	-0.018 (0.01)	-0.019 (0.011)	-0.018 (0.009)
pctcugr:zcpe1	-0.015 (0.012)	-0.015 (0.012)	-0.02 (0.012)	-0.016 (0.011)
Pchcvgr	0.048 (0.015)**	0.035 (0.014)*	0.03 (0.014)*	0.035 (0.014)*
Pctcugr	0.001 (0.005)	0 (0.005)	0 (0.005)	0 (0.005)
Zfldrt	-0.01 (0.029)	-0.011 (0.027)	-0.028 (0.033)	-0.021 (0.031)
zcpe1	-0.005 (0.036)	0.027 (0.033)	0.062 (0.039)	0.033 (0.035)
Popgr	0 (0)	0 (0)	0 (0)	0 (0)
Blkgr	0.03 (0.004)***	0.022 (0.004)***	0.026 (0.004)***	0.025 (0.004)***
Hspgr	0.033 (0.003)***	0.025 (0.003)***	0.029 (0.003)***	0.03 (0.003)
mdrnt2000	0 (0)	0 (0)*	0 (0)*	0 (0)*
pov2000	-0.022 (0.005)***	-0.016 (0.005)***	-0.017 (0.005)**	-0.019 (0.005)***

hsp2000 0.009 (0.002)*** 0.006 (0.002)** 0.013 (0.003)*** 0.01 (0.002)*	***
shrblk2000 0.01 (0.002)*** 0.007 (0.002)*** 0.015 (0.003)*** 0.011 (0.002)*	***
cbd_dist 0.01 (0.006) -0.001 (0.005) -0.033 (0.045) 0.019 (0.0	11)
lag.pchcvgr 0.009 (0.06)	
lag.pctcugr 0.008 (0.028)	
lag.zfldrt 0.123 (0.097)	
lag.zpowr -0.193 (0.187)	
lag.popgr 0 (0.001)	
lag.blkgr 0.016 (0.013)	
lag.hspgr -0.008 (0.009)	
lag.mdrnt2000 0.001 (0)	
lag.pov2000 0.009 (0.016)	
lag.hsp2000 -0.008 (0.006)	
lag.shrblk2000 -0.014 (0.006)*	
lag.cbd_dist 0.021 (0.05)	
lag.zfldrt:pchcvgr 0.088 (0.071)	
lag.zfldrt:pctcugr 0.019 (0.031)	
lag.pchcvgr:zpowr 0.024 (0.072)	
lag.pctcugr:zpowr -0.094 (0.091)	
rho/lambda 0.592 (0.059)*** 0.518 (0.075)*** 0.676 (0.062) ⁵	***
Adj.R2 0.2834	
AIC 1787.633 1716.6 1717.7 171	4.3
Log likelihood -875.8167 -839.3218 -823.8262 -838.1	588

 Table C5: Lagrange Multiplier diagnostics for OLS model estimating change in standardized value of SoVI (using WCPE function for technological hazards)

	Statistic	df	p-value
LM (error)	106.362	1	0.000
LM (lag)	104.920	1	0.000
Robust LM (error)	9.817	1	0.002
Robust LM (lag)	8.375	1	0.004
SARMA	114.737	2	0.000

Table C6: Likelihood Ratio (LR) test for spatial regression models estimating change in standardized value of SoVI (using WCPE function for technological hazards)

	Log Likelihood	Likelihood ratio with SDM	
Spatial Durbin Model	-823.826		
Spatial Lag Model	-839.3218	30.991*	
Spatial Error Model	-838.1588	28.665*	
* <i>p</i> < .05, ** <i>p</i> < .01, *** <i>p</i> < .001; df=16			

		0	,	
Variables	OLS	Spatial Lag	Spatial Durbin	Spatial Error
(Intercept)	-0.146 (1.822)	0.092 (1.782)	2.308 (4.876)	-1.087 (2.111)
zfldrt:pchcvgr	-0.158 (0.172)	-0.192 (0.168)	-0.181 (0.167)	-0.241 (0.165)
zfldrt:pctcugr	0.029 (0.068)	0.02 (0.066)	0 (0.065)	0.012 (0.064)
pchcvgr:zpowr	-0.017 (0.104)	-0.029 (0.101)	-0.12 (0.102)	-0.054 (0.099)
pctcugr:zpowr	0.278 (0.093)**	0.274 (0.091)**	0.212 (0.089)*	0.273 (0.089)**
Pchcvgr	0.343 (0.141)*	0.287 (0.138)*	0.256 (0.139)	0.284 (0.138)*
Pctcugr	0.154 (0.051)**	0.144 (0.05)**	0.142 (0.049)**	0.15 (0.049)**
Zfldrt	0.551 (0.272)*	0.46 (0.266)	0.304 (0.319)	0.449 (0.288)
Zpowr	-0.849 (0.341)*	-0.619 (0.333)	0.219 (0.421)	-0.358 (0.367)
Popgr	-0.01 (0.003)***	-0.008 (0.003)**	-0.01 (0.003)**	-0.009 (0.003)***
Blkgr	0.174 (0.037)***	0.155 (0.037)***	0.22 (0.039)***	0.187 (0.038)***
Hspgr	0.312 (0.027)**	0.278 (0.029)***	0.334 (0.032)***	0.316 (0.029)***
mdrnt2000	-0.003 (0.002)	-0.003 (0.002)	-0.003 (0.002)	-0.003 (0.002)
pov2000	-0.274 (0.046)***	-0.244 (0.045)***	-0.288 (0.051)***	-0.277 (0.048)***
hsp2000	0.139 (0.018)***	0.115 (0.019)***	0.184 (0.027)***	0.148 (0.021)***
shrblk2000	0.09 (0.018)***	0.076 (0.018)***	0.166 (0.029)***	0.105 (0.021)***
cbd_dist	0.062 (0.052)	0.019 (0.053)	-0.305 (0.439)	0.096 (0.078)
lag.pchcvgr			0.068 (0.589)	
lag.pctcugr			-0.079 (0.261)	
lag.zfldrt			1.155 (0.938)	
lag.zpowr			-1.308 (1.481)	
lag.popgr			0.01 (0.009)	
lag.blkgr			-0.123 (0.12)	
lag.hspgr			-0.061 (0.091)	
lag.mdrnt2000			0.002 (0.005)	
lag.pov2000			0.206 (0.152)	
lag.hsp2000			-0.137 (0.062)*	
lag.shrblk2000			-0.209 (0.062)***	
lag.cbd_dist			0.233 (0.481)	
lag.zfldrt:pchcvgr			1.581 (0.694)*	
lag.zfldrt:pctcugr			-0.067 (0.319)	
lag.pchcvgr:zpowr			-0.136 (0.588)	
lag.pctcugr:zpowr			-0.565 (0.59)	
rho/lambda		0.308 (0.074)***	0.284 (0.091)**	0.482 (0.08)***
Adj.R2	0.2872	. ,	· · ·	· · ·
AIC	5222.408	5208.6	5198.1	5196.5
Log likelihood	-2593.204	-2585.282	-2564.032	-2579.226
*p<.05, **p<.01, *	*** <i>p</i> <.001			

Table C7: Models estimating change in % of population under poverty (using power function for technological hazards)

165

	Statistic	df	p-value
LM (error)	32.925	1	0.000
LM (lag)	18.926	1	0.000
Robust LM (error)	14.241	1	0.000
Robust LM (lag)	0.242	1	0.623
SARMA	33.167	2	0.000

 Table C8: Lagrange Multiplier diagnostics for OLS model estimating change in % of population under poverty (using power function for technological hazards)

Table C9: Likelihood Ratio (LR) test for spatial regression models estimating change in % of population under poverty (using power function for technological hazards)

	Log Likelihood	Likelihood ratio with SDM
Spatial Durbin Model	-2564.03	
Spatial Lag Model	-2585.28	42.499***
Spatial Error Model	-2579.23	30.386*
*p<.0	.001; df=16	

 Table C10: Models estimating change in % of population under poverty (using WCPE function for technological hazards)

Variables	OLS	Spatial Lag	Spatial Durbin	Spatial Error
(Intercept)	-0.173 (1.821)	0.133 (1.778)	0.524 (4.846)	-1.088 (2.109)
zfldrt:pchcvgr	-0.122 (0.172)	-0.16 (0.167)	-0.159 (0.166)	-0.211 (0.165)
zfldrt:pctcugr	-0.002 (0.067)	-0.012 (0.065)	-0.013 (0.064)	-0.017 (0.064)
pchcvgr:zcpe1	-0.096 (0.1)	-0.11 (0.097)	-0.105 (0.109)	-0.109 (0.093)
pctcugr:zcpe1	0.38 (0.116)**	0.373 (0.113)***	0.341 (0.115)**	0.349 (0.11)**
Pchcvgr	0.308 (0.141)*	0.246 (0.137)	0.239 (0.138)	0.243 (0.137)
Pctcugr	0.167 (0.051)**	0.157 (0.05)**	0.155 (0.05)**	0.161 (0.049)**
Zfldrt	0.559 (0.272)*	0.462 (0.265)	0.324 (0.32)	0.452 (0.288)
zcpe1	-0.181 (0.333)	-0.002 (0.325)	0.359 (0.38)	0.084 (0.335)
Popgr	-0.01 (0.003)***	-0.008 (0.003)**	-0.01 (0.003)**	-0.009 (0.003)***
Blkgr	0.171 (0.037)***	0.15 (0.036)***	0.217 (0.039)***	0.182 (0.038)***
Hspgr	0.317 (0.028)***	0.28 (0.029)***	0.336 (0.032)***	0.317 (0.029)***
mdrnt2000	-0.002 (0.002)	-0.003 (0.002)	-0.003 (0.002)	-0.003 (0.002)
pov2000	-0.279 (0.046)***	-0.245 (0.045)***	-0.283 (0.051)***	-0.276 (0.048)***
hsp2000	0.135 (0.018)***	0.109 (0.019)***	0.183 (0.028)***	0.144 (0.021)***
shrblk2000	0.09 (0.018)***	0.075 (0.018)***	0.163 (0.029)***	0.105 (0.021)***
cbd_dist	0.071 (0.052)	0.024 (0.053)	-0.256 (0.439)	0.106 (0.078)
lag.pchcvgr			0.324 (0.581)	
lag.pctcugr			-0.084 (0.273)	
lag.zfldrt			0.785 (0.938)	
lag.zpowr			-3.002 (1.815)	
lag.popgr			0.012 (0.009)	_

		Table C10 (cont.)		
lag.blkgr			-0.092 (0.121)	
lag.hspgr			-0.083 (0.092)	
lag.mdrnt2000			0.003 (0.005)	
lag.pov2000			0.19 (0.152)	
lag.hsp2000			-0.122 (0.061)*	
lag.shrblk2000			-0.182 (0.061)**	
lag.cbd_dist			0.179 (0.479)	
lag.zfldrt:pchcvgr			1.616 (0.691)*	
lag.zfldrt:pctcugr			0.126 (0.3)	
lag.pchcvgr:zpowr			0.252 (0.701)	
lag.pctcugr:zpowr			0.204 (0.88)	
rho/lambda		0.328 (0.074)***	0.316 (0.089)***	0.483 (0.08)***
Adj.R2	0.2869			
AIC	5222.682	5206.3	5198.1	5194.7
Log likelihood	-2593.341	-2584.14	-2564.033	-2578.334
Moran's I for				
residuals	0.117***		0.125***	
p < .05, mp < .01, m	p < .001			

Table C11: Lagrange Multiplier diagnostics for OLS model estimating change in % of population under poverty (using WCPE function for technological hazards)

	Statistic	df	p-value
LM (error)	38.620	1	0.000
LM (lag)	22.746	1	0.000
Robust LM (error)	16.127	1	0.000
Robust LM (lag)	0.253	1	0.615
SARMA	38.873	2	0.000

Table C12: Likelihood Ratio (LR) test for spatial regression models estimating change in % of population under poverty (using WCPE function for technological hazards)

	Log Likelihood	Likelihood ratio with SDM	
Spatial Durbin Model	-2564.033		
Spatial Lag Model	-2584.14	40.214***	
Spatial Error Model	-2578.334	28.602*	
* <i>p</i> < .05, ** <i>p</i> < .01, *** <i>p</i> < .001; df=16			

	12110110	veennoro Bieur		
Variables	OLS	Spatial Lag	Spatial Durbin	Spatial Error
(Intercept)	0.162 (0.886)	0.248 (0.86)	2.481 (2.365)	-0.198 (1.06)
zfldrt:pchcvgr	0.072 (0.084)	0.095 (0.081)	0.069 (0.081)	0.078 (0.08)
zfldrt:pctcugr	-0.037 (0.033)	-0.045 (0.032)	-0.045 (0.032)	-0.049 (0.031)
pchcvgr:zpowr	-0.016 (0.05)	-0.016 (0.049)	-0.054 (0.05)	-0.025 (0.048)
pctcugr:zpowr	0.054 (0.045)	0.053 (0.044)	0.038 (0.043)	0.055 (0.043)
Pchcvgr	0.37 (0.069)***	0.328 (0.067)***	0.296 (0.067)***	0.333 (0.067)***
Pctcugr	0.026 (0.025)	0.028 (0.024)	0.025 (0.024)	0.026 (0.024)
Zfldrt	0.272 (0.132)*	0.223 (0.128)	-0.049 (0.155)	0.134 (0.141)
Zpowr	-0.044 (0.166)	0.031 (0.161)	0.328 (0.204)	0.143 (0.18)
Popgr	-0.003 (0.001)*	-0.002 (0.001)	-0.003 (0.001)	-0.002 (0.001)
Blkgr	0.255 (0.018)***	0.242 (0.018)***	0.273 (0.019)***	0.263 (0.018)***
Hspgr	0.003 (0.013)	-0.005 (0.013)	0.003 (0.016)	0.003 (0.014)
mdrnt2000	-0.001 (0.001)	-0.001 (0.001)	-0.001 (0.001)	-0.001 (0.001)
pov2000	-0.135 (0.022)***	-0.117 (0.022)***	-0.113 (0.025)***	-0.123 (0.023)***
hsp2000	0.053 (0.009)***	0.046 (0.009)***	0.04 (0.013)**	0.048 (0.01)***
shrblk2000	0.043 (0.009)***	0.041 (0.009)***	0.064 (0.014)***	0.047 (0.01)***
cbd_dist	0.003 (0.025)	-0.019 (0.025)	-0.069 (0.213)	0.014 (0.041)
lag.pchcvgr			0.079 (0.288)	
lag.pctcugr			-0.069 (0.126)	
lag.zfldrt			1.09 (0.455)*	
lag.zpowr			-0.363 (0.711)	
lag.popgr			0.006 (0.004)	
lag.blkgr			-0.173 (0.061)**	
lag.hspgr			-0.024 (0.041)	
lag.mdrnt2000			-0.001 (0.002)	
lag.pov2000			-0.025 (0.074)	
lag.hsp2000			0.014 (0.03)	
lag.shrblk2000			-0.07 (0.03)*	
lag.cbd_dist			0.039 (0.233)	
lag.zfldrt:pchcvgr			-0.086 (0.336)	
lag.zfldrt:pctcugr			0.128 (0.154)	
lag.pchcvgr:zpowr			-0.146 (0.285)	
lag.pctcugr:zpowr			-0.295 (0.285)	
rho/lambda		0.397 (0.072)***	0.395 (0.085)***	0.54 (0.075)***
Adj.R2	0.3327	· · · ·	· /	. /
AIC	4113.338	4092.2	4087	4084
Log likelihood	-2038.669	-2027.111	-2008.487	-2023.006
Moran's I for				
residuals	0.069***		0.075***	
k 05 ** 01 *	** < 001			

Table C13: Models estimating change in % of population Black and under poverty (using power function for technological hazards)

	Statistic	df	p-value
LM (error)	32.176	1	0.000
LM (lag)	25.299	1	0.000
Robust LM (error)	7.153	1	0.008
Robust LM (lag)	0.276	1	0.599
SARMA	32.452	2	0.000

Table C14: Lagrange Multiplier diagnostics for OLS model estimating change in % of population Black and under poverty (using power function for technological hazards)

Table C15: Likelihood Ratio (LR) test for spatial regression models estimating change in % of population Black and under poverty (using power function for technological hazards)

	Log Likelihood	Likelihood ratio with SDM	
Spatial Durbin Model	-2008.49		
Spatial Lag Model	-2027.11	37.247**	
Spatial Error Model	-2023.01	29.037*	
* <i>p</i> <.05, ** <i>p</i> <.01, *** <i>p</i> <.001; df=16			

 Table C16: Models estimating change in % of population Black and under poverty (using WCPE function for technological hazards)

Variables	OLS	Spatial Lag	Spatial Durbin	Spatial Error
(Intercept)	0.181 (0.885)	0.282 (0.858)	2.407 (2.346)	-0.158 (1.056)
zfldrt:pchcvgr	0.08 (0.083)	0.103 (0.081)	0.082 (0.08)	0.086 (0.08)
zfldrt:pctcugr	-0.043 (0.032)	-0.051 (0.031)	-0.046 (0.031)	-0.054 (0.031)
pchcvgr:zcpe1	-0.01 (0.049)	-0.011 (0.047)	-0.045 (0.053)	-0.016 (0.045)
pctcugr:zcpe1	0.094 (0.056)	0.098 (0.054)	0.069 (0.056)	0.102 (0.053)
Pchcvgr	0.362 (0.068)***	0.32 (0.066)***	0.296 (0.067)***	0.325 (0.067)***
Pctcugr	0.031 (0.025)	0.033 (0.024)	0.026 (0.024)	0.031 (0.024)
Zfldrt	0.271 (0.132)*	0.221 (0.128)	-0.046 (0.155)	0.131 (0.141)
zcpe1	-0.006 (0.162)	0.034 (0.157)	0.283 (0.184)	0.104 (0.163)
Popgr	-0.003 (0.001)*	-0.002 (0.001)	-0.003 (0.001)	-0.003 (0.001)
Blkgr	0.254 (0.018)***	0.241 (0.018)***	0.272 (0.019)***	0.262 (0.018)***
Hspgr	0.004 (0.013)	-0.004 (0.013)	0.004 (0.015)	0.004 (0.014)
mdrnt2000	-0.001 (0.001)	-0.001 (0.001)	-0.001 (0.001)	-0.001 (0.001)
pov2000	-0.135 (0.022)***	-0.116 (0.022)***	-0.109 (0.025)***	-0.122 (0.023)***
hsp2000	0.052 (0.009)***	0.045 (0.009)***	0.042 (0.013)**	0.047 (0.01)***
shrblk2000	0.043 (0.009)***	0.041 (0.009)***	0.064 (0.014)***	0.047 (0.01)***
cbd_dist	0.004 (0.025)	-0.019 (0.025)	-0.104 (0.213)	0.013 (0.041)
lag.pchcvgr			0.161 (0.284)	
lag.pctcugr			-0.16 (0.132)	
lag.zfldrt			0.988 (0.456)*	
lag.zpowr			-0.363 (0.872)	
lag.popgr			0.007 (0.004)	

Table C16 (cont.)					
lag.blkgr			-0.17 (0.061)**		
lag.hspgr			-0.04 (0.042)		
lag.mdrnt2000			-0.001 (0.002)		
lag.pov2000			-0.038 (0.074)		
lag.hsp2000			0.017 (0.03)		
lag.shrblk2000			-0.065 (0.029)*		
lag.cbd_dist			0.072 (0.232)		
lag.zfldrt:pchcvgr			-0.01 (0.334)		
lag.zfldrt:pctcugr			0.183 (0.145)		
lag.pchcvgr:zpowr			-0.137 (0.339)		
lag.pctcugr:zpowr			-0.789 (0.424)		
rho/lambda	0.1	398 (0.072)***	0.4 (0)***	0.536 (0.075)***	
Adj.R2).3338				
AIC 41	12.064	4090.5	4085.1	4082.7	
Log likelihood -203	38.032	-2026.242	-2007.56	-2022.33	
Moran's I for					
residuals 0.	07***		0.069***		

Table C17: Lagrange Multiplier diagnostics for OLS model estimating change in % of population Black and under poverty (using WCPE function for technological hazards)

	Statistic	df	p-value
LM (error)	33.185	1	0.000
LM (lag)	26.079	1	0.000
Robust LM (error)	7.386	1	0.007
Robust LM (lag)	0.281	1	0.596
SARMA	33.465	2	0.000

Table C18: Likelihood Ratio (LR) test for spatial regression models estimating change in % of population Black and under poverty (using WCPE function for technological hazards)

	Log Likelihood	Likelihood ratio with SDM	
Spatial Durbin Model	-2007.56		
Spatial Lag Model	-2026.242	37.364**	
Spatial Error Model	-2022.33	29.541*	
* <i>p</i> <.05, ** <i>p</i> <.01, *** <i>p</i> <.001; df=16			

Variables		Snatial Lag	Snatial Durbin	Spatial Error
(Intercent)	-1 812 (1 401)	_1 685 (1 38)		_2 232 (1 557)
zfldrt:nebeyer	-1.012(1.401)	-1.003(1.30)	-0.432(3.794)	-2.233 (1.337)
zfldrt:peteugr	-0.2(0.132)	-0.232(0.13)	-0.242(0.13)	$-0.273(0.129)^{\circ}$
ziturt.peteugi	0.043(0.032)	0.041(0.031)	0.022(0.031)	0.037(0.03)
penevgr.zpowr	-0.028 (0.08)	-0.034 (0.078)	-0.099(0.079)	-0.047(0.077)
peteugr:zpowr	$0.182(0.071)^{*}$	$0.179(0.07)^{*}$	0.155 (0.069)	$0.176(0.069)^{*}$
Penevgr	0.004 (0.109)	-0.012 (0.107)	0 (0.108)	-0.015 (0.107)
Peteugr	0.097 (0.039)*	0.089 (0.038)*	0.084 (0.038)*	0.095 (0.038)*
Zfldrt	0.404 (0.209)	0.369 (0.206)	0.458 (0.248)	0.416 (0.22)
Zpowr	-0.772 (0.262)**	-0.666 (0.258)**	-0.042 (0.327)	-0.523 (0.279)
Popgr	-0.005 (0.002)*	-0.005 (0.002)*	-0.006 (0.002)*	-0.005 (0.002)*
Blkgr	-0.017 (0.028)	-0.024 (0.028)	0.018 (0.03)	-0.009 (0.029)
Hspgr	0.387 (0.021)***	0.364 (0.023)***	0.402 (0.025)***	0.387 (0.022)***
mdrnt2000	-0.001 (0.001)	-0.001 (0.001)	-0.001 (0.001)	-0.001 (0.001)
pov2000	-0.077 (0.035)*	-0.067 (0.035)	-0.107 (0.039)**	-0.086 (0.036)*
hsp2000	0.091 (0.014)***	0.078 (0.015)***	0.136 (0.021)***	0.099 (0.015)***
shrblk2000	0.038 (0.014)**	0.03 (0.014)*	0.085 (0.022)***	0.046 (0.016)**
cbd_dist	0.011 (0.04)	-0.002 (0.04)	-0.072 (0.341)	0.027 (0.054)
lag.pchcvgr			-0.074 (0.455)	
lag.pctcugr			-0.048 (0.203)	
lag.zfldrt			-0.208 (0.726)	
lag.zpowr			-1.166 (1.151)	
lag.popgr			0.005 (0.007)	
lag.blkgr			-0.053 (0.093)	
lag.hspgr			-0.032 (0.077)	
lag.mdrnt2000			0.001 (0.004)	
lag.pov2000			0.2 (0.117)	
lag.hsp2000			-0.13 (0.048)**	
lag.shrblk2000			-0.131 (0.048)**	
lag.cbd_dist			-0.007 (0.373)	
lag.zfldrt:pchcvgr			1.69 (0.54)**	
lag.zfldrt:pctcugr			-0.103 (0.247)	
lag.pchcvgr:zpowr			-0.093 (0.457)	
lag.pctcugr:zpowr			-0.145 (0.458)	
rho/lambda		0.195 (0.072)**	0.227 (0.095)*	0.38 (0.087)***
Adi R2	0 4099		(
AIC	4818 134	4813 1	4807 5	4804
Log likelihood	-2391 067	-2387 557	-2368 733	-2382.99
Moran's I for	2371.007	2301.331	2300.733	2302.99
residuals	0.052***			

Table C19: Models estimating change in % of population Hispanic and under poverty (using
power function for technological hazards)

	Statistic	df	p-value
LM (error)	18.781	1	0.000
LM (lag)	7.876	1	0.005
Robust LM (error)	11.202	1	0.001
Robust LM (lag)	0.297	1	0.586
SARMA	19.078	2	0.000

Table C20: Lagrange Multiplier diagnostics for OLS model estimating change in % of population Hispanic and under poverty (using power function for technological hazards)

Table C21: Likelihood Ratio (LR) test for spatial regression models estimating change in % of population Hispanic and under poverty (using power function for technological hazards)

	Log	Likelihood ratio with	
	Likelihood	SDM	
Spatial Durbin			
Model	-2368.733		
Spatial Lag Model	-2387.557	37.649**	
Spatial Error Model	-2382.99	28.513*	
* <i>p</i> < .05, ** <i>p</i> < .01, *** <i>p</i> < .001; df=16			

Table C22: Models estimating change in % of population Hispanic and under poverty (using
WCPE function for technological hazards)

Variables	OLS	Spatial Lag	Spatial Durbin	Spatial Error
(Intercept)	-1.887 (1.402)	-1.719 (1.38)	-2.05 (3.771)	-2.305 (1.566)
zfldrt:pchcvgr	-0.175 (0.132)	-0.212 (0.13)	-0.229 (0.129)	-0.258 (0.128)*
zfldrt:pctcugr	0.023 (0.051)	0.021 (0.05)	0.015 (0.05)	0.017 (0.05)
pchcvgr:zcpe1	-0.092 (0.077)	-0.101 (0.076)	-0.075 (0.084)	-0.099 (0.073)
pctcugr:zcpe1	0.258 (0.089)**	0.252 (0.088)**	0.249 (0.09)**	0.229 (0.086)**
Pchcvgr	-0.013 (0.108)	-0.033 (0.106)	-0.008 (0.107)	-0.038 (0.107)
Pctcugr	0.107 (0.039)**	0.098 (0.039)*	0.097 (0.038)*	0.102 (0.038)**
Zfldrt	0.408 (0.209)	0.37 (0.206)	0.475 (0.248)	0.422 (0.22)
zcpe1	-0.273 (0.256)	-0.162 (0.253)	0.073 (0.295)	-0.114 (0.258)
Popgr	-0.005 (0.002)*	-0.005 (0.002)*	-0.006 (0.002)*	-0.005 (0.002)*
Blkgr	-0.019 (0.028)	-0.027 (0.028)	0.016 (0.03)	-0.011 (0.029)
Hspgr	0.39 (0.021)***	0.365 (0.023)	0.403 (0.025)***	0.388 (0.022)***
mdrnt2000	-0.001 (0.001)	-0.001 (0.001)	-0.001 (0.001)	-0.001 (0.001)
pov2000	-0.082 (0.035)*	-0.07 (0.035)*	-0.106 (0.039)**	-0.088 (0.036)*
hsp2000	0.089 (0.014)***	0.073 (0.015)***	0.134 (0.021)***	0.097 (0.016)***
shrblk2000	0.039 (0.014)**	0.03 (0.014)*	0.083 (0.022)***	0.047 (0.016)**
cbd_dist	0.018 (0.04)	0.003 (0.04)	-0.007 (0.34)	0.036 (0.055)
lag.pchcvgr			0.138 (0.448)	

Table C22 (cont.)				
lag.pctcugr			0.012 (0.212)	
lag.zfldrt			-0.454 (0.724)	
lag.zpowr			-2.99 (1.414)*	
lag.popgr			0.007 (0.007)	
lag.blkgr			-0.022 (0.093)	
lag.hspgr			-0.039 (0.077)	
lag.mdrnt2000			0.003 (0.004)	
lag.pov2000			0.192 (0.117)	
lag.hsp2000			-0.117 (0.047)*	
lag.shrblk2000			-0.109 (0.047)*	
lag.cbd_dist			-0.076 (0.372)	
lag.zfldrt:pchcvgr			1.653 (0.536)**	
lag.zfldrt:pctcugr			0.028 (0.232)	
lag.pchcvgr:zpowr			0.357 (0.544)	
lag.pctcugr:zpowr			0.917 (0.684)	
rho/lambda		0.216 (0.072)**	0.243 (0.093)**	0.395 (0.086)***
Adj.R2	0.4081			
AIC	4820.529	4813.8	4804.3	4803.8
Log likelihood	-2392.265	-2387.918	-2367.154	-2382.907
Moran's I for	0 050***			
	0.039			

Table C23: Lagrange Multiplier diagnostics for OLS model estimating change in % of population Hispanic and under poverty (using WCPE function for technological hazards)

	Statistic	df	p-value
LM (error)	23.489	1	0.000
LM (lag)	9.920	1	0.002
Robust LM (error)	13.971	1	0.000
Robust LM (lag)	0.402	1	0.526
SARMA	23.891	2	0.000

Table C24: Likelihood Ratio (LR) test for spatial regression models estimating change in % of population Hispanic and under poverty (using WCPE function for technological hazards)

		Likelihood	
	Log Likelihood	ratio with SDM	
Spatial Durbin			
Model	-2367.154		
Spatial Lag Model	-2387.918	41.528***	
Spatial Error Model	-2382.907	31.506*	
* <i>p</i> <.05, ** <i>p</i> <.01, *** <i>p</i> <.001; df=16			

APPENDIX D

HUD GUIDELINE FOR CALCULATING ACCEPTABLE SEPARATION DISTANCE (ASD)

24 CFR §51 (Subpart C) gives instructions for identifying hazardous materials and how to calculate Acceptable Separation Distance (ASD) for developing any kind of HUD assisted project. Appendix I and II of this section gives the list of the hazardous materials and shows the calculation methods to be followed. Those appendices are presented here which might be helpful for planners in case of location decisions for any multi-family low income housing project.

Office of the Secretary, HUD

(a) Of HUD to disapprove a project proposal if the siting is too close to a potential hazard not covered by this subpart, and (b) of HUD or any person or other entity to seek to abate or to collect damages occasioned by a nuisance, whether or not covered by the subpart.

Appendix I to Subpart C of Part 51— Specific Hazardous Substances

The following is a list of specific petroleum products and chemicals defined to be hazardous substances under §51.201.

HAZARDOUS LIQUIDS

Acetic Acid Ethyl Benzene Acetic Anhydride Ethyl Dichloride Ethyl Ether Acetone Acrylonitrile Gasoline Amyl Acetate Heptane Amyl Alcohol Hexane Benzene Isobutyl Acetate Butyl Acetate Isobutyl Alcohol Butyl Acrylate Isopropyl Acetate Butyl Alcohol Isopropyl Alcohol Carbon Bisulfide Jet Fuel and Carbon Disulfide Kerosene Cellosolve Methyl Alcohol Cresols Methyl Amyl Alcohol Crude Oil Methyl Cellosolve (Petroleum) Methyl Ethyl Ketone Cumene Naptha Cyclohexane Pentane No. 2 Diesel Fuel Propylene Oxide Ethyl Acetate Toluene Ethyl Acrylate Vinyl Acetate Ethyl Alcohol Xylene

HAZARDOUS GASES

Acetaldehy	vde	Liquefied Natural		
Butadiene	·	Gas (LNG)		
Butane		Liquefied Petroleum		
Ethene		Gas (LPG)		
Ethylene		Propane		
Ethylene (Oxide	Propylene		
Hydrogen		Vinyl (Chloride	
(Primary	Source:	"Urban	Development	

Siting with respect to Hazardous Commercial/Industrial Facilities," by Rolf Jensen and Associates, Inc., April 1982)

[49 FR 5105, Feb. 10, 1984; 49 FR 12214, Mar. 29, 1984]

Appendix II to Subpart C of Part 51— Development of Standards; Calculation Methods

I. Background Information Concerning the Standards

(a) Thermal Radiation:

(1) Introduction. Flammable products stored in above ground containers represent a definite, potential threat to human life and structures in the event of fire. The resulting fireball emits thermal radiation which is absorbed by the surroundings. Combustible structures, such as wooden houses, may be ignited by the thermal radiation being emitted. The radiation can cause severe burn, injurtes and even death to exposed persons some distance away from the site of the fire.

(2) Criteria for Acceptable Separation Distance (ASD). Wooden buildings, window drapes and trees generally ignite spontaneously when exposed for a relatively long period of time to thermal radiation levels of approximately 10,000 Btu/hr. sq. ft. It will take 15 to 20 minutes for a building to ignite at that degree of thermal intensity. Since the reasonable response time for fire fighting units in urbanized areas is approximately five to ten minutes, a standard of 10,000 BTU/ hr. sq. ft. is considered an acceptable level of thermal radiation for buildings.

People in outdoor areas exposed to a thermal radiation flux level of approximately 1,500 Btu/ft² hr will suffer intolerable pain after 15 seconds. Longer exposure causes blistering, permanent skin damage, and even death. Since it is assumed that children and the elderly could not take refuge behind walls or run away from the thermal effect of the fire within the 15 seconds before skin blistering occurs, unprotected (outdoor) areas, such as playgrounds, parks, yards, school grounds, etc., must be placed at such a distance from potential fire locations so that the radiation flux level is well below 1500 Btu/ft² hr. An acceptable flux level, particularly for elderly people and children, is 450 Btu/ft² hr. The skin can be exposed to this degree of thermal radiation for 3 minutes or longer with no serious detrimental effect. The result would be the same as a bad sunburn. Therefore, the standard for areas in which there will be exposed people, e.g. outdoor recreation areas such as playgrounds and parks, is set at 450 Btu/hr. sq. ft. Areas covered also include open space ancillary to residential structures, such as yard areas and vehicle parking areas.

(3) Acceptable Separation Distance From a Potential Fire Hazard. This is the actual setback required for the safety of occupied buildings and their inhabitants, and people in open spaces (exposed areas) from a potential fire hazard. The specific distance required for safety from such a hazard depends upon the nature and the volume of the substance. The Technical Guidebook entitled "Urban Development Siting With Respect to Hazardous/Commercial Industrial Facilities," which supplements this regulation, contains the technical guidance required to compute Acceptable Separation Distances (ASD) for those flammable substances most often encountered.

(b) Blast Overpressure: The Acceptable Separation Distance (ASD) for people and structures from materials prone to explosion is

175
Pt. 51, Subpt. C, App. II

dependent upon the resultant blast measured in pounds per square inch (psi) overpressure. It has been determined by the military and corroborated by two independent studies conducted for the Department of Housing and Urban Development that 0.5 psi is the acceptable level of blast overpressure for both buildings and occupants, because a frame structure can normally withstand that level of external exertion with no serious structural damage, and it is unlikely that human beings inside the building would normally suffer any serious injury. Using this as the safety standard for blast overpressure, nomographs have been developed from which an ASD can be determined for a given quant1fv of hazardous substance. These nomographs are contained in the handbook with detailed instructions on their use.

(c) Hazard evaluation: The Acceptable Separation Distances for buildings, which are determined for thermal radiation and blast overpressure, delineate separate identifiable danger zones for each potential accident source. For some materials the fire danger zone will have the greatest radius and cover the largest area, while for others the explosion danger zone will be the greatest. For example, conventional petroleum fuel products stored in unpressurized tanks do not emit blast overpressure of dangerous levels when ignited. In most cases, hazardous substances will be stored in pressurized containers. The resulting blast overpressure will be experienced at a greater distance than the resulting thermal radiation for the standards set in Section 51.203. In any event the hazard requiring the greatest separation distance will prevail in determining the location of HUDassisted projects.

The standards developed for the protection of people and property are given in the following table.

	Thermal radi- ation	Blast over- pressure
Amount of acceptable expo- sure allowed for building	10,000 BTU/ ft²hr.	0.5 psi.

24 CFR	Subtitle A	(4 - 1 - 1)	3 Edition)
--------	------------	-------------	------------

	Thermal radi- ation	Blast over- pressure
Amount of acceptable expo- sure allowed for people in open areas.	450 BTU/ft² hr	0.5 psi.

Problem Example

The following example is given as a guide to assist in understanding how the procedures are used to determine an acceptable separation distance. The technical data are found in the HUD Guidebook. Liquid propane is used in the example since it is both an explosion and a fire hazard.

In this hypothetical case a proposed housing project is to be located 850 feet from a 30,000 gallon liquid propane (LPG) tank. The objective is to determine the acceptable separation distance from the LPG tank. Since propane is both explosive and fire prone it will be necessary to determine the ASD for both explosion and for fire. The greatest of the two will govern. There is no dike around the tank in this example.

Nomographs from the technical Guidebook have been reproduced to facilitate the solving of the problem.

ASD For Explosion

Use Figure 1 to determine the acceptable separation distance for explosion.

The graph depicted on Figure 1 is predicated on a blast overpressure of 0.5 psi.

The ASD in feet can be determined by applying the quantity of the hazard (in gallons) to the graph.

In this case locate the 30,000 gallon point on the horizontal axis and draw a vertical line from that point to the intersection with the straight line curve. Then draw a horizontal line from the point where the lines cross to the left vertical axis where the AC-CEPTABLE SEPARATION DISTANCE of 660 feet is found.

Therefore the ASD for explosion is 660 feet

Since the proposed project site is located 850 feet from the tank it is located at a safe distance with regards to blast overpressure.





Figure 1

ASD For Fire

To determine the ASD for fire it will be necessary to first find the fire width (diameter of the fireball) on Figure 2. Then apply this to Figure 3 to determine the ASD.

Since there are two safety standards for fire: (a) 10,000 BTU/ft²hr. for buildings; and (b) 450 BTU/ft²hr. for people in exposed areas,

it will be necessary to determine an ASD for each.

To determine the fire width locate the 30,000 gallon point on the horizontal axis on *Figure 2* and draw a vertical line to the straight line curve. Then draw a horizontal line from the point where the lines cross to the left vertical axis where the FIRE WIDTH is found to be 350 feet.

355

177

Pt. 51, Subpt. C, App. II

Now locate the 350 ft. point on the horizontal axis of *Figure 3* and draw a vertical line from that point to curves 1 and 2. Then draw horizontal lines from the points where the lines cross to the left vertical axis where the ACCEPTABLE SEPARATION DISTANCES of 240 feet for buildings and 1,150 feet for exposure to people is found.

24 CFR Subtitle A (4-1-13 Edition)

Based on this the proposed project site is located at a safe distance from a potential fireball. However, exposed playgrounds or other exposed areas of congregation must be at least 1,150 feet from the tank, or be appropriately shielded from a potential fireball. (Source: HUD Handbook, "Urban Development Siting With Respect to Hazardous Commercial/Industrial Facilities.")



Figure 2

356

178



Figure 3

[49 FR 5105, Feb. 10, 1984; 49 FR 12214, Mar. 29, 1984]

357